## UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

## FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the

Securities Exchange Act of 1934

Date of report (Date of earliest event reported): September 7, 2010

## **CLEAN ENERGY FUELS CORP.**

(Exact Name of Registrant as Specified in its Charter)

Delaware (State or Other Jurisdiction of Incorporation) **001-33480** (Commission File Number) **33-0968580** (I.R.S. Employer Identification No.)

3020 Old Ranch Parkway, Suite 400, Seal Beach, California

(Address of Principal Executive Offices)

**90740** (Zip Code)

Registrant's telephone number, including area code: (562) 493-2804

Not Applicable

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

o Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

#### Item 2.01 Completion of Acquisition or Disposition of Assets.

On September 7, 2010, Clean Energy Fuels Corp, a Delaware corporation (the "Registrant"), acting through certain of its subsidiaries, completed its purchase of the advanced natural gas fueling compressor and related equipment manufacturing and servicing business (the "Acquired Business") of I.M.W. Industries Ltd., a British Columbia corporation ("IMW").

The terms and conditions of the purchase of the Acquired Business are set forth in an Asset Purchase Agreement (the "Asset Purchase Agreement") dated July 1, 2010 by and among Clean Energy, a California corporation and a wholly-owned subsidiary of the Registrant ("Clean Energy - CA"), 0884808 B.C. Ltd., a British Columbia corporation and a wholly-owned subsidiary of Clean Energy - CA ("Canadian AcqCo"), 0884810 B.C. Ltd., a British Columbia corporation and a wholly-owned subsidiary of Clean Energy - CA ("Canadian AcqCo"), 0884810 B.C. Ltd., a British Columbia corporation and a wholly-owned subsidiary of Canadian AcqCo ("Canadian OpCo" and collectively with Clean Energy - CA and Canadian AcqCo, "Purchasers"), on the one hand, and IMW, 652322 B.C. Ltd., a British Columbia corporation ("Holdco"), Miller Family Trust and Bradley N. Miller (IMW, Holdco, Miller Family Trust and Bradley N. Miller are collectively referred to herein as "Sellers"), on the other hand.

Pursuant to the Asset Purchase Agreement as amended, at closing the Sellers received (a) \$15.6 million in cash, (b) 4.0 million shares of the Registrant's common stock, (c) the following promissory notes issued by a subsidiary of the Registrant (collectively the "Notes"): (i) a promissory note with a principal amount of \$12,500,000 that is due and payable on January 31, 2011, (ii) a promissory note with a principal amount of \$12,500,000 that is due and payable on January 31, 2011, (ii) a promissory note with a principal amount of \$12,500,000 that is due and payable on January 31, 2012, (iii) a promissory note with a principal amount of \$12,500,000 that is due and payable on January 31, 2013, and (iv) a promissory note with a principal amount of \$12,500,000 that is due and payable on January 31, 2014 (each payment under such promissory notes will consist of \$5.0 million in cash and \$7.5 million in cash and/or shares of the Registrant's common stock (the exact combination of cash and/or stock to be determined at Purchasers' option)); (d) an earn-out arrangement pursuant to which Purchasers will, over a four year period, pay Sellers up to \$40.0 million based on the Acquired Business achieving certain minimum amounts of adjusted gross profit; and (e) Purchasers assuming certain liabilities of the Acquired Business. In addition, pursuant to a Security Agreement executed at closing, the Notes are secured by a security interest in the Acquired Business.

The Purchasers have also agreed to pay the Sellers an additional \$288,000 when the transfer of IMW Compressors (Shanghai) Co., Ltd. is approved by the Chinese regulatory authorities. The shares of the Registrant's common stock issued to the Sellers were not registered under the Securities Act in reliance upon an exemption from registration provided by Regulation S promulgated thereunder, as the issuance of such shares is an "offshore transaction" within the meaning of Rule 902 of Regulation S. Such securities may not be offered or sold in the United States absent registration or an applicable exemption from registration requirements. Pursuant to the Asset Purchase Agreement, the Registrant agreed to register the Sellers' shares of the Registrant's common stock on or prior to the closing to permit resale of the shares in the United States.

The Asset Purchase Agreement also provides that Sellers will, subject to certain limitations, indemnify Purchasers for damages and losses incurred or suffered by Purchasers as a result of, among other things, breaches of Sellers' representations, warranties and covenants contained in the Asset Purchase Agreement.

The foregoing description of the Asset Purchase Agreement does not purport to be complete and is qualified in its entirety by the Asset Purchase Agreement attached hereto as Exhibit 2.5 and incorporated herein by reference. In addition, in connection with the closing of the purchase of the Acquired Business, the parties to the Asset Purchase Agreement entered into an Amendment to the Asset Purchase Agreement, dated September 7, 2010, which is attached hereto as Exhibit 2.6 and incorporated herein by reference.

A copy of the Registrant's press release, dated September 7, 2010, announcing the completion of the purchase of the Acquired Business is attached hereto as Exhibit 99.3 and incorporated herein by reference.

#### Item 3.02. Unregistered Sales of Equity Securities.

The information set forth under Item 2.01 is hereby incorporated by reference into this Item 3.02.

#### Item 9.01 Financial Statements, Pro Forma Financial Information and Exhibits.

(a) Financial Statements of Acquired Business

Audited combined financial statements of I.M.W. Industries Group as of and for the year ended April 30, 2010 as required by this Item 9.01 (a) are attached as Exhibit 99.1 hereto and incorporated into this Item 9.01(a) by reference.

(b) Pro Forma Financial Information

The unaudited pro forma condensed combined financial information as required by this Item 9.01(b) is attached as Exhibit 99.2 hereto and incorporated into this Item 9.01(b) by reference.

(d) Exhibits

Exhibit No.	Description
2.5	Asset Purchase Agreement, dated July 1, 2010, among Clean Energy, a California corporation, 0884808 B.C. Ltd., a British Columbia corporation, and 0884810 B.C. Ltd., a British Columbia corporation, on the one hand, and I.M.W. Industries Ltd., a British Columbia corporation, 652322 B.C. Ltd., a British Columbia corporation, Miller Family Trust and Bradley N. Miller, on the other hand. Certain schedules and exhibits referenced in the Asset Purchase Agreement have been omitted in accordance with Item 601(b)(2) of Regulation S-K. A copy of any omitted schedule and/or exhibit will be furnished supplementally to the Securities and Exchange Commission upon request.
2.6	Amendment to Asset Purchase Agreement, dated as of September 7, 2010, by and among Clean Energy, a California corporation, 0884808 B.C. Ltd., a British Columbia corporation and a wholly-owned subsidiary of Clean Energy - CA, and Clean Energy Compression Corp, a British Columbia corporation formerly known as 0884810 B.C. Ltd and a wholly-owned subsidiary of Canadian AcqCo, on the one hand, and I.M.W. Industries Ltd., a British Columbia Corporation, B&M Miller Equity Holdings Inc., a successor by amalgamation to 652322 B.C. Ltd., a British Columbia corporation, Bradley N. Miller, Marion G. Miller and Miller Family Trust, on the other hand. Certain schedules and exhibits referenced in the Amendment to Asset Purchase Agreement have been omitted in accordance with Item 601(b)(2) of Regulation S-K. A copy of any omitted schedule and/or exhibit will be furnished supplementally to the Securities and Exchange Commission upon request.
5.1	Opinion of Morrison & Foerster LLP.
23.1	Consent of Independent Registered Public Accounting Firm KPMG LLP.
23.2	Consent of Morrison & Foerster LLP (included in Exhibit 5.1).
99.1	Audited combined financial statements of I.M.W. Industries Group as of and for the year ended April 30, 2010.
99.2	Unaudited pro forma condensed combined financial information.
99.3	Clean Energy Fuels Corp. Press Release dated September 7, 2010.
	2

#### SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

#### **Clean Energy Fuels Corp.**

Dated: September 7, 2010

By: /s/ Richard R. Wheeler Richard R. Wheeler Chief Financial Officer

#### ASSET PURCHASE AGREEMENT

#### by and among

## **CLEAN ENERGY,** a California corporation;

**0884808 B.C. Ltd.**, a British Columbia corporation;

**0884810 B.C. Ltd.**, a British Columbia corporation;

**I.M.W. INDUSTRIES LTD.** a British Columbia corporation;

**652322 B.C. Ltd.**, a British Columbia corporation;

#### BRADLEY N. MILLER; and

#### MILLER FAMILY TRUST

Dated as of July 1, 2010

#### TABLE OF CONTENTS

		Page
ARTICLE I	THE TRANSACTIONS	2
1.1	First Closing	2
1.2	Second Closing	4
1.3	Third Closing	4
1.4	Excluded Assets	
1.5	Assumed Liabilities	4 5
1.6	Excluded Liabilities	5
1.7	Assignment and Assumption	7
1.8	Closings	7
1.9	Purchase Price	7
1.10	Purchase Price Adjustments	11
ARTICLE II	CLOSING DELIVERIES	12
2.1	First Closing Deliveries by Sellers and Company	12
2.2	First Closing Deliveries by Purchasers	14
2.3	Second Closing Deliveries by Sellers	14
2.4	Second Closing Deliveries by Purchasers	15
2.5	Third Closing Deliveries by Sellers, Holdco and Company	15
2.6	Third Closing Deliveries by Purchasers	15
ARTICLE III	REPRESENTATIONS AND WARRANTIES REGARDING COMPANY AND COMPANY SUBSIDIARIES	16
3.1	Organization, Good Standing, Qualification	16
3.2	Charter Documents; Books and Records	16
3.3	Reserved	17
3.4	Authority; Binding Nature of Agreements	17
3.5	No Conflicts; Required Consents	17
3.6	Subsidiaries	18
3.7	Financial Statements	18
3.8	Absence of Undisclosed Liabilities	20
3.9	Absence of Changes	20
3.10	Accounts Receivable	21
3.11	Indebtedness	22
3.12	Inventory	22
3.13	Material Contracts	22
3.14	Intellectual Property	24
3.15	Employees and Consultants	28
3.16	Company Benefit Plans	30
3.17	Compliance with Laws	32

- 3.18
- Governmental Approvals Proceedings and Orders Environmental Matters 3.19 3.20
- 3.21 Taxes

# TABLE OF CONTENTS (continued)

i

4.1Authority; Validity of Contemplated Transactions404.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43			Page
3.23Rel Propery863.24Title; Sufficiency; Condition of Assets373.25Customers and Suppliers383.26Transactions with Affiliates393.27Probert393.28Product Return Policies; Waranties393.29Product Return Policies; Waranties393.30OFAC403.31Cultural Business403.32Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority; Validity of Contemplated Transactions404.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.5414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS435.1Organization and Good Standing425.2Authority; Binding Nature of Agreements435.3Brokers435.4Proceedings435.5Second Closing Shares436.1Company's Conduct of the Business Prior to Closing436.2Restrictions446.3Access to Information466.4<	3 77	Incurance	36
3.24Title: Sufficiency: Condition of Assets373.25Customers and Suppliers383.26Transactions with Affiliates383.27Brokers393.28Product Return Policies; Warranties393.29Product Liabilities403.31Cultural Business403.32Full Disclosure403.32Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority: Validity of Contemplated Transactions404.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.5414.7Conduct Prior to Closing424.8Finco Shares424.9Finco Shares425.1Organization and Good Standing425.2Authority: Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares435.6Construction435.6Construction435.6Construction435.6Second Closing Shares435.6Second Closing Shares435.6Construction446.1Construction456.2Restrictions on Company's Conduct of the Business Prior to Closing44<			
3.25     Customers and Suppliers     38       3.26     Transactions with Affiliates     39       3.27     Brokers     39       3.28     Product Return Policies; Warranties     39       3.30     OFAC     40       3.31     Cultural Business     40       3.32     Full Disclosure     40       ARTICLE IV     REPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO     40       ARTICLE IV     REPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO     40       4.1     Authority; Validity of Contemplated Transactions     40       4.2     No Conflicts; Required Consents     41       4.3     Proceedings     41       4.4     Brokers     41       4.5     Sophistication     41       4.6     Non-U.S     41       4.7     Conduct Prior to Closing     42       4.8     Finco     42       ARTICLE V     REPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS     42       5.1     Organization and Good Standing     42       5.2     Authority; Binding Nature of Agreements     42       5.3     Brokers     43       5.4     Proceedings     43       5.5     Status of Canadian AcqCo     43       5.6     Second Closing Sh			
3.26Transactions with Affiliates383.27Brokers393.28Product Return Policies; Warranties393.29Product Liabilities393.30OFAC403.31Cultural Business403.32Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority; Validity of Contemplated Transactions414.3Proceedings414.4Brokers414.5Sophistication414.6Nor-U.S414.7Conflict; Required Consents414.8Brokers414.4Brokers414.5Sophistication414.6Nor-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing436.2Restrictions456.4Certain Notifications466.5Reasonable Commercial Efforts476.6Consents476.7Consents47 <td></td> <td></td> <td></td>			
3.27Brokers393.28Product Reum Policies; Warranties393.30OFAC393.30OFAC403.31Cultural Business403.32Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority; Validity of Contemplated Transactions404.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing414.8Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing436.2Reatication456.4Cernin Norifications466.5Access to Information456.6Reasonable Commercial Efforts476.7Consents476.8Confinentiality47			
3.28Product Rerum Policies; Warranties393.20Product Liabilities393.30OFAC403.31Cultural Business403.32Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority: Validity of Contemplated Transactions414.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority: Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status Of Candian AcqCo435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing436.2Reatrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Reasonable Commercial Efforts476.6Considentiality47 </td <td></td> <td></td> <td></td>			
3.29Product Liabilities393.30OFAC403.31Cultural Business403.32Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority; Validity of Contemplated Transactions404.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing436.3No Solicitation456.4Certain Notifications456.5Access to Information456.6Resonable Commercial Efforts476.7Consents476.8Confidentality47			
3.30OFAC403.31Cultural Business403.32Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority; Validity of Contemplated Transactions404.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-US414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information456.4Certain Notifications466.5Consents476.8Confidentality47			
3.31 3.32Cultural Business Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority; Validity of Contemplated Transactions404.2No Conflicts; Required Consents404.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing S.2425.2Authority; Binding Nature of Agreements435.3Status of Canadian AcqCo435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing 6.3436.1Company's Conduct of the Business Prior to Closing 6.4436.3No Solicitation456.4Certain Nutifications466.5Access to Information456.6Resonable Commercial Efforts 6.7476.8Confidentiality47			
3.32Full Disclosure40ARTICLE IVREPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO404.1Authority; Validity of Contemplated Transactions404.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco424.9Finco424.9Finco425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Nutifications466.5Access to Information456.4Certain Nutifications466.5Access to Information456.6Consents476.7Consents476.8Conflictation466.4Centainon456.5Access to Information466.6Consents476.7Consents476.7Consents476			
4.1Authority; Validity of Contemplated Transactions404.2No Conflicts; Required Consents414.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority: Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS43ARTICLE VIPRE-CLOSING COVENANTS43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions456.4Certain Notifications456.5Access to Information456.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48			
4.2No Conflicts; Required Consents414.3Procedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco424.9Finco425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications456.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	ARTICLE IV	REPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO	40
4.2No Conflicts; Required Consents414.3Procedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco424.9Finco425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications456.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	4.1	Authority: Validity of Contemplated Transactions	40
4.3Proceedings414.4Brokers414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48			
4.4Brokers414.5Sophistication414.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.8Confidentiality48			
4.5Sophistication414.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares435.6Second Closing Shares436.1Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.8Confidentiality48		-	
4.6Non-U.S414.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing446.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.8Confidentiality48			
4.7Conduct Prior to Closing424.8Finco Shares424.9Finco42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.8Confidentiality48			41
4.8 4.9Finco42 42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS42ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing436.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.8Confidentiality48	4.7		42
ARTICLE VREPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS425.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48		-	42
5.1Organization and Good Standing425.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	4.9	Finco	42
5.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	ARTICLE V	REPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS	42
5.2Authority; Binding Nature of Agreements425.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	5.1	Organization and Good Standing	42
5.3Brokers435.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications456.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	5.2		42
5.4Proceedings435.5Status of Canadian AcqCo435.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications456.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	5.3		43
5.6Second Closing Shares43ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48			43
ARTICLE VIPRE-CLOSING COVENANTS436.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	5.5	Status of Canadian AcqCo	43
6.1Company's Conduct of the Business Prior to Closing436.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	5.6	Second Closing Shares	43
6.2Restrictions on Company's Conduct of the Business Prior to Closing446.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	ARTICLE VI	PRE-CLOSING COVENANTS	43
6.3No Solicitation456.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	6.1	Company's Conduct of the Business Prior to Closing	43
6.4Certain Notifications466.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48		Restrictions on Company's Conduct of the Business Prior to Closing	44
6.5Access to Information466.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	6.3	No Solicitation	45
6.6Reasonable Commercial Efforts476.7Consents476.8Confidentiality48	6.4	Certain Notifications	46
6.7Consents476.8Confidentiality48	6.5	Access to Information	46
6.8 Confidentiality 48		Reasonable Commercial Efforts	47
•	6.7		47
6.9 Indebtedness 49	6.8	Confidentiality	48
	6.9	Indebtedness	49

## ii

### TABLE OF CONTENTS

(continued)

		Page	
6.10	Public Announcements	49	
6.11	Books and Records	49	
6.12	Cooperation	49	
6.13	Non-Competition and Non-Solicitation Agreements	49	
6.14	Certain Employee Matters	52	
6.15	Registration of CLNE Shares	52	
6.16	Corporate Name	54	
6.17	SEC Filings	54	
6.13 6.14 6.15 6.16	Non-Competition and Non-Solicitation Agreements Certain Employee Matters Registration of CLNE Shares Corporate Name	49 52 52 54	9 2 2 4

ARTICLE VII	CONDITIONS TO CLOSING	54		
7.1	Conditions to Purchasers' Obligation to Consummate First Closing	54		
7.2	Conditions to Company's Obligation to Consummate First Closing	55		
7.3	Conditions to Company's Obligation to Consummate First Closing 55 Conditions to Purchasers' Obligation to Consummate Second Closing 55			
7.4	Conditions to Company's, Holdco's and Sellers' Obligations to Consummate Second Closing 56			
7.5	Conditions to Purchasers' Obligation to Consummate Third Closing 56			
7.6	Conditions to Company's, Holdco's and Sellers' Obligations to Consummate Third Closing	56		
7.7	Conditions to Obligations of Each Party to Close	57		
ARTICLE VIII	TERMINATION	57		
8.1	Circumstances for Termination	57		
8.2	Effect of Termination	58		
ARTICLE IX	INDEMNIFICATION	58		
9.1	Survival of Representations and Warranties	58		
9.2 Indemnification by Sellers		59		
9.3	Procedures for Indemnification	60		
9.4	Limitations on Indemnification	61		
9.5	Offset	62		
9.6	Remedies Cumulative	62		
9.7	Nature of Indemnification Payments	62		
ARTICLE X	TAX MATTERS	62		
10.1	Amended Returns; Tax Elections	62		
10.2	Straddle Period Allocations	62		
10.3	Tax Elections	63		
10.4	Termination of Tax Sharing Agreements	63		
10.5	Allocation of Purchase Price	63		
10.6	Cooperation, Access to Information, and Record Retention	63		
ARTICLE XI	MISCELLANEOUS PROVISIONS	64		
11.1	Expenses	64		
11.2	Interpretation	64		
	iii			

## TABLE OF CONTENTS

(continued)

		Page
11.3	Further Assurances	64
11.4	Notices	65
11.5	Entire Agreement	66
11.6	Modifications, Amendments and Waivers	66
11.7	Successors and Assigns	66
11.8	Governing Law	66
11.9	Severability	66
11.10	Specific Performance	66
11.11	Counterparts	66
11.12	Dispute Resolution	67
11.13	Attorneys' Fees, Costs and Expenses	67
11.14	No Third-Party Beneficiaries	67
11.15	Privacy	68

#### EXHIBITS AND SCHEDULES

#### **Exhibits**

Exhibit A — Certain Definitions

Exhibit B — Examples of Calculation of Earn-Out Payment

Exhibit C — Company Historical Net Working Capital Calculation

Schedule 1.1(a) — Accounts Receivable Schedule 1.1(b) — Inventory Schedule 1.1(c) — Machinery and Equipment Schedule 1.1(d) — Owned and Leased Vehicles Schedule 1.1(e) — Personal Property Schedule 1.1(f) — Real Property Leases and Improvements Schedule 1.1(g) — Personal Property Leases Schedule 1.1(h) — Company IP Assets Schedule 1.1(i) — Deposits and Advances Schedule 1.1(j) — Material Contracts Schedule 1.1(k) — Governmental Approvals Schedule 1.1(l) — Company Claims Schedule 1.4(d) — Certain Other Property Schedule 1.5 — Assumed Liabilities Schedule 3.1(i) — Company Organization Schedule 3.1(ii) — Company Subsidiary Organization Schedule 3.5 — Consents Schedule 3.6 — Company Subsidiaries Schedule 3.7(f) — Accounts Payable Aging Schedule 3.11 — Indebtedness Schedule 3.13 — Material Contracts Schedule 3.14(a) — Registered IP Assets Schedule 3.14(e) — Certain Company Contracts Schedule 3.14(f) — Third Party Rights in Company IP Assets Schedule 3.14(j) — Payments Related to Company IP Assets Schedule 3.14(k) — Form of Non-Disclosure Agreement Schedule 3.15(a) — Employee Manuals Schedule 3.15(b) — Employees Schedule 3.16(a) — Company Benefit Plans Schedule 3.18(a) — Governmental Approvals Schedule 3.22 — Insurance Schedule 3.23(b) — Leased Real Property Schedule 3.25(a) — Customers Schedule 3.25(b) — Suppliers Schedule 3.28 — Product Return Policies; Warranties Schedule 6.1 — Company's Conduct of the Business Prior to Closing Schedule 6.2 — Restrictions on Company's Conduct of the Business Prior to Closing Schedule 6.9 — Permitted Encumbrances Schedule 6.14 — Key Employees

v

#### ASSET PURCHASE AGREEMENT

**THIS ASSET PURCHASE AGREEMENT** (the "*Agreement*") is made and entered into this 1<sup>st</sup> day of July, 2010, by and among Clean Energy, a California corporation ("*Clean Energy*"), 0884808 B.C. Ltd., a British Columbia corporation and a wholly-owned subsidiary of Clean Energy ("*Canadian AcqCo*"), 0884810 B.C. Ltd., a British Columbia corporation and a wholly-owned subsidiary of Canadian AcqCo ("*Canadian OpCo*" and collectively with Clean Energy and Canadian AcqCo, "*Purchasers*"), on the one hand, and I.M.W. Industries Ltd., a British Columbia corporation ("*Company*"), 652322 B.C. Ltd., a British Columbia corporation ("*Holdco*"), Miller Family Trust and Bradley N. Miller (each of Miller Family Trust and Bradley N. Miller a "*Seller*" and together, "*Sellers*"), on the other hand. Certain capitalized terms used in this Agreement are defined on <u>Exhibit A</u> hereto.

#### RECITALS

WHEREAS, Company, together with the Company Subsidiaries, is engaged in the business of designing, manufacturing and supplying industrial products, parts and services, including (a) compressors, dispensers, storage systems and services for compressed natural gas, and (b) high pressure gas system design and manufacturing, together with such (i) high tolerance mechanical component design and manufacturing, (ii) electrical and PLC control design and manufacturing, (iii) CNC machining, (iv) specialized coded pressure and structural welding, (v) heat transfer system design and manufacturing, (vi) specialized custom metal forming, (vii) testing and product performance verification services, and (viii) ISO certified quality control systems as is ancillary or related to (a) and (b) (the "*Business*");

**WHEREAS**, Purchasers desire to purchase from Company, and Company desires to sell to Purchasers, substantially all of the assets, properties, rights and claims of the Business on the terms and conditions set forth herein;

**WHEREAS**, at the First Closing, Company will sell the First Closing Purchased Assets to Canadian OpCo and enter the non-competition and nonsolicitation agreements set forth in <u>Section 6.13(a)</u> of this Agreement, and Canadian OpCo will (a) deliver to Company the First Closing Cash Payment, the First Closing Note, the First Closing Shares, the Future Payment Notes, the Earnout Shares, the Depreciable Assets Shares, and (b) assume the Assumed Liabilities;

**WHEREAS**, immediately following the First Closing, Company will transfer the Purchased Shares (other than the Retained Purchased Shares), the First Closing Note, the First Closing Shares, the Earnout Shares and the Depreciable Assets Shares to Holdco;

**WHEREAS**, at the Second Closing, Sellers will transfer the Finco Shares to Canadian OpCo, and Canadian OpCo will deliver to Sellers the Second Closing Shares;

WHEREAS, immediately following the Second Closing, Sellers will transfer the Second Closing Shares to Holdco;

1

**WHEREAS**, at the Third Closing, (a) Holdco will (i) sell its Purchased Shares to Canadian AcqCo, (ii) sell the Earnout Shares to Canadian AcqCo, (iii) sell the Depreciable Assets Shares to Canadian AcqCo, (iv) sell the First Closing Note and the First Closing Shares to Clean Energy, and (v) sell the Second Closing Shares to Canadian AcqCo, (b) Company will sell the Retained Purchased Shares to Finco; (c) Clean Energy will deliver the CLNE Shares to Holdco, and (d) Canadian AcqCo will (i) deliver the Year 1 Note to Holdco, (ii) pay to Sellers a cash amount equal to the principal amount of the Finco Note, and (iii) agree to make the Earn-Out Payments.

WHEREAS, at the Third Closing, Holdco and Sellers will enter the non-competition and non-solicitation agreements set forth in <u>Section 6.13(b)</u> of this Agreement;

**WHEREAS**, Canadian OpCo and Company will make the Section 85 Election, Section 22 Election, Section 20(24) Election, Section 56.4(3) Election and the Section 167 Election;

**NOW, THEREFORE**, in consideration of the foregoing recitals and the mutual representations, warranties, covenants and promises contained herein, the adequacy and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

#### AGREEMENT

#### ARTICLE I

#### THE TRANSACTIONS

1.1 First Closing. Subject to the terms and conditions of this Agreement, at the First Closing, Company shall sell, transfer, convey, assign and deliver to Canadian OpCo, and Canadian OpCo will purchase from Company, all of Company's right, title and interest in the assets, properties, goodwill and rights of Company used, held for use or intended to be used in the Business, other than the Excluded Assets and the Purchased Shares (collectively the "*First Closing Purchased Assets*"), including the following:

(a) <u>Accounts Receivable</u>. All Accounts Receivable, including the Accounts Receivable listed on <u>Schedule 1.1(a)</u>;

(b) <u>Inventory</u>. All Inventory, including all Inventory listed on <u>Schedule 1.1(b)</u>;

(c) <u>Machinery and Equipment</u>. All tools, machinery and equipment used, held for use or intended to be used in the Business, wherever located and whether held by Company or third parties (the "*Machinery and Equipment*"), including the Machinery and Equipment listed on <u>Schedule 1.1(c)</u>;

(d) <u>Owned and Leased Vehicles</u>. All vehicles owned by Company and all rights in vehicle leases to which Company is a party, in either case that are used, held for use or intended to be used in the Business (the "**Owned and Leased Vehicles**"), including the Owned and Leased Vehicles listed on <u>Schedule 1.1(d)</u>;

2

(e) <u>Personal Property</u>. All personal property, office furnishings, supplies and other tangible personal property used, held for use or intended to be used in the Business (the "*Personal Property*"), including the Personal Property listed on <u>Schedule 1.1(e)</u>;

(f) <u>Leased Real Property</u>. All rights in Real Property Leases used, held for use or intended to be used in the Business, together with all of Company's right, title and interest in all Improvements, including the Real Property Leases and the Improvements listed on <u>Schedule 1.1(f)</u>;

(g) <u>Personal Property Leases</u>. All rights in leases of personal property to which Company is a party used, held for use or intended to be used in the Business (the "*Personal Property Leases*"), including the Personal Property Leases listed on <u>Schedule 1.1(g)</u>;

(h) <u>Intellectual Property</u>. All Company IP Assets, including the Company IP Assets listed on <u>Schedule 1.1(h)</u>;

(i) <u>Deposits and Advances</u>. All performance and other bonds, security and other deposits, advance payments, prepaid credits, prepaid accounts and deferred charges used, held for use or intended to be used in, or arising out of or relating to, the Business (the "*Deposits and Advances*"), including the Deposits and Advances listed on <u>Schedule 1.1(i)</u>;

(j) <u>Contracts</u>. All rights under all Material Contracts listed on <u>Schedule 1.1(j)</u>;

(k) <u>Governmental Approvals</u>. All Governmental Approvals (and pending applications therefor) used, held for use or intended to be used in the Business, including the Governmental Approvals listed on <u>Schedule 1.1(k)</u>;

(l) <u>Claims</u>. All Company's insurance claims, rights to any insurance proceeds, and other similar claims relating to the Business or any Purchased Asset or Assumed Liability (the "*Company Claims*"), including the Company Claims listed on <u>Schedule 1.1(l)</u>;

(m) <u>Books and Records</u>. All books, files, papers, agreements, correspondence, databases, information systems, programs, software, documents and records relating to the Purchased Assets or the Assumed Liabilities, or used, held for use or intended to be used in the Business, on whatever

medium (the "Books and Records");

- (n) <u>Goodwill</u>. All goodwill generated by or associated with the Business;
- (o) <u>Domain Names</u>. The Domain Names and the data files for the content within the Domain Names;
- (p) <u>Management Services Agreement</u>. All rights of Company under Article VII of the Management Services Agreement.
- (q) <u>Cash</u>. All cash, cash equivalents and marketable securities of Company; and
  - 3

(r) <u>Other Assets</u>. All other assets, properties, rights and claims (other than the Purchased Shares and the Excluded Assets) used, held for use or intended to be used in the Business.

In addition, at the First Closing Company will enter the non-competition and non-solicitation agreements set forth in <u>Section 6.13(a)</u> of this Agreement.

**1.2** Second Closing. At the Second Closing, Sellers will sell, transfer, convey, assign and deliver to Canadian OpCo, and Canadian OpCo will purchase from Sellers, all of their right, title and interest in the Finco Shares.

**1.3 Third Closing**. At the Third Closing:

(a) Holdco will sell, transfer, convey, assign and deliver to Canadian AcqCo, and Canadian AcqCo will purchase from Holdco, all of Holdco's right, title and interest in all of its Purchased Shares;

(b) Company will sell, transfer, convey, assign and deliver to Finco, and Finco will purchase from Company, all of Company's right, title and interest in all of the Retained Purchased Shares;

(c) Holdco will sell, transfer, convey, assign and deliver to Canadian AcqCo, and Canadian AcqCo will purchase from Holdco, all of Holdco's right, title and interest in the Earnout Shares;

(d) Holdco will sell, transfer, convey, assign and deliver to Canadian AcqCo, and Canadian AcqCo will purchase from Holdco, all of Holdco's right, title and interest in the Depreciable Assets Shares;

(e) Holdco will sell, transfer, convey, assign and deliver to Canadian AcqCo, and Canadian AcqCo will purchase from Holdco, all of Holdco's right, title and interest in the Second Closing Shares;

(f) Holdco will sell, transfer, convey, assign and deliver to Clean Energy, and Clean Energy will purchase from Holdco, all of Holdco's right, title and interest in the First Closing Note and the First Closing Shares; and

(g) Holdco and Sellers will enter the non-competition and non-solicitation agreements set forth in <u>Section 6.13(b)</u> of this Agreement.

**1.4 Excluded Assets**. Notwithstanding <u>Sections 1.1</u>, <u>1.2</u> and <u>1.3</u>, the following assets of Company (the "*Excluded Assets*") shall not be included in the Purchased Assets:

(a) <u>Certain Debt</u>. Any intercompany or intracompany receivable cash balances between Company and any of its Affiliates or between any of its Affiliates;

4

(b) <u>Corporate Documents</u>. The corporate seal, certificate of incorporation, notice of articles, articles, minute book, share transfer records, or other records related to the corporate organization of Company;

(c) <u>Employee Benefit Contracts</u>. Company Benefit Plans and contracts of insurance for employee group medical, dental and life

- insurance plans;
  - (d) <u>Certain Other Property</u>. The assets listed on <u>Schedule 1.4(d)</u>;
  - (e) <u>Insurance Policies</u>. All insurance policies (except to the extent specified in <u>Section 1.1(l)</u>);

(f) <u>Records</u>. All personnel records and other records that Company is required by law to retain in its possession (provided that, to the extent permitted by applicable Legal Requirements, copies of such records will be provided to Purchasers); and

(g) <u>Rights Under Certain Agreements</u>. All of Company's rights under the Transaction Agreements.

**1.5 Assumed Liabilities**. Subject to the terms and conditions of this Agreement, at the Closing, Company shall assign, and Purchasers shall assume, the Assumed Liabilities. For the purposes of this Agreement, the "*Assumed Liabilities*" shall mean only the following Liabilities of Company, and in any event shall not include any Excluded Liabilities:

(a) Any Liability arising after the Closing Date under the Material Contracts listed on <u>Schedule 1.1(j)</u>; and

(b) The Liabilities of Company and the Company Subsidiaries specifically listed on <u>Schedule 1.5</u>.

**1.6** Excluded Liabilities. Except for the Assumed Liabilities, Purchasers shall not assume and shall not be liable or responsible for any Liability of Company (collectively, the "Excluded Liabilities"). Without limiting the foregoing, Company shall retain and be responsible for, and Purchasers shall not be obligated to assume, and do not assume, any Liability at any time arising from or attributable to:

(a) Any assets, properties or Contracts that are not included in the Purchased Assets;

(b) Any breaches of any Company Contract on or prior to the Closing Date or any payments or amounts due under any Company Contract on or prior to the Closing Date;

(c) Taxes attributable to or imposed upon Finco, Company and the Company Subsidiaries, or attributable to or imposed upon the Purchased Assets or the Business for the Pre-Closing Period (including any Pre-Closing Taxes);

(d) Any loans, other Indebtedness, or accounts payable;

5

(e) Accidents, misconduct, negligence, or breach of fiduciary duty occurring on or prior to the Closing Date;

(f) Any Proceeding initiated at any time, to the extent related to any action or omission on or prior to the Closing Date, including any Liability for (i) infringement or misappropriation of Intellectual Property Rights; (ii) breach of product warranties or warranty claims; (iii) injury, death, property damage or losses caused by Company Products; or (iv) violations of any Legal Requirements;

(g) Any Company Benefit Plans or any employee group medical, dental or life insurance plans or any other employee matter;

(h) Payments to employees or former employees of Company (including any employees of Company who are subsequently hired by Purchasers) or for payroll taxes imposed on or payable by Company or any of the Company Subsidiaries for taxable periods (and portions thereof) beginning on or before the Closing Date;

(i) The performance of this Agreement and the Transaction;

(j) Any Environmental and Safety Laws, which Liability relates to or arises out of (i) any acts or omissions of Company on or prior to the Closing Date, including any Release, management, disposal or arranging for disposal of Hazardous Materials in connection with the Business or the Purchased Assets or Assumed Liabilities or activities or operations occurring or conducted in connection with any predecessor operations of the Business or otherwise or (ii) any facts, circumstances or conditions existing on or prior to the Closing Date relating to Hazardous Materials;

(k) Any costs or expenses incurred in shutting down and removing equipment not purchased by Purchaser and any expenses associated with any Company Contracts not assumed by Purchaser hereunder;

(l) Any Liability for expenses and fees incurred by Company incidental to the preparation of the Transaction Documents, preparation or delivery of materials or information requested by Purchaser, and the consummation of the Transaction, including all broker, counsel and accounting fees;

(m) Any Legal Requirement applicable to Company, the Company Subsidiaries, the Purchased Assets or the Assumed Liabilities on or prior to the Closing Date or any Liability for a violation of such a Legal Requirement;

(n) Any claim made by any employee or former employee of Company (including any employees of Company who are subsequently hired by Purchasers) relating to the Transactions or employment with the Company prior to the Closing;

(o) Any notice, pay in lieu of notice, severance obligations or other special compensation arrangements with employees or former employees of Company (including any employees of Company who are subsequently hired by Purchasers) relating to the Transactions or employment with the Company prior to the Closing;

(p) Any obligations of Company under the Management Services Agreement.

6

#### 1.7 Assignment and Assumption.

(a) Notwithstanding anything herein to the contrary, if an attempted sale, assignment, transfer or delivery of any Purchased Asset would be ineffective without the Consent of any third party (including any Governmental Authority), or if such an act would violate the rights of any third party (including any Governmental Authority) in the Purchased Assets or otherwise affect adversely the rights of Purchasers in the Purchased Assets, and the applicable Consent has not been obtained on or prior to the Closing Date, this Agreement shall not constitute an actual or attempted sale, assignment, transfer or delivery of such Purchased Asset (each, a "*Restricted Asset*"). Unless and until any such Consent is obtained, such Restricted Asset shall not constitute an Assumed Liability for any purpose hereunder.

(b) In any such case, if Purchasers shall determine, in their sole discretion, to proceed with the Closing, Company shall use best efforts to obtain, as soon as practicable, such Consent. Purchasers shall cooperate reasonably with Company in obtaining such Consents, provided that Purchasers shall not be required to pay any cash consideration therefor or give or allow to remain in effect any guaranty, letter of credit, performance bond or other financial assurance.

(c) Until such Consent shall have been obtained, Company shall at its expense effect an alternate arrangement, in the form of a license, sublease, operating agreement or other arrangement, in any case satisfactory to Purchasers, which results in Purchasers receiving all the benefits and bearing all the ordinary course costs, liabilities and other obligations with respect to each Restricted Asset.

#### 1.8 Closings.

(a) The closing of the transactions contemplated by <u>Section 1.1</u> (the "*First Closing*") shall take place at the offices of Morrison & Foerster LLP, 12531 High Bluff Drive, San Diego, California, at 10:00 A.M. on the third (3rd) Business Day after the day on which all of the conditions to closing set forth in <u>Article VII</u> are satisfied or waived (other than conditions that are intended to be satisfied at the Closing), or at such other date, time or place as the parties may agree (the "*Closing Date*").

(b) The closing of the transactions contemplated by <u>Section 1.2</u> (the "**Second Closing**") shall take place at the offices of Morrison & Foerster LLP, 12531 High Bluff Drive, San Diego, California, at 10:30 A.M. on the Closing Date.

(c) The closing of the transactions contemplated by <u>Section 1.3</u> (the "*Third Closing*") shall take place at the offices of Morrison & Foerster LLP, 12531 High Bluff Drive, San Diego, California, at 11:00 A.M. on the Closing Date (the First Closing, the Second Closing and the Third Closing are sometimes collectively referred to herein as the "*Closing*").

**1.9 Purchase Price**. The aggregate consideration to be paid by Purchasers to Company, Holdco and Sellers for the Purchased Assets shall be determined and paid as follows (the consideration payable pursuant to this <u>Section 1.9</u> is referred to herein as the "*Purchase Price*"):

7

(a) At the First Closing, in exchange for the First Closing Purchased Assets and the non-competition and non-solicitation agreements set forth in Section 6.13(a) of this Agreement, Canadian OpCo will:

(i) pay to Company (by wire transfer of immediately available funds to a bank account designated by Company no less than five (5) Business Days prior to the Closing) an amount of cash equal to Fifteen Million Dollars (\$15,000,000), <u>plus (or minus)</u> the Estimated Net Working Capital Adjustment Amount (as calculated pursuant to <u>Section 1.10(a)</u>), and <u>minus</u> the principal amount of the Finco Note (such amount paid pursuant to this <u>Section 1.9(a)(i)</u>, the "**First Closing Cash Payment**");

(ii) deliver to Company the First Closing Note, the First Closing Shares and the Future Payment Notes;

- (iii) deliver to Company the Earnout Shares;
- (iv) deliver to Company the Depreciable Assets Shares; and
- (v) assume the Assumed Liabilities.

The allocation of the Purchase Price among the First Closing Purchased Assets and the non-competition and non-solicitation agreements set forth in <u>Section 6.13(a)</u> will be determined pursuant to <u>Section 10.5</u>.

- (b) At the Second Closing, in exchange for the Finco Shares, Canadian OpCo will deliver to Sellers the Second Closing Shares.
- (c) At the Third Closing:

(i) in exchange for the First Closing Note and the First Closing Shares, Clean Energy will deliver to Holdco a number of shares of CLNE Stock (the "*CLNE Shares*") equal to the aggregate of (A) Thirty Million Dollars (\$30,000,000) divided by the lower of the Midpoint Price and the Pre-Closing Price <u>plus</u> (B) Thirty Million Dollars (\$30,000,000) divided by the Midpoint Price;

(ii) in exchange for the Purchased Shares, the Retained Purchased Shares (which will be transferred by Company to Finco) and the Depreciable Assets Shares, Canadian AcqCo will deliver to Holdco the Year 1 Note (it being understood that a portion of the Year 1 Note will be allocated to the Retained Purchased Shares in connection with the procedure set forth in <u>Section 10.5</u>);

(iii) in exchange for the Second Closing Shares, Canadian AcqCo will pay to Sellers (by wire transfer of immediately available funds to a bank account designated by Sellers no less than five (5) Business Days prior to the Closing) an amount of cash equal to the principal amount of the Finco Note; and

8

(iv) in exchange for the Earnout Shares, Canadian AcqCo will pay to Holdco one or more earn-out payments (each an "*Earn-Out Payment*" and collectively the "*Earn-Out Payments*"), based upon the Adjusted Gross Profit of the Business for each Earn-Out Period, as follows (it being understood that the aggregate Earn-Out Payments paid to Holdco pursuant to this Agreement will not exceed Forty Million Dollars (\$40,000,000)):

(1) If, for any Earn-Out Period, the Business achieves Adjusted Gross Profit equal to or greater than 120% of the applicable Adjusted Gross Profit Target, then Canadian AcqCo will pay Holdco an amount of cash equal to 23.0% of the Adjusted Gross Profit of the Business for such Earn-Out Period.

(2) If, for any Earn-Out Period, the Business achieves Adjusted Gross Profit equal to or greater than 110%, but less than 120%, of the applicable Adjusted Gross Profit Target, then Canadian AcqCo will pay Holdco an amount of cash equal to 20.5% of the Adjusted Gross Profit of the Business for such Earn-Out Period.

(3) If, for any Earn-Out Period, the Business achieves Adjusted Gross Profit equal to or greater than 100%, but less than 110%, of the applicable Adjusted Gross Profit Target, then Canadian AcqCo will pay Holdco an amount of cash equal to 18.0% of the Adjusted Gross Profit of the Business for such Earn-Out Period.

(4) If, for any Earn-Out Period, the Business achieves Adjusted Gross Profit equal to or greater than 90%, but less than 100%, of the applicable Adjusted Gross Profit Target, then Canadian AcqCo will pay Holdco an amount of cash equal to 9.5% of the Adjusted Gross Profit of the Business for such Earn-Out Period.

(5) If, for any Earn-Out Period, the Business achieves Adjusted Gross Profit equal to or greater than 80%, but less than 90%, of the applicable Adjusted Gross Profit Target, then Canadian AcqCo will pay Holdco an amount of cash equal to 4.5% of the Adjusted Gross Profit of the Business for such Earn-Out Period.

(6) If, for any Earn-Out Period, the Business achieves Adjusted Gross Profit equal to or greater than 70%, but less than 80%, of the applicable Adjusted Gross Profit Target, then Canadian AcqCo will pay Holdco an amount of cash equal to 2.5% of the Adjusted Gross Profit of the Business for such Earn-Out Period.

(7) If, for any Earn-Out Period, the Business achieves Adjusted Gross Profit of less than 70% of the applicable Adjusted Gross Profit Target, then Holdco will not receive any Earn-Out Payment for such Earn-Out Period.

(d) As promptly as practicable, but in no event later than sixty (60) days following the last day of an Earn-Out Period, Purchaser shall cause a calculation of the Adjusted Gross Profit for such Earn-Out Period to be prepared and delivered to Sellers (the "*AGP Statement*") and shall provide Sellers the work papers used in the preparation of the AGP Statement. Except as set forth below, the AGP Statement shall be deemed to be and shall be final, binding and conclusive on the parties upon the earlier of (the "*AGP Resolution Date*"): (i) Sellers' delivery of a written notice to Purchasers of their approval of the AGP Statement; (ii) the failure of Sellers to notify Purchasers in writing of a dispute with the AGP Statement within thirty (30) days of the delivery of such documents to Sellers; (iii) the resolution of all disputes, pursuant to <u>Section 1.9(e)</u>, by Purchasers and Sellers; and (iv) the resolution of all disputes, pursuant to <u>Section 1.9(e)</u>, by the Independent Accounting Firm.

9

(e) Sellers may dispute any amounts reflected on the AGP Statement by delivery of a written notice to Purchasers (the "AGP Statement Dispute Notice"). If Sellers deliver an AGP Statement Dispute Notice to Purchasers, Purchasers and Sellers shall attempt to reconcile the parties' differences, and any resolution by them as to any disputed amounts shall be final, binding and conclusive on the parties. If Purchasers and Sellers are unable to reach a resolution within thirty (30) days after the delivery of the AGP Statement Dispute Notice, Purchasers and Sellers shall submit their respective determinations and calculations and the items remaining in dispute for resolution to an independent accounting firm of international reputation mutually acceptable to Purchasers and Sellers (the "Independent Accounting Firm"). The parties shall cause the Independent Accounting Firm to submit a report to Purchasers and Sellers with a determination regarding the remaining disputed items, within thirty (30) days after submission of the matter, and such report shall be final, binding and conclusive. Any party may seek enforcement of such report pursuant to Section 11.12(b). The fees, costs and expenses of the Independent Accounting Firm that is unsuccessfully disputed by each such party as finally determined by the Independent Accounting Firm bears to the total amount of such remaining disputed items.

(f) Each Earn-Out Payment that is earned by Holdco in accordance with <u>Section 1.9(c)(iv)</u> will be paid by Canadian AcqCo to Holdco within thirty (30) days following the AGP Resolution Date.

(g) Gross Profit will be calculated in accordance with GAAP, provided that if the application of GAAP results in a substantial change to any Adjusted Gross Profit Target, then the parties will work together to determine revised percentage payouts that reasonably preserve the intent of Section 1.9(c)(iv). For purposes of clarity, (i) examples of the calculation of the Earn-Out Payment are included as Exhibit B to this Agreement, (ii) Sellers, Holdco and Company agree that Purchasers have ultimate authority over the operation of the Business following the Closing Date, Purchasers do not have any obligation to take any actions toward achieving or maximizing the Earn-Out Payment, and in no event will any business decision of Purchasers that affects the operation of the Business have any bearing on Purchasers' obligations under this Agreement, and (iii) all of Purchasers' Earn-Out Obligations are subject to offset as set forth in Section 9.5.

(h) In addition, for purposes of this agreement, (i) no fractional CLNE Shares will be issued pursuant to this Agreement and all share amounts will be rounded down to the nearest whole share, and (ii) notwithstanding any provision of this Agreement, if the issuance of CLNE Shares pursuant to this Agreement would result in a Person holding CLNE voting securities valued (as such value is determined pursuant to the provisions of the HSR Act) at more than the applicable size of transaction threshold in effect under the HSR Act at the time of such issuance (the "*HSR Threshold*"), then Purchasers shall withhold a number of CLNE Shares having a value (as such value is determined pursuant to the provisions of the HSR Act) equal to the amount in excess of the applicable HSR Threshold and, in lieu of issuing such shares, deliver to such Person an amount of cash equal to such value.

10

#### 1.10 Purchase Price Adjustments.

(a) No later than five (5) Business Days prior to the Closing Date, Company shall deliver to Purchasers a statement (the "**Preliminary Closing Statement**") setting forth in reasonable detail a good faith calculation of (i) the Company Estimated Net Working Capital Adjusted Amount, (ii) the Colombia Estimated Net Working Capital Adjusted Amount, (iii) the China Estimated Net Working Capital Adjusted Amount and (iv) the Bangladesh Estimated Net Working Capital Adjusted Amount (the aggregate amount of (i), (ii), (iii), and (iv), positive or negative, the "**Estimated Net Working Capital**"). The First Closing Cash Payment shall be <u>increased</u>, dollar-for-dollar, by the amount by which the Estimated Net Working Capital is a positive number and shall be <u>decreased</u>, dollar-for-dollar, by the amount by which the Estimated Net Working Capital is a negative number (the amount by which the First Closing Cash Payment is adjusted, positive or negative, the "**Estimated Net Working Capital Adjusted Amount**").

(b) As promptly as practicable, but in no event later than sixty (60) days following the Closing Date, Purchasers shall cause to be prepared and delivered to Sellers a statement (the "*Final Closing Statement*") which sets forth in reasonable detail a calculation of (i) the Company Closing Net Working Capital Adjusted Amount, (ii) the Colombia Closing Net Working Capital Adjusted Amount, (iii) the Colombia Closing Net Working Capital Adjusted Amount, (iii) the Bangladesh Closing Net Working Capital Adjusted Amount (the aggregate amount of (i), (ii), (iii), and (iv), positive or negative, the "*Closing Net Working Capital*"). Purchasers shall provide Sellers the work papers used in the preparation of the Final Closing Statement. Except as set forth below, the Final Closing Statement shall be deemed to be and shall be final, binding and conclusive on the parties upon the earlier of (the "*Final Resolution Date*"): (i) Sellers' delivery of a written notice to Purchasers of their approval of the Final Closing Statement; (ii) the failure of Sellers to notify Purchasers in writing of a dispute with the Final Closing Statement within thirty (30) days of the delivery of such documents to Sellers; (iii) the resolution of all disputes, pursuant to <u>Section 1.10(c)</u>, by Purchasers and Sellers; and (iv) the resolution of all disputes, pursuant to <u>Section 1.10(c)</u>, by the Independent Accounting Firm.

(c) Sellers may dispute any amounts reflected on the Final Closing Statement by delivery of a written notice to Purchasers (the "*Final Closing Statement Dispute Notice*"). If Sellers deliver a Final Closing Statement Dispute Notice to Purchasers, Purchasers and Sellers shall attempt to reconcile the parties' differences, and any resolution by them as to any disputed amounts shall be final, binding and conclusive on the parties. If Purchasers and Sellers are unable to reach a resolution within thirty (30) days after the delivery of the Final Closing Statement Dispute Notice, Purchasers and Sellers shall submit their respective determinations and calculations and the items remaining in dispute for resolution to the Independent Accounting Firm. The parties shall cause the Independent Accounting Firm to submit a report to Purchasers and Sellers with a determination regarding the remaining disputed items, within thirty (30) days after submission of the matter, and such report shall be final, binding and conclusive. Any party may seek enforcement of such report pursuant to <u>Section 11.12(b)</u>. The fees, costs and expenses of the Independent Accounting Firm shall be paid by Purchasers and Sellers in the same proportion that the aggregate amount of such remaining disputed items so submitted to the Independent Accounting Firm that is unsuccessfully disputed by each such party as finally determined by the Independent Accounting Firm bears to the total amount of such remaining disputed items.

11

(d) The Purchase Price shall be increased or decreased on a dollar-for-dollar basis by the amount by which the Closing Net Working Capital finally determined in accordance with <u>Sections 1.10(b)</u> and <u>1.10(c)</u> (the "*Final Net Working Capital*") is greater or less than the amount of the Estimated Net Working Capital determined in accordance with <u>Section 1.10(a)</u>. Any such adjustment to the Purchase Price shall be referred to as a "*Net Working Capital Adjustment*."

(i) If the Final Net Working Capital is greater than the Estimated Net Working Capital, Purchasers shall pay an aggregate amount of cash equal to such difference to Company by wire transfer of immediately available funds within five (5) Business Days of the Final Resolution Date.

(ii) If the Final Net Working Capital is less than the Estimated Net Working Capital, Company and Sellers shall return an aggregate amount of cash equal to such difference to Purchasers by wire transfer of immediately available funds within five (5) Business Days of the Final Resolution Date.

#### ARTICLE II

#### **CLOSING DELIVERIES**

2.1 First Closing Deliveries by Sellers and Company. At or prior to the First Closing, Sellers and/or Company shall (i) take all steps necessary to place Purchasers in actual possession and operating control of the First Closing Purchased Assets and (ii) deliver, or cause to be delivered, the following items to Purchasers, all of which shall be in form and substance acceptable to Purchasers:

(a) <u>General Assignment and Bill of Sale</u>. A General Assignment and Bill of Sale covering all of the First Closing Purchased Assets (the "*General Assignment and Bill of Sale*");

(b) <u>Intellectual Property Assignment</u>. Any and all documents necessary to properly record the assignment to Purchasers of all of Company's right, title and interest in and to the Company IP Assets, including: (i) a trademark assignment for all of the Trademarks (the "*Trademark Assignment*"); (ii) a patent assignment for all of the Patents (the "*Patent Assignment*"); (iii) a copyright assignment for all of the Copyrights (the "*Copyright Assignment*"); and (iv) a domain name assignment for all of the Domain Names (the "*Domain Name Assignment*").

(c) <u>Assignments of Leases</u>. Assignments of all Real Property Leases and Personal Property Leases;

12

 (d)
 Estoppel Letters. Duly executed estoppel letters (the "Estoppel Letters") from each lessor of Leased Real Property or Personal

 (e)
 Owned and Leased Vehicles. Vehicle titles and assignments sufficient to transfer title to the Owned and Leased Vehicles;

 (f)
 Employment Agreements. Employment Agreements duly executed by each Key Employee (the "Employment Agreements");

(g) <u>Releases</u>. Release agreements, duly executed by each Seller;

(h) <u>Certificate of Representations and Warranties</u>. A Certificate executed on behalf of Company by its President certifying the matters in <u>Sections 7.1(a), (c) and (d)</u>;

(i) <u>Legal Opinions</u>. Legal opinions from Counsel to Sellers, Company and the Company Subsidiaries dated as of the Closing Date;

(j) <u>Resignations</u>. Evidence satisfactory to Purchasers of the resignation of each of the directors and each of the officers of the Company Subsidiaries in office immediately prior to the Closing as directors and/or officers, as applicable, of the Company Subsidiaries, effective as of the Closing;

(k) <u>Release of Encumbrances</u>. Evidence reasonably satisfactory to Purchasers that all Encumbrances (other than Permitted Encumbrances) on the Purchased Assets shall have been released prior to the Closing;

(l) <u>Transaction Agreements</u>. Duly executed copies of each of the Transaction Agreements to which any Seller, Holdco, Company or any Company Subsidiary is a party;

(m) <u>Consents</u>. Duly executed Consents of all Governmental Authorities and third parties required by Purchasers to consummate the Transaction, including the consents listed on <u>Schedule 3.5</u>;

(n) <u>Certificates of Good Standing</u>.

(i) A certificate from the British Columbia Registrar of Companies as to Company's and Finco's good standing and a clearance certificate issued by the Ministry of Finance of British Columbia under section 99 of the *Social Services Tax Act* (B.C.) attesting to the payment of all applicable taxes; and

(ii) A certificate from the jurisdiction of incorporation or formation, as the case may be, of each Company Subsidiary, as to each Company Subsidiary's good standing and payment of all applicable taxes.

(o) <u>Secretary's Certificate</u>. A certificate duly executed by the Secretary of Company, attaching correct and complete copies of the Company's certificate of incorporation, notice of articles, articles and resolutions of the Board of Directors and shareholders of Company authorizing this Agreement and the Transaction;

13

(p) <u>Preliminary Closing Statement</u>. The Preliminary Closing Statement;

(q) <u>Tax Elections</u>. The executed Section 85 Election, Section 22 Election, Section 20(24) Election and Section 167 Election (such elections to be held in escrow by counsel to Company until completion of the procedures set forth in <u>Section 10.5</u>);

(r) <u>FIRPTA Certificate</u>. A certificate issued pursuant to and in compliance with Treasury Regulations Section 1.1445-2(c)(3) (the "*FIRPTA Certificate*"), certifying that the Finco Shares are not U.S. real property interests within the meaning of Section 897 of the Internal Revenue Code of 1986, as amended, and a notice to the U.S. Internal Revenue Service regarding the FIRPTA Certificate and in compliance with Treasury Regulations Section 1.897-2(h)(2) (the "*IRS Notice*"), together with written authorization for Purchasers to deliver the IRS Notice to the IRS on behalf of Sellers upon the Closing;

(s) <u>Management Services Agreement</u>. An assignment of all rights of Company under Article VII of the Management Services Agreement; and

(t) <u>Other Documentation</u>. Such other certificates, instruments or documents required pursuant to the provisions of this Agreement or otherwise necessary or appropriate to transfer the First Closing Purchased Assets in accordance with the terms hereof and consummate the Transaction, and to vest in Purchasers and their successors and assigns full, complete, absolute, legal and equitable title to the First Closing Purchased Assets, free and clear of all Encumbrances (other than Permitted Encumbrances), including such certificates, instruments and documents to be executed or delivered by Company or Sellers pursuant to the terms of this Agreement.

2.2 **First Closing Deliveries by Purchasers**. At or prior to the First Closing, Purchasers shall deliver or cause to be delivered to Company:

(a) <u>Wire Transfer</u>. A wire transfer to HSBC Bank Canada for credit to Company's account, in the amount of the First Closing Cash

Payment;

- (b) <u>Promissory Notes</u>. The First Closing Note and the Future Payment Notes;
- (c) <u>Security Agreement</u>. A security agreement relating to the First Closing Note and the Future Payment Notes;
- (d) <u>Shares</u>. A certificate or certificates representing the Earnout Shares, the Depreciable Assets Shares and the First Closing Shares;
- and
- (e) <u>Transaction Agreements</u>. Duly executed copies of each of the Transaction Agreements to which any Purchaser is a party.
- 2.3 Second Closing Deliveries by Sellers. At or prior to the Second Closing, Sellers shall deliver or cause to be delivered to Purchasers:

14

(a) <u>Stock Certificates</u>. A certificate or certificates representing the Finco Shares duly endorsed in blank for transfer to Purchasers or accompanied by stock powers duly executed in blank; and

(b) <u>Work Papers</u>. Documentation of the tax advisor of Holdco in support of the representations and warranties set forth in <u>Section 3.21(i)</u>, and such work papers and documentation shall be in form satisfactory to Purchaser.

(c) <u>Other Documentation</u>. Such other certificates, instruments or documents required pursuant to the provisions of this Agreement or otherwise necessary or appropriate to transfer the Purchased Assets to be transferred at the Second Closing in accordance with the terms hereof and consummate the Transaction, and to vest in Purchasers and its successors and assigns full, complete, absolute, legal and equitable title to the Purchased Assets to be transferred at the Second Closing, free and clear of all Encumbrances (other than Permitted Encumbrances), including such certificates, instruments and documents to be executed or delivered by Company, Holdco or Sellers pursuant to the terms of this Agreement.

2.4 Second Closing Deliveries by Purchasers. At or prior to the Second Closing, Purchasers shall deliver or cause to be delivered to Sellers a certificate or certificates representing the Second Closing Shares.

2.5 Third Closing Deliveries by Sellers, Holdco and Company. At or prior to the Third Closing, Sellers, Holdco and Company shall deliver or cause to be delivered to Purchasers:

(a) <u>First Closing Note and First Closing Shares</u>. The First Closing Note and the First Closing Shares.

(b) <u>Stock Certificates</u>. A certificate or certificates representing the Purchased Shares, the Earnout Shares, the Second Closing Shares and the Depreciable Assets Shares duly endorsed in blank for transfer to Purchasers or accompanied by stock powers duly executed in blank; and

(c) <u>Other Documentation</u>. Such other certificates, instruments or documents required pursuant to the provisions of this Agreement or otherwise necessary or appropriate to transfer the Purchased Assets to be transferred at the Third Closing in accordance with the terms hereof and consummate the Transaction, and to vest in Purchasers and its successors and assigns full, complete, absolute, legal and equitable title to the Purchased Assets to be transferred at the Third Closing, free and clear of all Encumbrances (other than Permitted Encumbrances), including such certificates, instruments and documents to be executed or delivered by Company, Holdco or Sellers pursuant to the terms of this Agreement.

**2.6** Third Closing Deliveries by Purchasers. At or prior to the Third Closing, Purchasers shall deliver or cause to be delivered to Holdco and Sellers:

- (a) <u>CLNE Shares</u>. the CLNE Shares to be issued at the Second Closing;
- (b) <u>Promissory Note</u>. The Year 1 Note; and

the Finco Note.

(c) <u>Wire Transfer</u>. A wire transfer to HSBC Bank Canada for credit to Sellers' account, in an amount equal to the principal amount of

15

#### ARTICLE III

#### REPRESENTATIONS AND WARRANTIES REGARDING COMPANY AND COMPANY SUBSIDIARIES

Sellers, Company and Holdco jointly and severally represent and warrant to Purchasers that the statements contained in this <u>Article III</u> are correct and complete as of the date of this Agreement and will be correct and complete as of the Closing Date (as though made then and as though the Closing Date were substituted for the date of this Agreement throughout this <u>Article III</u> except to the extent such representations and warranties expressly speak as of an earlier date), except as specifically set forth in the disclosure schedule delivered by Sellers, Company and Holdco to Purchasers on the date hereof (the "Company Disclosure Schedule"). Nothing in the Company Disclosure Schedule shall be deemed adequate to disclose an exception to a representation or warranty made herein, however, unless the applicable representation or warranty is identified in the Company Disclosure Schedule and the Company Disclosure Schedule identifies the exception with reasonable particularity and describes the relevant facts in reasonable detail. Without limiting the generality of the foregoing, the mere listing (or inclusion of a copy) of a document or other item shall not be deemed adequate to disclose an exception to a representation or warranty made herein (unless the representation or warranty pertains to the existence of the document or other item itself). The Company Disclosure Schedule will be arranged in paragraphs corresponding to the lettered and numbered paragraphs contained in this <u>Article III</u>.

3.1 Organization, Good Standing, Qualification. <u>Schedule 3.1(i)</u> sets forth Company's jurisdiction of organization and each province, state or other jurisdiction in which Company is qualified to do business. <u>Schedule 3.1(ii)</u> sets forth the jurisdiction of organization and each state or other jurisdiction in which any Company Subsidiary is qualified to do business. Company and each Company Subsidiary: (i) is a corporation incorporated, duly organized, validly existing and in good standing under the laws of its jurisdiction of organization; (ii) is duly qualified to conduct business and is in corporate and tax good standing under the laws of each jurisdiction in which the nature of its business, the operation of its assets or the ownership or leasing of its properties requires such qualification; and (iii) has full corporate power and authority required to own, lease and operate its assets and to carry on its Business as now being conducted and as presently proposed to be conducted.

#### 3.2 Charter Documents; Books and Records.

(a) Company has delivered to Purchasers accurate, correct and complete copies of (i) the certificate of incorporation, notice of articles, articles (or similar charter documents) of Company and each Company Subsidiary, including all amendments thereto, as presently in effect; (ii) all share records of each Company Subsidiary, including each Company Subsidiary's share ledger and copies of any share certificates issued by any Company Subsidiary; (iii) all minutes and other records of all meetings and other proceedings (including any actions taken by written consent or otherwise without a meeting) of the shareholders of Company and each Company Subsidiary, Company's Board of Directors and each Company Subsidiary's Board of Directors and each Company and each Company Subsidiary; and (iv) all books of account and other financial records of Company and each Company Subsidiary.

(b) The minute books of Company and each Company Subsidiary accurately and completely reflect all material corporate actions of such Entities' shareholders, Board of Directors and any committees.

(c) (i) Company is not in violation of any of the provisions of its certificate of incorporation, notice of articles, articles (or similar charter documents) or Resolutions, and, to the Knowledge of Company, no condition or circumstance exists that likely would (with or without notice or lapse of time) constitute or result directly or indirectly in such a violation; and (ii) no Company Subsidiary is in violation of any of the provisions of its certificate of incorporation, notice of articles, articles (or similar charter documents) or Resolutions, and, to the Knowledge of Company, no condition or circumstance exists that likely would (with or without notice or lapse of incorporation, notice of articles, articles (or similar charter documents) or Resolutions, and, to the Knowledge of Company, no condition or circumstance exists that likely would (with or without notice or lapse of time) constitute or result directly or indirectly in such a violation.

#### 3.3 Reserved.

3.4 Authority; Binding Nature of Agreements. Company has all requisite corporate power and authority to execute and deliver this Agreement and all other Transaction Agreements to which Company is a party and to carry out the provisions of this Agreement and the other Transaction Agreements. The execution, delivery and performance by Company of this Agreement and the other Transaction Agreements have been approved by all requisite action on the part of Company. This Agreement has been duly and validly executed and delivered by Company. Each of this Agreement and the other Transaction Agreements constitutes, or upon execution and delivery, will constitute, the legal, valid and binding obligation of Company, enforceable against Company in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws and equitable principles related to or limiting creditors' rights generally and by general principles of equity.

**3.5 No Conflicts; Required Consents.** No Consents other than those set forth in <u>Schedule 3.5</u> are required with respect to Company's, Holdco's and Sellers' execution and delivery of this Agreement, the other Transaction Agreements and the consummation of the Transaction. The execution, delivery and performance of this Agreement or any other Transaction Agreement by Company do not and will not (with or without notice or lapse of time):

(a) conflict with, violate or result in any breach of (i) any of the provisions of Company's or any Company Subsidiary's certificate of incorporation, notice of articles, articles (or similar charter documents); (ii) any Resolutions; (iii) any of the terms or requirements of any Governmental Approval held by Company or any Company Subsidiary or any of their respective employees; or (iv) any provision of any Company Contract;

(b) give any Governmental Authority or other Person the right to (i) challenge the Transaction; (ii) exercise any remedy or obtain any relief under any Legal Requirement or any Order to which Company or any Company Subsidiary is subject; (iii) declare a default of, exercise any remedy under, accelerate the performance of, cancel, terminate, modify or receive any payment under any Company Contract; or (iv) revoke, suspend or modify any Governmental Approval;

(c) result in the imposition or creation of any Encumbrance upon or with respect to any assets of Company or any Company Subsidiary (other than any Encumbrances created in connection with the Year 1 Note and the Future Payment Notes); or

(d) except for applicable requirements, if any, under any Antitrust Law, require Company or any Company Subsidiary to obtain any Consent or make or deliver any filing or notice to a Governmental Authority.

3.6 Subsidiaries. Schedule 3.6 sets forth for each Subsidiary of Company (each a "Company Subsidiary" and collectively, the "Company Subsidiaries") (i) its name and jurisdiction of incorporation, (ii) the number of authorized shares for each class of its capital shares, (iii) the number of issued and outstanding shares of each class of its capital shares, the names of the holders thereof, and the number of shares held by each such holder, and (iv) the number of its capital shares held in treasury. All of the issued and outstanding capital shares of each Company Subsidiary have been duly authorized and are validly issued, fully paid, and non-assessable, and have been issued in compliance with all applicable Legal Requirements, including applicable securities Legal Requirements. Company or one or more Company Subsidiaries hold of record and own beneficially all of the outstanding shares of each Company Subsidiary, free and clear of any restrictions on transfer (other than restrictions under applicable Legal Requirements), Taxes, Encumbrances, options, warrants, purchase rights, Contracts, commitments, equities, claims, and demands. There are no outstanding or authorized options, warrants, purchase rights, subscription rights, conversion rights, exchange rights, or other contracts or commitments that could require Company or any Company Subsidiaries to sell, transfer, or otherwise dispose of any capital shares of any of Company Subsidiary or that could require any Company Subsidiary to issue, sell, or otherwise cause to become outstanding any of its own capital shares. There are no outstanding share appreciation, phantom share, profit participation, or similar rights with respect to any Company Subsidiary. There are no voting trusts, proxies, or other agreements or understandings with respect to the voting of any capital shares of any Company Subsidiary. Neither Company nor any Company Subsidiary controls directly or indirectly or has any direct or indirect equity participation in any corporation, partnership, trust, or other business association that is not a Company Subsidiary. Neither Company nor any Company Subsidiary owns or has any right to acquire, directly or indirectly, any outstanding capital shares of, or other equity interests in, any Person. The Purchased Shares constitute all of the outstanding capital shares and any other outstanding equity or ownership interests of the Company Subsidiaries.

#### 3.7 Financial Statements.

(a) <u>Company Financial Statements</u>. Company has previously delivered to Purchasers the following financial statements (the "*Company Financial Statements*"): (i) the audited balance sheets, and the related statements of operations, changes in shareholders' equity, and cash flows, of Company as of and for the fiscal years ended April 30, 2009, 2008 and 2007, together with the notes thereto, and (ii) the unaudited balance sheets, and the related unaudited statements of operations, changes in shareholder's equity, and cash flows, of Company as of and for the twelve months ended April 30, 2010.

(b) <u>IMW Colombia Financial Statements</u>. Company has previously delivered to Purchasers the following financial statements (the "*IMW Colombia Financial Statements*"): (i) the audited balance sheets, and the related statements of operations, changes in shareholders' equity, and cash flows, of IMW Colombia Ltda. ("*IMW Colombia*") as of and for the fiscal years ended December 31, 2009, 2008 and 2007, together with the notes thereto, and (ii) the unaudited balance sheets, and the related unaudited statements of operations, changes in shareholder's equity, and cash flows, of IMW Colombia as of and for the four months ended April 30, 2010.

<sup>17</sup> 

(c) <u>IMW China Financial Statements</u>. Company has previously delivered to Purchasers the following financial statements (the "*IMW China Financial Statements*"): the audited balance sheets, and the related statements of operations, changes in shareholders' equity, and cash flows, of IMW Compressors (Shanghai) Co., Ltd. ("*IMW China*") as of and for the fiscal years ended December 31, 2009, 2008 and 2007, together with the notes thereto.

(d) <u>IMW Bangladesh Financial Statements</u>. Company has previously delivered to Purchasers the following financial statements (the "*IMW Bangladesh Financial Statements*"): (i) the unaudited (externally prepared) balance sheets, and the related statements of operations, changes in shareholders' equity, and cash flows, of IMW CNG Bangladesh Ltd. ("*IMW Bangladesh*") as of and for the fiscal years ended June 30, 2009, 2008 and 2007, together with the notes thereto, and (ii) the unaudited balance sheets, and the related unaudited statements of operations, changes in shareholder's equity, and cash flows, of IMW Bangladesh as of and for the ten months ended April 30, 2010. The Company Financial Statements, IMW Colombia Financial Statements, the IMW China Financial Statements and the IMW Bangladesh Financial Statements are collectively referred to herein as the "*Financial Statements*").

(e) All of the Financial Statements (i) are true, accurate and complete in all respects; (ii) are consistent with the books and records of Company and the Company Subsidiaries; (iii) present fairly and accurately the financial condition of Company and the Company Subsidiaries as of the respective dates thereof and the results of operations, changes in shareholder's equity and cash flows of Company and the Company Subsidiaries for the periods covered thereby; and (iv) have been prepared in accordance with Canadian GAAP, applied on a consistent basis throughout the periods covered; provided, however, that the interim balance sheets are subject to year-end adjustments consistent with past practice (which will not be material individually or in the aggregate) and do not contain all of the footnotes required by Canadian GAAP. All reserves established by Company and the Company Subsidiaries and set forth in the Financial Statements are adequate for the purposes for which they were established. Company and the Company Subsidiaries have delivered to Purchasers copies of all letters from Company's and the Company Subsidiaries' auditors delivered to Company and the Company Subsidiaries during the past thirty-six (36) months together with copies of all responses thereto.

(f) <u>Schedule 3.7(f)</u> sets forth an accurate, correct and complete breakdown and aging of each of Company's and each Company Subsidiary's accounts payable (including to all of its suppliers) as of April 30, 2010.

19

**3.8 Absence of Undisclosed Liabilities**. Neither Company nor any Company Subsidiary has any Liabilities other than (i) those set forth in the Financial Statements; and (ii) those incurred in the ordinary course of business since December 31, 2009 which are not individually or in the aggregate material in nature.

**3.9 Absence of Changes**. Since December 31, 2009, (i) Company and the Company Subsidiaries have conducted the Business in the ordinary course of business and (ii) no event or circumstance has occurred that could reasonably have a Material Adverse Effect on Company or the Business. Without limiting the generality of the foregoing, since December 31, 2009, neither Company nor any Company Subsidiary has:

(a) acquired by merging, amalgamating, arranging or consolidating with, or by purchasing any equity securities or assets of, or by any other manner, any business or any Entity;

(b) sold, transferred, leased, licensed or otherwise encumbered any of its assets, except in the ordinary course of business consistent with past practice;

(c) entered into any Contracts, except on commercially reasonable terms in the ordinary course of business;

(d) violated any applicable Legal Requirement;

(e) changed or announced any change to the Company Products;

(f) violated, terminated or amended any Company Contract or Governmental Approval;

(g) commenced any Proceeding;

(h) issued, sold, contracted to issue or sell, pledged, disposed of, granted, encumbered, spilt, reclassified or authorized the issuance, sale, pledge, disposition, grant, split, reclassification or encumbrance of (1) any equity or similar interest in any Company Subsidiary or (2) any options, warrants, convertible securities or other rights of any kind to acquire any equity or similar interest in any Company Subsidiary;

(i) purchased, leased, licensed or otherwise acquired any assets, except for supplies acquired by the Company or any Company Subsidiary in the ordinary course of business;

- (j) made any capital expenditure in excess of \$100,000, individually or in the aggregate;
- (k) incurred any Indebtedness;

(1) provided any credit, loan, advance, guaranty, endorsement, indemnity, warranty or mortgage to any Person, including any of the customers, shareholders, officers, employees or directors of Company, other than those made in the ordinary course of business;

20

(m) changed its credit practices, accounting methods or practices or standards used to maintain its books, accounts or business

records;

(n) changed the terms of its Accounts Payable, Accounts Receivable or Subsidiary Accounts Receivable or taken any action directly or indirectly to cause or encourage any acceleration or delay in the payment, collection or generation of its accounts, Accounts Receivable or Subsidiary

#### Accounts Receivable;

(o) incurred or become subject to any Liability or settled any Liability, contingent or otherwise, except current Liabilities in the ordinary course of business;

(p) amended its certificate of incorporation, notice of articles, articles or similar charter documents;

(q) accelerated, amended or changed the period of exercisability or vesting of any options or rights to acquire any equity or similar ownership interest in Company or other rights granted under any Company Benefit Plans, except as specifically contemplated in this Agreement;

(r) hired any new employee other than in the ordinary course of business; terminated any officer or key employee of Company; increased the annual level of compensation of any existing employee other than in the ordinary course of business and consistent with past practice; established or adopted any Company Benefit Plan; or granted any bonuses, benefits or other forms of direct or indirect compensation to any employee, officer, director or consultant other than in the ordinary course of business and consistent with past practice;

(s) made any severance payments to any employee, officer or director or entered into any Company Contracts that contemplate such severance payments;

(t) with respect to any Company Subsidiary, made, amended, or revoked any election relating to Taxes; adopted or changed any accounting method relating to Taxes; filed any amendment to any Tax Return; entered into any Tax sharing, allocation, indemnity or similar agreement; entered into any closing agreement; settled or compromised any claim or assessment relating to Taxes; consented to any extension or waiver of the limitations period applicable to any Taxes or Tax Returns; or

(u) entered into any Contract or otherwise agreed to take any of the actions described in clauses (a) through (t) above.

**3.10** Accounts Receivable. The Accounts Receivable and Subsidiary Accounts Receivable are (i) valid and legally binding obligations of the account debtor enforceable in accordance with its terms, free and clear of all Encumbrances, and not subject to setoffs, adverse claims, counterclaims, assessments, defaults, prepayments, defenses, and conditions precedent; (ii) a true and correct statement of the account for merchandise actually sold and delivered to, or for services actually performed for and accepted by, such account debtor; and (iii) fully collectible and will be collected within sixty (60) days.

21

**3.11 Indebtedness**. <u>Schedule 3.11</u> lists all Indebtedness of Company or any Company Subsidiary, setting forth as to each item the principal amount outstanding, the per annum interest rate and the maturity date. All such Indebtedness is reflected on the Financial Statements and Company is not in breach (or has received notice of breach or default) or default under any of the terms or conditions set forth in any loan document or other document or instrument related thereto. All such Indebtedness is prepayable at any time without penalty or premium at the option of Company, and neither the execution of this Agreement nor the consummation of the Transaction will result in any penalty or incurrence of any additional obligation or change of any terms with respect to any such Indebtedness.

**3.12 Inventory**. All items in Company's Inventory consist of items of a quality, quantity and condition usable and salable in the ordinary course of Company's business without discount or reduction, conform to generally accepted standards in the industry of which Company is a part, are free of material defects, materially meet Company's current standards and specifications and comply in all material respects with applicable Legal Requirements, are at levels consistent with past practices of the business of Company, and are physically present without damage, loss or shrinkage. The value of the Inventory reflected on the Financial Statements was in each instance, (a) prepared from Company's books and records in accordance with Canadian GAAP applied consistently with past practices and correctly valued on such balance sheet, (b) not subject to any write-down or write-off in excess of the Inventory reserves stated in the Financial Statements and (c) net of adequate reserves for damaged, excess, slow moving, obsolete and unsalable items.

#### 3.13 Material Contracts.

(a) <u>Schedule 3.13</u> sets forth an accurate, correct and complete list (specifying the applicable subsection of this <u>Section 3.13(a)</u> to which such disclosure applies) of all Contracts to which Company or any Company Subsidiary is a party ("*Company Contracts*") to which any of the descriptions set forth below may apply (the "*Material Contracts*"):

(i) Contracts relating to profit sharing, share option, employee share purchase or other plan or arrangements providing for deferred or other compensation (including any bonuses or other remuneration and whether in cash or otherwise), to employees, former employees or consultants, or any other employee benefit plan or arrangement, or any collective bargaining agreement or any other agreement with any labor union, or severance agreements, programs, policies or arrangements;

(ii) Contract for the employment by Company or any Company Subsidiary of any officer, individual employee or other Person on a full-time, part-time, consulting or other basis involving annual compensation in excess of \$100,000 or Contracts relating to loans by Company or any Company Subsidiary to officers, directors, shareholders or Affiliates of Company or any Company Subsidiary;

(iii) Contract or indenture relating to borrowed money or other Indebtedness or the mortgaging, pledging or otherwise placing an Encumbrance on any material asset or group of assets of Company or any Company Subsidiary;

(iv) Contracts under which Company or any Company Subsidiary is lessee of or holds or operates any personal property owned by any other Person, except for any lease of personal property under which the aggregate annual rental payments do not exceed \$20,000;

any property, real or perso	(v) onal, owne	Contracts under which Company or any Company Subsidiary is lessor or that permits any third party to hold or operate d or controlled by Company or any Company Subsidiary;
involves consideration in	(vi) the aggreg	Contract or group of related Contracts with the same party or group of affiliated parties the performance of which ate in excess of \$100,000 annually;
third party Intellectual Pro	(vii) operty that	Contracts that constitute a license or sublicense to which Company or any Company Subsidiary is authorized to use any is material to the Business, excluding generally commercially available, off-the-shelf software programs;
Asset that is material to the	(viii) ne Busines	Contracts that constitute a license or sublicense pursuant to which any third party is authorized to use any Company IP s;
collaboration with respect	(ix) t to, any Co	Contracts relating to the research, development, clinical trial, supply, manufacture, marketing or co-promotion of, or ompany Product;
	(x)	Contracts relating to sales, distribution, warehousing, logistics, shipping, supply or franchise arrangements;
without penalty and invol	(xi) ve conside	Contracts with a term of more than six months which are not terminable by Company upon less than 30 days' notice ration in excess of \$100,000 annually;
the world;	(xii)	Contracts prohibiting Company or any of its Affiliates from freely engaging in any business or competing anywhere in
	(xiii)	any other Contract which is material to Company or the Business;
pursuant to <u>Section 3.25;</u>	(xiv) or	any Contract with any of the customers or suppliers of Company or any Company Subsidiary required to be disclosed
	(xv)	any amendment, supplement or modification (whether written or oral) of any of the foregoing.
	<b>C</b>	

(b) Company has delivered to Purchasers accurate, correct and complete copies of all Material Contracts, including all amendments, supplements, modifications and waivers thereof. All Material Contracts are in writing. All Company Contracts that are not Material Contracts do not, in the aggregate, represent a material portion of the Liabilities of Company.

23

(c) Each Material Contract is currently valid and in full force and effect, and is enforceable by Company in accordance with its terms.

(d) Neither Company nor any Company Subsidiary is in breach, default, and no party has notified Company or any Company Subsidiary that it is in breach, default, under any Material Contract. No event has occurred, and no circumstance or condition exists, that might (with or without notice or lapse of time) (a) result in a violation or breach of any of the provisions of any Material Contract; (b) give any Person the right to declare a default or exercise any remedy under any Material Contract; (c) give any Person the right to accelerate the maturity or performance of any Material Contract or to cancel, terminate or modify any Material Contract; or (d) otherwise have a Material Adverse Effect on Company, the Business or any Company Subsidiary in connection with any Material Contract. Neither Company nor any Company Subsidiary has waived any of its rights under any Material Contract. To the Knowledge of Company, no party to any Material Contract other than Company or any Company Subsidiary is in breach or default under such Material Contract.

(e) The performance of the Material Contracts will not result in any violation of or failure by Purchasers, Company or any Company Subsidiary to comply with any Legal Requirement.

(f) The Material Contracts constitute all of the Contracts that Company and the Company Subsidiaries are party to in order to conduct the Business in the manner in which such Business is currently being conducted by Company and the Company Subsidiaries.

(g) The assignment to Purchasers of any of the Company Contracts shall not result in Purchasers being bound by, or subject to, any non-compete or other restriction on the operation or scope if its businesses, including the Business.

#### 3.14 Intellectual Property.

(a) <u>Schedule 3.14(a)</u> sets forth an accurate and complete list of all Patents, registered Trademarks, pending applications for registrations of any Trademarks, including ITU Applications, unregistered Trademarks, registered Copyrights, and pending applications for registration of Copyrights, owned or filed by or on behalf of Company or any Company Subsidiary ("*Registered IP Assets*") and other Patents within the Company IP Assets. <u>Schedule 3.14(a)</u> specifies as to each the nature of such item, lists the jurisdictions in which each such item has been issued or registered or in which any such application for such issuance and registration has been filed, and indicates any applicable registration or application number.

(b) All necessary registration, maintenance and renewal fees in connection with the Registered IP Assets have been paid and all necessary documents and certificates in connection with such Registered IP Assets have been filed with the relevant patent, copyright, trademark or other authorities in the United States or foreign jurisdictions, as the case may be, for the purposes of maintaining such Registered IP Assets. There are no actions that must be taken by Company or any Company Subsidiary within one hundred twenty (120) days following the Closing Date, including the payment of any registration, maintenance or renewal fees or the filing of any responses to office actions, documents, applications or certificates for the purposes

of obtaining, maintaining, perfecting, preserving or renewing any Registered IP Assets. To the maximum extent provided for by, and in accordance with, applicable Legal Requirements, Company or a Company Subsidiary has recorded in a timely manner each such assignment of a Registered IP Asset assigned to Company or any Company Subsidiary with the relevant governmental authority, including the United States Patent and Trademark Office (the "*PTO*"), the U.S. Copyright Office or their respective counterparts in any relevant foreign jurisdiction, as the case may be.

(c) Company has no Knowledge (i) of any facts, circumstances, or information that would (x) render any of the Company IP Assets invalid or unenforceable; (y) would adversely affect any pending application for any of the Registered IP Assets; or (z) would adversely affect, limit, restrict, impair, or impede the ability of Company to use and practice the Company IP Assets upon and after the Closing on an exclusive basis, to conduct the Business in the manner in which the Business is currently being conducted or is planned to be conducted by Company and the Company Subsidiaries; or (ii) that any Company IP Assets are involved in any interference, reexamination, cancellation, or opposition proceeding, or any other currently pending or threatened Proceeding challenging the ownership, use, validity, or enforceability of, any Company IP Assets. The Company IP Assets are valid, subsisting, in full force and effect, have not been abandoned or passed into the public domain, and were prosecuted in good faith.

(d) Company and the Company Subsidiaries have taken reasonable steps sufficient to safeguard and maintain the secrecy and confidentiality of and its proprietary rights in all of the Company IP Assets not otherwise protected by patents, patent applications, or copyright or trademark law. Company has no Knowledge of any misappropriation or unauthorized disclosure of any Confidential Information related to Company, any Company Subsidiary or the Business or breach of obligations of confidentiality with respect to such. No present or former employee of Company or any Company Subsidiary has any right, title or interest, directly or indirectly, in whole or in part, in any Company IP Assets, including by virtue of any Moral Right or right with respect to publicity, privacy, name or identity. In each case in which Company or any Company Subsidiary has acquired ownership of any Company IP Assets from any Person, including any officer or employee, Company or a Company Subsidiary has obtained a valid and enforceable assignment sufficient to irrevocably transfer all rights, and a wavier of any and all non-assignable rights, including Moral Rights, in such Company IP Assets (including the right to sue for all past, present, and future infringements, unauthorized uses or disclosures, or misappropriations with respect thereto) to Company or a Company Subsidiary. To the Knowledge of Company, no employee, consultant or independent contractor of Company or any Company Subsidiary is, as a result of or in the course of such employee's, consultant's or independent contractor's engagement by Company or any Company Subsidiary in default or breach of any material term of any employment agreement, non-disclosure agreement, assignment of invention agreement, or similar agreement.

(e) <u>Schedule 3.14(e)</u> sets forth a complete and accurate list of all Company Contracts (i) granting any Intellectual Property License, (ii) containing a covenant either not to compete or not challenge the validity or ownership of the Intellectual Property Rights of a third party or otherwise limiting Company's or any Company Subsidiary's ability to (x) exploit fully any of the Company IP Assets or (y) conduct the Business in any market or geographical area or with any Person, or (iii) under which Company or any Company Subsidiary has agreed to, or

25

assumed, any obligation or duty to warrant, indemnify, reimburse, hold harmless, guaranty or otherwise assume or incur any obligation or liability, or provide a right of rescission, with respect to the infringement, unauthorized use or disclosure, or misappropriation by Company, any Company Subsidiary or other Person of the Intellectual Property Rights of any Person other than Company. None of the parties to any such Contract is in breach thereof, and there is no dispute under any such Contract regarding the scope of such Contract, or performance under such Contract, including with respect to any payments to be made or received by Company or any Company Subsidiary thereunder. No Contract grants to or authorizes the retention of any joint ownership interest or exclusive license of, or other exclusive right with respect to, any of the Company IP Assets by any Person.

(f) Company is the sole and exclusive owner of all right, title and interest in and to all of the Registered IP Assets, and each of the other Copyrights in any works of authorship prepared by or for Company or any Company Subsidiary that resulted from or arose out of any work performed by or on behalf of Company or any Company Subsidiary or by any employee, officer, consultant or contractor of any of them which are Company IP Assets. Each of the Company IP Assets is either (i) solely and exclusively owned by Company or a Company Subsidiary and has not been exclusively licensed to any other Person, or (ii) is duly and validly licensed to Company or a Company Subsidiary for use and practice in the manner in which it is being used and practiced or is planned to be used and practiced in the conduct of the Business by Company and the Company Subsidiaries. Upon the Closing, each of the Company IP Assets will be owned by Purchasers or will be immediately available for use and practice by Purchasers on terms and conditions identical to those under which Company or a Company Subsidiary currently uses or practices such Company IP Assets, and fully assignable and transferable, alienable and licensable by Purchasers, without any affirmative act by Company, any Company Subsidiary, Purchasers or any other Person, including any assignment, release, Consent, approvals, immunities or other rights not yet obtained or payment of any kind. Such ownership and right to use and practice are (and upon the Closing, will be) free and clear of, and without liability under, any adverse claim of any kind, including any Encumbrance. Other than as set forth on <u>Schedule 3.14(f)</u>, no Person other than Purchasers shall have any right, title and interest in the Company IP Assets, including by virtue of any Intellectual Property License, immediately following the Closing.

(g) Neither the execution, delivery, and performance of this Agreement nor the consummation of the transactions contemplated hereby will (i) conflict with, result in a breach of or default under, modify, accelerate, or terminate, or give rise to a right to modify, accelerate, or terminate, any Intellectual Property License, (ii) will cause the forfeiture or termination or give rise to a right of forfeiture or termination of or result in any loss of, or the diminishment in value of, any Company IP Assets, including by in any way impairing the right of Purchasers to use or practice or bring any action for the infringement, unauthorized use or disclosure, or misappropriation of any Company IP Assets, (iii) result in (x) Purchasers, Company or any Company Subsidiary granting to any Person any right to, or with respect to, any of the Intellectual Property Rights owned by, or licensed to, Company, any Company Subsidiary, Purchasers or any of their respective Affiliates; (y) Company, any Company Subsidiary, Purchasers or any of their respective Affiliates; (y) Company, any Company Subsidiary, Purchasers or any of their respective Affiliates being bound by, or subject to, any non-compete or other restriction on the operation or scope of its businesses, including the Business; or (z) Company, or any Company Subsidiary, Purchasers or any of their respective Affiliates being obligated to pay any royalties or other amounts to any Person. No Company IP Assets or Products are subject to any Proceeding, Order, stipulation or Contract that restricts in any manner the use, transfer or licensing thereof by Company or any Company Subsidiary or that may affect the validity, use or enforceability of such Company IP Assets.

(h) The conduct of the Business in the manner in which the Business is currently being conducted or is planned to be conducted by Company and Company Subsidiaries, including but not limited to the design, development, use, import, export, branding, advertising, promotion, marketing, manufacture and sale of products (including any currently under development) and services (including manufacturing and process development services) in connection with the Business, does not and will not, and will not when operated substantially in the same manner following the Closing, constitute infringement, unauthorized use or disclosure, or misappropriation of any Intellectual Property Rights or Confidential Information of any Person, violate any right of any Person (including any Moral Right or Publicity Right), defame or libel any Person or constitute unfair competition or trade practices under the laws of any jurisdiction, and neither Company nor any Company Subsidiary has received notice from any Person claiming that such operation or any Company Product in connection with the Business infringes, uses or discloses without authorization, or misappropriates any Intellectual Property Rights or Confidential Information of any Person (including any Moral Right or Publicity Right), invites the taking of a license, authorization, covenant not to sue or the like under any Intellectual Property Rights or Confidential Information, or defames or libels any Person or constitutes unfair competition or trade practices under the laws of any jurisdiction (nor does Company have Knowledge of any basis therefor). Neither Company nor any Company Subsidiary is party to any pending or, to the Knowledge of Company, threatened Proceeding which involves a claim of infringement, unauthorized use or disclosure, or misappropriation or otherwise arising out of a right or claimed right of any Person with respect to any Intellectual Property Rights or Confidential Information.

(i) The Company IP Assets owned by or licensed to Company and included in the Purchased Assets include all of the Intellectual Property Rights and Technology that are used to conduct the Business by Company and the Company Subsidiaries.

(j) Except with respect to licenses of commercial off-the-shelf software or under a license listed in <u>Schedule 3.14(j)</u>, neither Company nor any Company Subsidiary is required, obligated, or under any liability whatsoever, to make any payments by way of royalties, fees or otherwise to any owner, licensor of, or other claimant to any Company IP Assets, or other Person, with respect to the use or practice thereof or in connection with the conduct of the Business in the manner in which the Business is currently being conducted or is planned to be conducted by Company and Company Subsidiaries.

(k) To the Knowledge of Company, no Person is infringing, using or disclosing without authorization, or misappropriating any Company IP Assets owned by or exclusively licensed to Company or any Company Subsidiary. Neither Company nor any Company Subsidiary has made any such claims against any Person with respect to any Company IP Assets, and neither Company nor any Company Subsidiary has invited any Person to take a license, authorization, covenant not to sue or the like with respect to any Company IP Assets. Each current employee of Company and each Company Subsidiary has signed the form of non-disclosure agreement set forth in <u>Schedule 3.14(k)</u>.

27

(1) Company and the Company Subsidiaries have complied with all applicable Legal Requirements and with their respective published privacy policies relating to privacy and data security, including with respect to the collection, use, disclosure and transfer of Personally-Identifiable Information. To the Knowledge of Company, there have been no unauthorized or accidental accesses, acquisitions, disclosures, intrusions or breaches of security of Personally-Identifiable Information maintained by or on behalf of Company or any Company Subsidiary, nor any complaints, notices to, or claims asserted by any person or entity (including any Governmental Authority) regarding the collection, use, transmission, disclosure or sharing of Personally-Identifiable Information by Company or any Company Subsidiary.

(m) No funding, facilities, or personnel of any Governmental Authority were used, directly or indirectly, to develop or create, in whole or in part, any Company IP Assets.

(n) Neither Company nor any Company Subsidiary is or ever was a member or promoter of, or a contributor to, any industry standards setting body or similar organization in connection with which it has become obligated to grant or offer to any other Person any license or right to any Company IP Assets.

#### 3.15 Employees and Consultants.

(a) <u>Employment Status</u>. No employee of Company or any Company Subsidiary has been granted the right to continued employment by Company, any Company Subsidiary or any successors of the foregoing. To the Knowledge of Company, no officer, director, employee or independent contractor of Company or the Company Subsidiaries (collectively, the "*Workers*") intends to terminate his or her employment or other relationship with Company, nor does Company have a present intention to terminate the employment or other relationship of any Worker. <u>Schedule 3.15(a)</u> lists all employee manuals and handbooks and policy statements relating to the employment or other relationship of Workers with Company and the Company Subsidiaries as of May 31, 2010 and Company has delivered to Purchasers accurate and complete copies of the same.

(b) <u>Compensation</u>. <u>Schedule 3.15(b)</u> sets forth an accurate and complete list of all (i) employees of Company and the Company Subsidiaries, including each employee's name, title or position, present annual compensation (including bonuses, commissions and deferred compensation), accrued and unused paid vacation and other paid leave, years of service, interests in any incentive compensation plan, vested and unvested equity interests, and estimated entitlements to receive supplementary retirement benefits or allowances (whether pursuant to a contractual obligation or otherwise), (ii) individuals who are currently performing services for Company or a Company Subsidiary who are classified as independent contractors, including the respective compensation of each consultant or independent contractor, and (iii) any employees or independent contractors of an Affiliate of Company that perform services for Company or a Company Subsidiary. <u>Schedule 3.15(b)</u> sets forth all (i) bonuses, severance payments, termination pay, equity interests, and other compensation or benefits of any kind (whether pursuant to an oral or written agreement or understanding, statute, or otherwise) paid to, accrued

with respect to, any present or former Worker since December 31, 2009; (ii) increases in any employee's wage or salary since December 31, 2009 (other than increases in the ordinary course of business consistent with past practice) or (iii) increases or changes in any other benefits or insurance provided to any employees since December 31, 2009 (other than increases or changes in the ordinary course of business consistent with past practice) or (iii) increases or business consistent with past practice). Company has listed in <u>Schedule 3.15(b)</u> and delivered to Purchasers accurate and complete copies of all plans, programs, policies, letters, agreements, Contracts, and other arrangements or understandings between Company or any Company Subsidiary and any Workers under which Company or a Company Subsidiary may be obligated to provide to any Worker notice, pay in lieu of notice, severance or any other compensation or benefits as a result of the transaction contemplated by this Agreement or upon termination of employment or any other relationship with Company or any Company Subsidiary. All independent contractors have been properly classified under applicable Legal Requirements.

(c) <u>Continuing Benefits</u>. There is no former employee of Company or any Company Subsidiary who is receiving or is scheduled to receive (or whose spouse or other dependent is receiving or is scheduled to receive) any benefits (whether from Company or otherwise) relating to such former employee's employment with Company or any Company Subsidiary. Since January 1, 2003, neither Company nor any Company Subsidiary has been a party to or bound by any Contract that creates or grants to any Person, or provides for the creation or grant of, any share appreciation right, phantom share right or similar right or interest.

(d) Unions. Neither Company nor any Company Subsidiary is or, to the Knowledge of the Company, ever has been a party to or otherwise bound by any collective bargaining or other agreements with any trade union, council of trade unions, employee association or other labor organization and no such agreement is presently being negotiated. Since January 1, 2003, there has not been a representation vote respecting any of the employees of Company or any Company Subsidiary and, to the Knowledge of Company, there are no union organizing campaigns being conducted to authorize representation by any trade union, employee association or other labor organization. No trade union, employee association or other labor organization holds bargaining rights with respect to any Workers by way of certification, interim certification, voluntary recognition or succession rights, applied to be certified as the bargaining agent of any of the Workers or applied to have the Company or any Company Subsidiary declared a common or related employer. Since January 1, 2003, there has not been any slowdown, work stoppage, lockout, strike, or other labor dispute or union organizing activity or dispute, affecting Company or any of its Workers while performing services for Company or any Company Subsidiary. There is not now pending, and to the Knowledge of Company no Person has threatened to commence, any such slowdown, work stoppage, lockout, strike, or other labor dispute or union organizing activity or any similar activity or dispute, nor has any event occurred, nor does any condition or circumstance exist, that likely would directly or indirectly give rise to or provide a basis for the commencement of any such slowdown, work stoppage, lockout, strike, or other labor dispute or union organizing activity or any similar activity or dispute. Neither Company nor any Company Subsidiary has engaged in any unfair labor practices and no unfair labor practices complaint, grievance or arbitration proceeding is pending or threatened against the Company or a

29

(e) <u>Disputes</u>. There are no claims, disputes, grievances, or controversies pending or, to the Knowledge of Company, threatened involving any Worker or group of Workers. There are no charges, investigations, administrative proceedings or formal complaints of discrimination (including discrimination based upon sex, age, family status, marital status, race, place of origin, religion, sexual orientation, physical or mental disability or due to a criminal or summary conviction offence unrelated to employment or intended employment) threatened or pending before any Governmental Authority against Company or any Company Subsidiary pertaining to any Worker.

(f) <u>Compliance with Legal Requirements</u>. Each Worker is in compliance with all applicable visa and work permit requirements. No visa or work permit held by a Worker will expire during the six month period beginning at the date of this Agreement. Since January 1, 2003, Company and the Company Subsidiaries have complied with all Legal Requirements related to the terms and conditions of employment or retention of its Workers, including but not limited to wages and other compensation, overtime requirements, classification of employees and independent contractors under federal and provincial laws, hours of work, leaves of absence, pay equity, immigration, occupational health and safety, workers' compensation, human rights and the payment of employment insurance, Canada Pension Plan and other Taxes. Neither Company nor any Company Subsidiary has any Liability under any Legal Requirements related to employment or retention of any Worker and attributable to an event occurring or a state of facts existing prior to the date of this Agreement.

#### 3.16 Company Benefit Plans.

(a) <u>Schedule 3.16(a)</u> contains a true, correct and complete list of each pension, profit-sharing, savings, retirement, employment, collective bargaining, consulting, severance pay, termination, executive compensation, incentive compensation, deferred compensation, bonus, share purchase, share option, share appreciation, phantom share or other equity-based compensation, change-in-control, retention, salary continuation, vacation, sick leave, disability, death benefit, group insurance, health, welfare, hospitalization, medical, dental, life (including all individual life insurance policies as to which Company or any of its subsidiaries is the owner, the beneficiary or both), employee loan, educational assistance or fringe plan, program, policy, practice, Contract or arrangement, whether written or oral, formal or informal, funded or unfunded, insured or self-insured, registered or unregistered, and any other employee benefit plan, program, policy, practice, agreement, or arrangement (including any funding mechanism therefor now in effect or required in the future as a result of the transaction contemplated by this Agreement or otherwise) (i) under which any current or former employee, director, consultant or independent contractor of Company or any Company Subsidiary has any present or future right to benefits and (ii) that is maintained, sponsor or contribute to, or (iii) with respect to which Company or any Company Subsidiary has any direct or indirect Liability, whether contingent or otherwise (each, a "**Company Benefit Plan**").

30

(b) Company has provided to Purchasers with respect to each applicable Company Benefit Plan true, correct and complete copies of all the Company Benefit Plans as amended, together with all related documentation, including, without limitation, funding and investment management agreements, summary plan descriptions, the most recent actuarial valuations, financial statements and asset statements, all material opinions and memoranda (whether externally or internally prepared) and all material correspondence with all regulatory authorities or other relevant persons. No changes have occurred or are expected to occur which would materially affect the information contained in the actuarial reports, financial statements or asset statements required to be provided to Purchasers pursuant to this provision.

(c) No Company Benefit Plan contains or has ever contained a "defined benefit provision" as such term is defined in subsection 147.1(1) of the *Income Tax Act* (Canada).

(d) All Company Benefit Plans have been established, registered, administered, communicated and invested in accordance with all Legal Requirements. No fact or circumstance exists which could adversely affect the registered status of any such Company Benefit Plan. Neither the Company, nor any of its agents or delegates, has breached any fiduciary obligation with respect to the administration or investment of any Company Benefit Plan.

(e) Company has made all contributions and paid all premiums in respect of each Company Benefit Plan in a timely fashion in accordance with the terms of each Company Benefit Plan and applicable Laws. Company has paid in full all contributions and premiums for the period up to

the Closing Date even though not otherwise required to be paid until a later date or has made full and adequate disclosure of and provision for such contributions and premiums in the Books and Records.

(f) No Company Benefit Plan is a multi-employer pension plan as defined under the provisions of any applicable Legal Requirement, nor has the Company or any Company Subsidiary ever contributed to or been a participating employer under any such multi-employer pension plan.

(g) No Company Benefit Plan provides for retiree benefits or for benefits to retired employees or to the beneficiaries or dependants of retired employees.

(h) Other than in respect of routine claims for benefits, no Company Benefit Plan, no administrator of any Company Benefit Plan, and no member of any body which administers a Company Benefit Plan, nor the Company, is subject to any pending action, investigation, examination, claim (including claims for income taxes, interest, penalties, fines or excise taxes) or any other proceeding initiated by any Person, and there exists no state of facts which could reasonably be expected to give rise to any such action, investigation, examination, claim or other proceeding.

(i) The consummation of the transactions contemplated by this Agreement will not conflict with or result in any violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation, modification or acceleration of any obligation or loss of any benefit under any Company Benefit Plan, trust or loan that will or may result in any payment (whether of severance pay or otherwise), acceleration, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation

31

to fund benefits with respect to any employee, or any other material liability for Company, any Company Subsidiary or Purchasers. Neither the consummation of the transactions contemplated by this Agreement nor any termination of the employment of any of Company's or any Company Subsidiary's employees will result in or give rise to (i) any obligation to make any severance, retention, termination, change of control, or other payments to present or former employees of Company or any Company Subsidiary, other than obligations to provide notice or pay in lieu of notice to current Workers who are not offered employment by Purchasers, or (ii) the acceleration of any other rights or benefits to any present or former employee or consultant of Company or any Company Subsidiary, whether pursuant to an oral or written agreement or understanding, statute, or otherwise.

(j) Each Company Benefit Plan can be amended, terminated or otherwise discontinued after the Closing Date in accordance with its terms, without liability to Company, any Company Subsidiary, or Purchasers. None of the Company Benefit Plans will be subject to any surrender fees, deferred sales charges, commissions, or other fees upon termination other than the normal and reasonable administrative fees associated with their amendment, transfer or termination.

#### 3.17 Compliance with Laws.

(a) Company and the Company Subsidiaries are, and at all times since January 1, 2003 have been, in full compliance with each Legal Requirement that is applicable to such Entity or any of their respective properties, assets, operations or businesses, and no event has occurred, and no condition or circumstance exists, that might (with or without notice or lapse of time) constitute, or result directly or indirectly in, a default under, a breach or violation of, or a failure to comply with, any such Legal Requirement. Neither Company nor any Company Subsidiary has received any notice from any third party regarding any actual, alleged or potential violation of any Legal Requirement.

(b) To the Knowledge of Company, no Governmental Authority has proposed or is considering any Legal Requirement that may affect Company, any Company Subsidiary, their respective properties, assets, operations or businesses, or Company's rights thereto, except to the extent that any such Legal Requirement, if adopted or otherwise put into effect, individually or in the aggregate, will not be material to Company.

#### 3.18 Governmental Approvals.

(a) Company and the Company Subsidiaries have all Governmental Approvals that are necessary or appropriate in connection with the ownership and use of their respective properties or assets or their operation of their respective businesses. Company and the Company Subsidiaries have made all filings with, and given all notifications to, all Government Authorities as required by all applicable Legal Requirements. <u>Schedule 3.18(a)</u> contains an accurate, correct and complete list and summary description of all Governmental Approvals, filings or notifications that are material to the Company, the Company Subsidiaries and the Business. Each such Governmental Approval, filing and notification is valid and in full force and effect, and there is not pending or threatened any Proceeding which could result in the suspension, termination, revocation, cancellation, limitation or impairment of any such Governmental Approval, filing or notification. No violations have been recorded in respect of any Governmental Approvals, and Company knows of no meritorious basis therefor. No fines or penalties are due and payable in respect of any Governmental Approval or any violation thereof.

32

(b) Company has delivered to Purchasers accurate and complete copies of all of the Governmental Approvals, filings and notifications identified in <u>Schedule 3.18(a)</u>, including all renewals thereof and all amendments thereto.

#### 3.19 Proceedings and Orders.

(a) There is no Proceeding pending or threatened against or affecting Company or any Company Subsidiary, any of their respective properties, assets, operations or businesses, or their rights relating thereto. To the Knowledge of Company, no event has occurred, and no condition or circumstance exists, that might directly or indirectly give rise to or serve as a basis for the commencement of any such Proceeding. Company has delivered to Purchasers true, accurate and complete copies of all pleadings, correspondence and other documents relating to any such Proceeding. No insurance company has asserted in writing that any such Proceeding is not covered by the applicable policy related thereto.

(b) Neither Company, the Company Subsidiaries, their respective officers, directors, agents or employees, nor any of their respective properties, assets, operations or businesses, nor any of their respective rights relating to any of the foregoing, is subject to any Order or any proposed Order.

#### **3.20** Environmental Matters.

(iii)

(a) As used in this Agreement, the following terms shall have the meanings indicated below:

(i) *"Environmental and Safety Law"* shall mean any Legal Requirement that pertains to (i) the condition or protection of air, groundwater, surface water, drinking water, land or soil, surface or subsurface strata or medium, natural resources or other environmental media, (ii) the protection of human public or occupational health and safety, (iii) the generation, treatment, manufacturing, use, storage, handling, recycling, presence, Release, disposal, transportation or shipment of any Hazardous Materials, or (iv) any other Legal Requirement that relates to pollution or the protection of the environment.

(ii) "Hazardous Materials" shall mean any material, substance, waste, compound, pollutant or contaminant listed, defined, designated or classified as hazardous, toxic, flammable, explosive, reactive, corrosive, infectious, carcinogenic, mutagenic or radioactive or as hazardous waste, special waste, a pollutant, a deleterious substance or a source of contamination or pollution or otherwise regulated by any Governmental Authority or under any Environmental and Safety Law, including petroleum or petroleum products (including crude oil) and any derivative or by-products thereof, natural gas, synthetic gas and any mixtures thereof, or any substance that is or contains mold, radon, friable asbestos, polychlorinated biphenyls, urea formaldehyde or lead, in each case, whether or not defined as hazardous under any Environmental and Safety Law.

33

the past.

"Property" shall mean all real property leased or owned by Company or any Company Subsidiary either currently or in

(iv) "*Release*" shall mean any releasing, spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, migrating, dumping, or disposing of Hazardous Materials (including the abandonment or discarding of barrels, containers or other closed receptacles containing Hazardous Materials) into the environment, whether intentional or unintentional, negligent or non-negligent, sudden or non-sudden, accidental or non-accidental.

(v) "Facilities" shall mean all buildings and Improvements on the Property.

(b) The Property and the operations of Company and the Company Subsidiaries have at all times since January 1, 2003 been and are in full compliance with all Environmental and Safety Laws.

(c) Company and each Company Subsidiary, as applicable, has obtained and owns or possesses all Governmental Approvals that are required under Environmental and Safety Laws to conduct the Business as it is presently conducted (collectively, the "*Environmental Permits*"), and Company or the applicable Company Subsidiary is in compliance in all material respects with all such Environmental Permits, and all such Environmental Permits are in full force and effect, without conditions or restrictions.

(d) Since January 1, 2003, there has been no Release at the Property or any Facility. No Hazardous Materials are present in, on, or under, or migrating to, or from, the Property.

(e) There are no Proceedings pending or threatened against Company, any Company Subsidiary, any Seller or any of their respective Affiliates relating to a violation of Environmental and Safety Law, noncompliance with Environmental Permits, or alleging liability for the presence of Hazardous Materials. Neither Company nor any Company Subsidiary has received any notice from any Person or Governmental Authority regarding or alleging, and no condition or circumstance currently exists that is reasonably likely to result in (with or without notice or lapse of time or both), a violation or failure to comply with any Environmental and Safety Law or Environmental Permit.

(f) None of the following exists at the Property or any of the Facilities: any asbestos-containing material in any form which is friable; PCBs; active or out-of-service or underground storage tanks or sites from which such storage tanks have been removed; or landfills, solid waste management units, surface impoundments, waste piles or land disposal areas of any kind.

(g) Company has made available to Purchasers complete and correct copies of all studies, reports, surveys, assessments, audits, correspondence, investigations, analysis, tests, and other documents (whether in hard copy or electronic form) in Company's, any Company Subsidiary's or any of their respective Affiliates' possession regarding the presence or alleged presence of Hazardous Materials at, on, or affecting the Property, or regarding the Seller's compliance with any Environmental and Safety Laws.

34

#### 3.21 Taxes.

(a) Each of Company and the Company Subsidiaries has correctly computed all Taxes owing by it, prepared and duly and timely filed all federal, state, provincial, municipal, local and foreign returns, estimates, information statements, elections, designations, reports and any other filings related to Tax Returns required to be filed by it, has timely paid all Taxes which are or may become due and payable on or prior to the date hereof and has made adequate provision for Taxes in the Financial Statements. Each of Company and the Company Subsidiaries has made adequate and timely installments of Taxes required to be made.

(b) Each of Company and the Company Subsidiaries has withheld from each payment made to any of its past and present shareholders, directors, officers, employees and agents the amount of all Taxes and other deductions required to be withheld including, without limitation, withholding of Taxes pursuant to Sections 1441 and 1442 of the *Internal Revenue Code* and has paid such amounts when due, in the form required under appropriate Legal Requirement, or made adequate provision for the payment of such amounts to the proper receiving authorities. The amount of Tax withheld but not remitted by Company and/or the Company Subsidiaries will be retained in the appropriate accounts and will be remitted by Company and/or the Company Subsidiaries to the appropriate authorities when due.

(c) Each of Company and the Company Subsidiaries has collected from each receipt from any of the past and present customers (or other Person paying amounts to Company and/or the Company Subsidiaries) or has self-assessed the amount of all Taxes (including goods and services tax and provincial and/or harmonized sales taxes) required to be collected or self-assessed and has paid and remitted such Taxes when due, in the form required under appropriate Legal Requirement, or made adequate provision for the payment of such amounts to the proper receiving authorities. The amount of Tax collected but not remitted by Company and/or the Company Subsidiaries will be retained in the appropriate accounts and remitted by it to the appropriate authorities when due.

(d) Each of Company and the Company Subsidiaries is not and, to the Company's Knowledge, will not be subject to any assessments, reassessments, levies, penalties or interest with respect to Taxes which will result in any liability on its part in respect of any period ending on or prior to the Closing Date.

(e) Each of Company and the Company Subsidiaries has not been and is not currently required to file any returns, reports, elections, designations or other filings with any taxation authority located in any jurisdiction outside British Columbia and Canada.

(f) Each of Company and the Company Subsidiaries have made available to Purchasers or its accountants, copies of all foreign, federal, state, provincial, municipal and local income, added value, GST, HST and PST Tax Returns for each of Company and the Company Subsidiaries filed for all periods beginning with the year ending April 30, 2005.

(g) Company is a registrant within the meaning of Part IX of the *Excise Tax Act* (Canada) and its registration number is 10247 4194 RT0001.

35

(h) The transfer by Sellers to Finco of Holdco preferred shares (the "*Holdco Shares*"), in exchange for Finco preferred shares, will occur at fair market value and no election under Section 85 of the *Income Tax Act* (Canada) will be made with respect to such transfer.

(i) The exchange of the Holdco Shares for the Finco Note will not result in any Tax under Part IV of the *Income Tax Act* (Canada) or any other Taxes to Finco.

**3.22 Insurance**. <u>Schedule 3.22</u> sets forth an accurate and complete list of all insurance policies, self-insurance arrangements and fidelity bonds, currently in effect, that insure Company, any Company Subsidiary, the Business and/or the Purchased Assets (collectively, the "*Insurance Policies*"). Company has delivered to Purchasers true, correct and complete copies of all Insurance Policies. Each Insurance Policy is valid, binding, and in full force and effect. Neither Company nor any Company Subsidiary is in breach of any Insurance Policy, and no event has occurred which, with notice or the lapse of time, would constitute such a breach, or permit termination, modification, or acceleration, of any Insurance Policy. Since January 1, 2003, Company has not received any notice of cancellation or non-renewal of any Insurance Policy. <u>Schedule 3.22</u> sets forth an accurate and complete list of all claims filed by Company against its Insurance Policies in the last thirty-six (36) months. There is no claim under any Insurance Policy that has been improperly filed or as to which any insurer has questioned, disputed or denied liability. Company has not received any notice of, nor to the Knowledge of Company, do any facts or circumstances exist that might result in, a material increase in the premium for any Insurance Policy.

#### 3.23 Real Property.

(a) <u>Owned Real Property</u>. Neither Company nor any Company Subsidiary owns any real property.

(b) Leased Real Property. Schedule 3.23(b) sets forth an accurate and complete list of all permits, licenses and leasehold or subleasehold estates and other rights to use or occupy any land, buildings or structures, improvements, fixtures, or other interest in real property held by Company or any Company Subsidiary which involve, individually, consideration in excess of \$100,000 annually (the "Leased Real Property") (including the street address, complete legal description and parcel identification number of each Leased Real Property and the name of the lessor) and a list of all Real Property Leases and other Company Contracts affecting each Leased Real Property. Company has been in lawful possession of the premises covered by each Real Property Lease since the commencement of the original term of such Real Property Lease and the Real Property Leases constitute all leases and other rights in existence under which any Person has any right to use, possess or occupy any portion of the Leased Real Property. Company has delivered to Purchasers accurate and complete copies of each Real Property Lease and each Company Contract affecting the Leased Real Property. All Real Property Leases are legally valid and binding and in full force and effect in accordance with their respective terms and there exists no default thereunder, no notice of default has been received by Company or occurrence or condition exists which could result in a default thereunder or termination thereof.

36

(c) All buildings, structures, fixtures, building systems and equipment, and all components thereof, including the roof, foundation, load-bearing walls and other structural elements thereof, heating, ventilation, air conditioning, mechanical, electrical, plumbing and other building systems, environmental control, remediation and abatement systems, sewer, storm and waste water systems, irrigation and other water distribution systems, parking facilities, fire protection, security and surveillance systems, and telecommunications, computer, wiring and cable installations, included in the Leased Real Property (the "*Improvements*") are, to the Company's Knowledge, in good condition and repair and sufficient for the operation of the Business. To the Company's Knowledge there are no structural deficiencies or latent defects affecting any of the Improvements and there are no facts or conditions affecting any of the Improvements that would, individually or in the aggregate, interfere in any respect with the use or occupancy of the Improvements or any portion thereof in the operation of the Business as currently conducted thereon.

(d) Neither Company nor any tenant under any of the Real Property Leases has (i) made any material alterations, additions or improvement to the Leased Real Property that are required to be removed (or of which any landlord or sublandlord could require removal) at termination of the applicable lease term; or (ii) entered into any written sublease, license, option, right, concession or other agreement or arrangement granting to any Person the right to use or occupy such Leased Real Property or any portion thereof or interest therein.

(e) None of Company nor any Company Subsidiary has received written notice from any Governmental Authority or other Person that the use and occupancy of any of the Leased Real Property, as currently used and occupied, and the conduct of the Business thereon, as currently

conducted, violates in any material respect any deed restrictions or applicable Legal Requirements, including building codes, zoning, subdivision or other land use or similar laws. None of Company nor any Company Subsidiary has received written notice of any zoning or other land-use regulation proceeding, or any change in any applicable Legal Requirements, which to Company's knowledge, could affect the use and operation of the Leased Real Property as currently used and operated in any material respect.

(f) To the Knowledge of Company, there are no expropriation, eminent domain, condemnation or similar proceedings pending or threatened affecting any portion of the Leased Real Property.

(g) Company and the Company Subsidiaries have all material licenses, certificates of occupancy, permits, authorizations and approvals required to operate the Business and utilize the Leased Real Property to the extent required in the conduct of the Business as currently conducted on such Leased Real Property. Company and the Company Subsidiaries have all material easements and utility rights required by Legal Requirements or by the normal use and operation of the Leased Real Property to the extent required in the conduct of the Business as currently conducted on such Leased Real Property.

3.24 Title; Sufficiency; Condition of Assets. Company has good and valid title to all of the Purchased Assets free and clear of any Encumbrances (other than Permitted Encumbrances). Company has full right and power to sell, convey, assign, transfer and deliver to Purchasers good and valid title to all of the Purchased Assets, free and clear of any and all

37

Encumbrances (other than Permitted Encumbrances). The Purchased Assets are not subject to any preemptive right, right of first refusal or other right or restriction. The sale, transfer and assignment of the Purchased Assets as contemplated by this Agreement will give Purchasers possession of, and the right to use, all the assets used by Company and the Company Subsidiaries to conduct the Business. Upon Closing, Purchasers will be entitled to the continued possession and use of all Purchased Assets. The Purchased Assets: (i) are in good operating condition and repair, ordinary wear and tear excepted; (ii) are suitable and adequate for continued use in the ordinary course of business; and (iii) conform to all Legal Requirements.

#### 3.25 Customers and Suppliers.

(a) <u>Customers</u>. <u>Schedule 3.25(a)</u> sets forth an accurate, correct and complete list of the 20 largest customers, representatives, dealers or distributors (whether pursuant to a commission, royalty or other arrangement) of the Business, determined on the basis of sales revenues, for each of the fiscal years ended April 30, 2010 and April 30, 2009.

(b) <u>Suppliers</u>. <u>Schedule 3.25(b)</u> sets forth an accurate, correct and complete list of:

(i) the 10 largest suppliers of the Business, determined on the basis of costs of items purchased for each of the fiscal years ended April 30, 2010 and April 30, 2009; and

(ii) all sole source suppliers of the Business.

(c) Neither Company nor any Company Subsidiary has entered into any Contract under which Company or any Company Subsidiary or any of their respective Affiliates is restricted from conducting business with any class of customers, in any geographic area, during any period of time or in any segment of the market. There is no purchase commitment which provides that any supplier will be the exclusive supplier of or distributor for Company or any Company Subsidiary. There is no purchase commitment requiring Company or any Company Subsidiary to purchase the entire output of a supplier.

(d) None of Company or any Company Subsidiary has received any notice or other communication, has received any other information indicating, and otherwise has no Knowledge, that any current customer or supplier required to be listed on <u>Schedule 3.25(a)</u> or <u>Schedule 3.25(b)</u> may cease dealing with Company or any Company Subsidiary, may otherwise materially reduce the volume of business transacted by such Person with Company or any Company Subsidiary has any reason to believe that any such Person will cease to do business with Purchasers after, or as a result of, consummation of the Transaction, or that such Person is threatened with bankruptcy or insolvency. Company has no Knowledge of any fact, condition or event which may, by itself or in the aggregate, adversely affect its relationship with any such Person.

**3.26** Transactions with Affiliates. No Affiliate of Company (a) owns, directly or indirectly, any debt, equity or other interest in any Entity with which Company, any Company Subsidiary or any Seller is affiliated, has a business relationship or competes other than Affiliates

38

that own less than one percent (1%) of the issued and outstanding capital stock of a publicly-traded competitor of Company; (b) is indebted to Company, any Company Subsidiary or any Seller, nor is Company or any Company Subsidiary indebted (or committed to make loans or extend or guarantee credit) to any Affiliate of Company other than with respect to any of Company's obligations to pay accrued salaries, reimbursable expenses or other standard employee benefits; (c) has any direct or indirect interest in any asset, property or other right used in the conduct of or otherwise related to the Business; (d) has any claim or right against Company, any Company Subsidiary or any Seller, and no event has occurred, and no condition or circumstance exists, that might (with or without notice or lapse of time) directly or indirectly give rise to or serve as a basis for any claim or right in favor of any Affiliate against Company, any Company Subsidiary or any Seller; (e) is a party to any Company Contract or has had any direct or indirect interest in, any Company Contract, transaction or business dealing of any nature involving Company, any Company Subsidiary or any Seller; or (f) received from or furnished to Company or any Company Subsidiary any goods or services (with or without consideration).

**3.27 Brokers**. Neither Company nor any Company Subsidiary has retained any broker or finder or incurred any liability or obligation for any brokerage fees, commissions or finders fees with respect to this Agreement or the Transaction.

3.28 Product Return Policies; Warranties. <u>Schedule 3.28</u> sets forth the refund or return policies ("*Return Policies*") of, and all Warranties given or made by Company, any Company Subsidiary or Sellers. None of Company, any Company Subsidiary or Sellers has extended or granted any refund

rights or given or made any Warranties with respect to any Company Products, other than as set forth in <u>Schedule 3.28</u>. Except as set forth on <u>Schedule 3.28</u>, none of the Company Contracts contains any implied or explicit Warranty or support obligations with respect to Company Products, including support obligations that may arise on or after the Closing Date with respect to Company Products that have been delivered to customers prior to the Closing Date. All of the Company Products are in conformity with all applicable contractual commitments and applicable Legal Requirements and all express and implied warranties, if any. Company has accrued a reserve for Warranties as set forth in the Financial Statements and such reserve is sufficient to cover all Warranties of Company and the Company Subsidiaries outstanding as of the Closing Date. No Company Product has been subject to any product recall or epidemic failure or other material failure in design or quality and, to the Knowledge of Company, there is no basis for any of the foregoing. Neither Company nor any Company Subsidiary has been notified of any claims for (and Company does not have any Knowledge of any threatened claims for) any extraordinary product returns, warranty obligations or product services relating to any of the Company Products. There have been no product recalls, withdrawals or seizures with respect to any of the Company Products.

3.29 **Product Liabilities**. Since January 1, 2003, neither Company nor any Company Subsidiary has had, nor does Company or any Company Subsidiary have, any liability (and, to Company's Knowledge, there is no reasonable basis for any present or future action, suit, proceeding, hearing, investigation, charge, complaint, claim or demand against it giving rise to any Liability) arising out of any injury to individuals or property as a result of the ownership, possession or use of any of the Company Products.

39

**3.30 OFAC**. Neither Company nor any Company Subsidiary (a) is engaged in any transaction with any Person that is subject to any sanctions administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury, or (b) is lending, contributing or otherwise making available any resources (including financial resources) to any Person that is subject to any sanctions administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury, or (b) is lending, contributing or otherwise making available any resources (including financial resources) to any Person that is subject to any sanctions administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury.

**3.31 Cultural Business**. Company does not provide any of the services, or engage in any of the activities of a "cultural business" within the meaning of the *Investment Canada Act* (Canada).

3.32 Full Disclosure. No statement (including the representations, warranties and covenants) by Company, any Company Subsidiary, Holdco or Sellers herein or in any exhibit or schedule hereto, including the Company Disclosure Schedule, or in any Transaction Agreement, certificate, written statement or document furnished by Company, any Company Subsidiary, Holdco or Sellers pursuant to this Agreement or in connection with the Transaction contains or will contain at the Closing any untrue statement of a material fact, or omits or will omit at the Closing to state any material fact necessary in order to make the statements contained herein or therein, in the light of the circumstances under which made, not misleading.

#### ARTICLE IV

#### REPRESENTATIONS AND WARRANTIES OF SELLERS AND HOLDCO

Sellers and Holdco jointly and severally represent and warrant to Purchasers that the statements contained in this <u>Article IV</u> are correct and complete as of the date of this Agreement and will be correct and complete as of the Closing Date (as though made then and as though the Closing Date were substituted for the date of this Agreement throughout this <u>Article IV</u> except to the extent such representations and warranties expressly speak as of an earlier date).

4.1 Authority; Validity of Contemplated Transactions. Sellers and Holdco have all requisite corporate power and authority to execute and deliver this Agreement and all other Transaction Agreements to which they are a party and to carry out the provisions of this Agreement and the other Transaction Agreements. The execution, delivery and performance by Sellers and Holdco of this Agreement and the other Transaction Agreements have been approved by all requisite action on the part of Sellers and Holdco. This Agreement has been duly and validly executed and delivered by Sellers and Holdco. Each of this Agreement and the other Transaction Agreements to which any of Sellers and Holdco are a party constitutes, or upon execution and delivery, will constitute, the legal, valid and binding obligation of Sellers and Holdco, enforceable against Sellers and Holdco in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws and equitable principles related to or limiting creditors' rights generally and by general principles of equity.

**4.2 No Conflicts; Required Consents.** The execution, delivery and performance of this Agreement or any other Transaction Agreement by Sellers and Holdco do not and will not (with or without notice or lapse of time):

(a) conflict with, violate or result in any breach of (i) any of the provisions of any of Sellers' or Holdco's certificate of incorporation, notice of articles, articles or similar organization documents (as applicable), (ii) any of the terms or requirements of any Governmental Approval held by any Seller or Holdco; or (iii) any provision of any Contract to which any Seller or Holdco is a party;

(b) give any Governmental Authority or other Person the right to (i) challenge the Transaction; (ii) exercise any remedy or obtain any relief under any Legal Requirement or any Order to which any Seller or Holdco is subject; (iii) declare a default of, exercise any remedy under, accelerate the performance of, cancel, terminate, modify or receive any payment under any Contract to which any Seller or Holdco is a party; or (iv) revoke, suspend or modify any Governmental Approval;

(c) result in the imposition or creation of any Encumbrance upon or with respect to the Purchased Assets (other than any Encumbrance created in connection with the Year 1 Note and the Future Payment Notes); or

(d) except for applicable requirements, if any, under any Antitrust Law, require any Seller or Holdco to obtain any Consent or make or deliver any filing or notice to a Governmental Authority.

**4.3 Proceedings**. There are no Proceedings pending or threatened against or affecting any Seller or Holdco or any of their respective Affiliates (a) challenging or seeking to restrain, delay or prohibit the Transaction or (b) preventing any Seller or Holdco from performing its obligations under this Agreement or any Transaction Documents to which it is a party.

**4.4 Brokers**. No Seller or Holdco has retained any broker or finder or incurred any Liability or obligation for any brokerage fees, commissions or finders fees with respect to this Agreement or the Transaction.

4.5 Sophistication. Each Seller and Holdco acknowledges that such Seller and Holdco is sophisticated and has such knowledge and experience in financial and business matters that such Seller and Holdco is capable of evaluating the merits and risks of the Transaction. Each Seller and Holdco acknowledges that such Seller and Holdco has had adequate time and opportunity to review this Agreement and the Transaction Agreements to which such Seller and Holdco is a party and all other documents requested by such Seller and Holdco with such Seller's and Holdco's own legal counsel, tax and financial advisor. Each Seller and Holdco is relying solely on such counsel and advisors for legal, tax and investment advice with respect to the Transaction.

**4.6 Non-U.S. Persons**. Each Seller and Holdco (a) is not a "U.S. person" within the meaning of Rule 902 of Regulation S under the Securities Act, as presently in effect, and such Seller is not acquiring the CLNE Shares for the account or benefit of any such U.S. person,

41

(b) agrees to transfer or resell the CLNE Shares only in accordance with the provisions of Regulation S, pursuant to registration under the Securities Act, or pursuant to an available exemption from registration and agrees not to engage in hedging transactions with regard to CLNE Shares unless in compliance with the Securities Act, (c) agrees that any certificates for any CLNE Shares issued to such Seller or Holdco shall contain a legend to the effect that transfer is prohibited except in accordance with the provisions of Regulation S, pursuant to registration under the Securities Act or pursuant to an available exemption from registrations involving CLNE Shares may not be conducted unless in compliance with the Securities Act, and (d) agrees that Purchasers are permitted to refuse to register any transfer of any CLNE Shares issued to such Seller or Holdco not made in accordance with the provisions of Regulation under the Securities Act, or pursuant to an available exemption.

4.7 Conduct Prior to Closing. Each Seller and Holdco agrees that between the date of this Agreement and the Closing, unless Purchasers shall otherwise consent in writing (which Purchasers may refuse to do in their sole and absolute discretion) or except as expressly contemplated by this Agreement, such Seller and Holdco shall not sell, dispose of, mortgage, pledge or otherwise transfer or encumber any interest in any of the Purchased Assets held by such Seller or Holdco.

**4.8 Finco Shares.** Sellers are the record and beneficial owner of all right, title and interest, in and to all of the Finco Shares, free and clear of all Encumbrances. The delivery to Purchasers of the Finco Shares as contemplated in this Agreement will transfer to Purchasers valid title to the Finco Shares, free and clear of any and all Encumbrances. The Finco Shares constitute all of the outstanding capital shares and other equity or ownership interests of Finco.

**4.9 Finco**. Finco was formed solely for the purpose of engaging in the Transaction, has engaged in no other business activities and has conducted its operations only as contemplated hereby.

#### ARTICLE V

#### REPRESENTATIONS AND WARRANTIES REGARDING PURCHASERS

Each Purchaser represents and warrants to Company, Holdco and Sellers that the statements contained in this <u>Article V</u> are correct and complete as of the date of this Agreement and will be correct and complete as of the Closing Date (as though made then and as though the Closing Date were substituted for the date of this Agreement throughout this <u>Article V</u> except to the extent such representations and warranties expressly speak as of an earlier date).

5.1 **Organization and Good Standing**. Such Purchaser is a corporation duly organized, validly existing and in good standing under the laws of its jurisdiction of organization.

**5.2 Authority; Binding Nature of Agreements**. Such Purchaser has all requisite corporate power and authority to execute and deliver this Agreement and all other Transaction Agreements to which it is a party and to carry out the provisions of this Agreement and the other Transaction Agreements. The execution, delivery and performance by such Purchaser of this

42

Agreement and the other Transaction Agreements have been approved by all requisite action on the part of such Purchaser. This Agreement has been duly and validly executed and delivered by such Purchaser. Each of this Agreement and the other Transaction Agreements constitutes, or upon execution and delivery, will constitute, the legal, valid and binding obligation of such Purchaser, enforceable against such Purchaser in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws and equitable principles related to or limiting creditors' rights generally and by general principles of equity.

5.3 **Brokers**. Such Purchaser has not retained any broker or finder or incurred any liability or obligation for any brokerage fees, commissions or finders fees with respect to this Agreement or the Transaction.

**5.4 Proceedings**. Subject to any Proceeding or any notice of a Proceeding referred to in <u>Section 7.1(g)</u>, there are no Proceedings pending or, to the Knowledge of such Purchaser, threatened against or affecting such Purchaser or any of its Affiliates (a) challenging or seeking to restrain, delay or prohibit the Transaction or (b) preventing such Purchaser from performing in all material respects its obligations under this Agreement or any Transaction Documents to which it is a party.

5.5 Status of Canadian AcqCo. Canadian AcqCo is a taxable Canadian corporation.

**5.6** Second Closing Shares. The Second Closing Shares will have legal stated capital and a redemption value equal to the fair market value of the Finco Shares.

#### ARTICLE VI

#### PRE-CLOSING COVENANTS

6.1 Company's Conduct of the Business Prior to Closing. Except as set forth on <u>Schedule 6.1</u>, from and after the date hereof and prior to the Closing Date, except with the prior written consent of Purchasers, Company shall, and shall cause the Company Subsidiaries to:

(a) Conduct the Business in the ordinary course of business (including Patent prosecution);

(b) Pay all of its Liabilities and Taxes when due, subject to good faith disputes over such Liabilities or Taxes;

(c) Maintain insurance coverage in amounts adequate to cover the reasonably anticipated risks of Company; and

(d) Use commercially reasonable efforts to (i) preserve intact all rights of Company to retain its employees and (ii) maintain good relationships with employees, licensors, licensees, suppliers, contractors, distributors, customers, and others having business dealings with Company.

43

6.2 **Restrictions on Company's Conduct of the Business Prior to Closing**. Except as set forth in <u>Schedule 6.2</u>, from and after the date hereof and prior to the Closing Date, except with the prior written consent of Purchasers, Company shall not, and Company shall cause the Company Subsidiaries not to:

(a) Acquire by merging, amalgamating, arranging or consolidating with, or by purchasing any equity securities or assets of, or by any other manner, any business or any Entity;

(b) Sell, transfer, lease, license, assign, option or otherwise encumber any of its assets (including any Company IP Assets), except in the ordinary course of business consistent with past practice;

(c) Take any action not announced prior to the date of this Agreement with respect to the customers, suppliers or distributors of Company or any Company Subsidiary, including providing promotions, coupons, discounts or price increases, other than in the ordinary course of business consistent with past practice;

(d) Enter into any Contracts, except on commercially reasonable terms in the ordinary course of business;

(e) Violate any applicable Legal Requirement;

(f) Change or announce any change to the Company Products;

(g) Violate, terminate or amend any Company Contract or Governmental Approval;

(h) Commence any Proceeding other than for (i) the routine collection of Accounts Receivable and Subsidiary Accounts Receivable or (ii) injunctive relief on the grounds that Company or any Company Subsidiary has suffered immediate and irreparable harm not compensable in money damages;

(i) Issue, sell, contract to issue or sell, pledge, dispose of, grant, encumber, split, reclassify or authorize the issuance, sale, pledge, disposition, grant, split, reclassification or encumbrance of (i) any equity or similar interest in any Company Subsidiary or (ii) any options, warrants, convertible securities or other rights of any kind to acquire any equity or similar interest in any Company Subsidiary;

(j) Purchase, lease, license or otherwise acquire any assets, except for supplies acquired by Company or any Company Subsidiary in the ordinary course of business;

(k) Make any capital expenditure in excess of \$50,000, individually or in the aggregate;

(l) Incur any Indebtedness other than within the terms of existing credit facilities;

44

(m) Provide any credit, loan, advance, guaranty, endorsement, indemnity, warranty or mortgage to any Person, including any of the customers, shareholders, officers, employees or directors of Company, other than those made in the ordinary course of business;

(n) Change its credit practices, accounting methods or practices or standards used to maintain its books, accounts or business records;

(o) Change the terms of its Accounts Payable, Accounts Receivable or Subsidiary Accounts Receivable or take any action directly or indirectly to cause or encourage any acceleration or delay in the payment, collection or generation of its accounts, Accounts Receivable or Subsidiary Accounts Receivable, other than in the ordinary course of business consistent with past practice;

(p) Incur or become subject to any Liability or settle any Liability, contingent or otherwise, except current Liabilities in the ordinary course of business;

(q) Amend its certificate of incorporation, notice of articles, articles or similar charter documents;

(r) Accelerate, amend or change the period of exercisability or vesting of any rights granted under any Company Benefit Plans, except as specifically contemplated in this Agreement;

(s) Hire any new employee other than in the ordinary course of business; terminate any officer or key employee of Company; increase the annual level of compensation of any existing employee; establish or adopt any Employee Benefit Plan; or grant any bonuses, benefits or other forms of direct or indirect compensation to any employee, officer, director or consultant other than in the ordinary course of business and consistent with past practice;

(t) Make any severance payments to any employee, officer or director, except payments made pursuant to written agreements outstanding as of the date of this Agreement and listed in <u>Schedule 3.15(b)</u>;

(u) With respect to any Company Subsidiary, make, amend, or revoke any election relating to Taxes; adopt or change any accounting method relating to Taxes; file any amendment to any Tax Return; enter into any Tax sharing, allocation, indemnity or similar agreement; enter into any closing agreement; settle or compromise any claim or assessment relating to Taxes; consent to any extension or waiver of the limitations period applicable to any Taxes or Tax Return; or

(v) Enter into any Contract or otherwise agree, in writing or otherwise, to take any of the actions described in <u>Section 6.2(a)</u> through (<u>u</u>) above, or any action that would make any of its representations or warranties contained in this Agreement untrue or incorrect in any material respect or prevent it from performing or cause it not to perform its covenants hereunder.

6.3 No Solicitation. Until the earlier of (a) the Closing or (b) the termination of this Agreement pursuant to its terms, Sellers, Holdco and Company shall not, and they shall cause the Company's Representatives, Holdco's Representatives and Sellers' Representatives not to,

45

directly or indirectly, (i) initiate, solicit or encourage any inquiries, or make any statements to third parties which may reasonably be expected to lead to any proposal concerning the sale of Company or any Company Subsidiary (whether by way of merger, amalgamation, arrangement, purchase of capital shares, purchase of assets or otherwise) (a "*Competing Transaction*"); or (ii) hold any discussions or enter into any agreements with, or provide any information or respond to, any third party concerning a proposed Competing Transaction or cooperate in any way with, agree to, assist or participate in, solicit, consider, entertain, facilitate or encourage any effort or attempt by any third party to do or seek any of the foregoing. If at any time prior to the earlier of (x) the Closing or (y) the termination of this Agreement pursuant to its terms, Sellers, Holdco or Company is approached in any manner by a third party concerning a Competing Transaction (a "*Competing Party*"), Sellers shall promptly inform Purchasers regarding such contact and furnish Purchasers with a copy of any inquiry or proposal, or, if not in writing, a description thereof, including the name of such Competing Party, and Sellers shall keep Purchasers informed of the status and details of any future notices, requests, correspondence or communications related thereto.

6.4 Certain Notifications. From the date of this Agreement until the Closing, Sellers, Holdco and Company shall promptly notify Purchasers in writing regarding any:

(a) Action taken by Company or any Company Subsidiary not in the ordinary course of business and any circumstance or event that could reasonably be expected to have a Material Adverse Effect on Company or the Business;

(b) Fact, circumstance, event, or action by Company or any Company Subsidiary (i) which, if known on the date of this Agreement, would have been required to be disclosed in or pursuant to this Agreement; or (ii) the existence, occurrence, or taking of which would result in any of the representations and warranties of Company or Sellers contained in this Agreement or in any Transaction Agreement not being true and correct when made or at Closing;

(c) Breach of any covenant or obligation of Sellers, Holdco or Company hereunder; and

(d) Circumstance or event which will result in, or could reasonably be expected to result in, the failure of Sellers, Holdco or Company to timely satisfy any of the conditions to Closing set forth in <u>Article VII</u>;

provided, that the delivery of any notice pursuant to this <u>Section 6.4</u> shall not limit or otherwise affect any remedies available to the party receiving such notice.

6.5 Access to Information. From the date of this Agreement until the Closing, Sellers and Company shall (i) permit Purchasers and their Representatives to have free and complete access at all reasonable times, and in a manner so as not to interfere with the normal business operations of Company, to all premises, properties, personnel, Persons having business relationships with Company and any Company Subsidiary (including suppliers, licensees, customers and distributors), books, records (including Tax records), Contracts, and documents of or pertaining to Company and the Company Subsidiaries; (ii) furnish Purchasers and/or Purchasers' Representatives with all financial, operating and other data and information related

46

to Company and the Company Subsidiaries (including copies thereof), as Purchasers may reasonably request; and (iii) otherwise cooperate and assist, to the extent reasonably requested by Purchasers, with Purchasers' investigation of Company and the Company Subsidiaries. No information or knowledge obtained in any investigation pursuant to this <u>Section 6.5</u> shall affect or be deemed to modify any representation or warranty contained herein or the conditions to the obligations of the parties to consummate the Transaction.

6.6 Reasonable Commercial Efforts. From the date of this Agreement until the Closing, each of Company, Sellers, Holdco and Purchasers shall use their respective reasonable commercial efforts to cause to be fulfilled and satisfied all of the other party's conditions to Closing set forth in <u>Article VII</u>.

#### 6.7 Consents.

(a) Company shall use its commercially reasonable efforts to obtain all Consents needed to consummate the Transaction or that are listed on <u>Schedule 3.5</u>.

(b) Company and Purchasers shall file all filings required under the Antitrust Laws of any jurisdiction, as soon as practicable following the date hereof, but in any event within ten (10) Business Days following the date hereof (the date on which such filing is made, the "**Original** *Filing Date*"). Purchasers shall pay the applicable filing fee required in connection with any antitrust filing required in connection with the Transaction. Company and Purchasers shall respond promptly to any request for information that may be issued by any Government Antitrust Entity. Subject to the terms and conditions herein, Purchasers and Company shall use commercially reasonable efforts to cause the waiting periods under the Antitrust Laws of any jurisdiction, as applicable, to terminate or expire at the earliest possible date after the Original Filing Date (it being understood that this provision is not intended to require any party to seek early termination of any applicable waiting period). For purposes of this Agreement, "Antitrust Laws" shall mean any federal, state or foreign statutes, rules, regulations, orders or decrees that are designed to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade. Without limiting the generality of Company's undertakings pursuant to this <u>Section 6.7(b)</u>, Company shall, in each case with the consent of Purchasers:

(i) use its commercially reasonable efforts to prevent the entry in a Proceeding brought under any Antitrust Law by a Governmental Authority with jurisdiction over the enforcement of any applicable Antitrust Laws ("*Government Antitrust Entity*") or any other party of any permanent or preliminary injunction or other order that would make consummation of the Transaction in accordance with the terms of this Agreement unlawful or that would prevent or delay such consummation; <u>provided</u> that, Purchasers and their counsel shall be responsible for all discussions with any Government Antitrust Entity (after consultation with Company and its counsel) to the maximum extent permitted by Legal Requirement and except as required by any Government Antitrust Entity; and

(ii) take promptly, in the event that such an injunction or order has been issued in such a Proceeding, any and all steps, including the appeal thereof or the posting of a bond, necessary to vacate, modify or suspend such injunction or order so as to permit such consummation on a schedule as close as possible to that contemplated by this Agreement.

47

(c) Subject to applicable Legal Requirements and subject to all applicable privileges, including the attorney-client privilege, Purchasers and Company will consult and cooperate with one another in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any party hereto relating to proceedings under any Antitrust Law, and shall promptly inform the other of any oral communication with, and provide copies of written communications with, any Governmental Antitrust Entity regarding any such filings or this transaction.

(d) Notwithstanding anything to the contrary in this <u>Section 6.7</u> or elsewhere in this Agreement, Purchasers shall not be required to (i) obtain any Consent, (ii) accept material concessions in connection with obtainment of any Consent, (iii) incur any material Liability or obligation of any kind or (iv) violate any applicable Legal Requirement.

#### 6.8 Confidentiality.

(a) Each party hereto agrees that, for a period commencing on the Closing Date and ending three (3) years following the date that Bradley N. Miller (a Seller under this Agreement) ceases to be an employee of Purchasers or an Affiliate of Purchasers, it will not, and will use reasonable efforts to ensure that its Affiliates will not, through any action or inaction, use (except as contemplated by this Agreement), or disclose to any other Person, any Confidential Information relating to the other party (it being understood that following Closing this Section 6.8 shall apply to the use or disclosure of Company's Confidential Information by Sellers and Holdco and after Closing shall not apply with respect to the use or disclosure of the Confidential Information of Company by Purchasers); provided, however, that the foregoing prohibitions shall not apply to (i) disclosures that are required by any Legal Requirement (including any rule or regulation of the SEC or of a stock exchange which may require such disclosure) or by a Governmental Authority; (ii) information that is ascertainable or obtained from public or published information or is otherwise publicly known through no wrongful act of the using or disclosing party; (iii) information received from a Person not known after reasonable inquiry to the using or disclosing party to be under an obligation to keep such information that was rightfully known by the disclosing party before receipt from the other party; and (vi) information disclosed to or filed with any Person for the purpose of obtaining consents to, or the financing of, the transactions contemplated by this Agreement. Notwithstanding anything herein to the contrary, each party to this Agreement may (without prior notification to, or approval or consent by, any other party) disclose to taxing authorities and/or to such party's representatives (including outside counsel and advisors) any confidential or non-public information that is required to be disclosed in connection with such party's tax fillings,

(b) Notwithstanding Subsection (a) above, in the event a party is required to disclose Confidential Information of another party (in such event, such party is a "*Nondisclosing Party*," and the party required to disclose is the "*Disclosing Party*") pursuant to any Legal Requirement, and would otherwise be prohibited from doing so under this <u>Section 6.8</u>, the Disclosing Party shall: (i) promptly notify the Nondisclosing Party of the existence, terms and circumstances surrounding such requirement; (ii) consult with the Nondisclosing Party on the advisability of taking legally available steps to resist or narrow such request; and (iii) if

disclosure of such Confidential Information is required, furnish only that portion of the Confidential Information which the Disclosing Party is legally compelled to disclose and advise the Nondisclosing Party reasonably in advance of such disclosure so that the Nondisclosing Party may seek an appropriate protective order or other reliable assurance that confidential treatment will be accorded such Confidential Information. The Disclosing Party shall not oppose actions by the Nondisclosing Party to obtain an appropriate protective order or other reliable assurance that confidential treatment will be accorded such Confidential Information. 6.9 Indebtedness. At Closing, Company shall cause there to be no Encumbrances on any of the Purchased Assets, other than Permitted Encumbrances.

**6.10 Public Announcements**. No party to this Agreement shall make, or cause or permit to be made, any press release or public announcement, or otherwise communicate with any news media, in respect of this Agreement or the Transaction without the prior written consent of the other parties, and the parties shall cooperate as to the timing and contents of any such press release or public announcement; provided, however, that Purchasers and any Affiliate of Purchasers may, without the prior consent of, but with written notice to, the other parties, make such press release, public disclosure, securities law filings, or public announcement as may be required in Purchasers' opinion under applicable Legal Requirements, including stock exchange rules and regulations (including the filing of Forms 8-K announcing the Transaction and the filing of this Agreement with the SEC as may be required under applicable Legal Requirements).

6.11 **Books and Records**. At the Closing, Sellers and Company shall deliver or cause to be delivered to Purchasers all original documents, books and records relating to the Purchased Assets and Company Subsidiaries that will not automatically transfer to Purchasers (other than any books and records that are Excluded Assets).

**6.12 Cooperation**. After the Closing, upon the request of Purchasers, Sellers, Holdco and Company shall (i) execute and deliver any and all further materials, documents and instruments of conveyance, transfer or assignment as may reasonably be requested by Purchasers to effect, record or verify the transfer to, and vesting in Purchasers, of Sellers', Holdco's and Company's right, title and interest in and to the Purchased Assets, free and clear of all Encumbrances (other than Permitted Encumbrances), in accordance with the terms of this Agreement; and (ii) cooperate with Purchasers, at Purchasers' expense, to enforce the terms of any Company Contracts, including terms relating to confidentiality and Intellectual Property Rights, and to contest or defend against any Proceeding relating to the Transaction or to the operation of Company's Business before the Closing Date.

#### 6.13 Non-Competition and Non-Solicitation Agreements.

#### (a) <u>First Closing</u>

(i) Effective as of the First Closing, Company agrees that, for a period commencing on the Closing Date and ending three (3) years following the date that Bradley N. Miller (a Seller under this Agreement) ceases to be an employee of Purchasers or an Affiliate of Purchasers (the "*Noncompetition Period*"), Company will not (and will cause all of its Affiliates

49

not to), directly or indirectly, anywhere Company and the Company Subsidiaries are conducting their business as of the Closing Date: (i) engage in any Prohibited Business (as defined below); or (ii) own or purchase any interest in, manage, operate, or finance any Person that engages in any Prohibited Business. "*Prohibited Business*" means the business of Company and the Company Subsidiaries as currently conducted or as proposed to be conducted.

(ii) <u>Non-Solicitation of Employees</u>. During the Noncompetition Period, Company shall not, directly or indirectly, whether through a third Person or Affiliate or otherwise, recruit, solicit, induce, invite or otherwise retain or encourage any Person who is employed by or has a contractual or consulting relationship with Purchasers or the Business to accept employment with or enter into a consulting or other business relationship with any Person other than Purchasers or their respective Affiliates, or terminate any such relationship with Purchasers or their respective affiliates.

(iii) <u>Non-Solicitation of Customers and Suppliers</u>. During the Noncompetition Period, Company shall not, for its own account or for the account of any other Person, call upon any Person that has purchased any Company Products or that has acted (or may act in the future) as a supplier or vendor of Company or Purchasers with respect to the Company Products for the purpose of soliciting or selling products or services in competition with the Business. During the Noncompetition Period, Company shall not induce any current, former or potential customer, supplier or vendor of the Business, Purchasers or any of their respective Affiliates to cease doing business in whole or in part with, or otherwise interfere with the business of, the Business, Purchasers or any of their respective Affiliates.

(iv) The covenants contained in Subsections (i)-(iii) shall be construed as a series of separate covenants, one for each county, city, state, province and country throughout the world. Except for geographic coverage, each such separate covenant shall be deemed identical in terms to the covenant contained in Subsections (i)-(iii). If, in any Proceeding, a court refuses to enforce any of such separate covenants (or any part thereof), then such unenforceable covenant (or such part) shall be eliminated from this Agreement to the extent necessary to permit the remaining separate covenants (or portions thereof) to be enforced. In the event that the provisions of this <u>Section 6.13(a)</u> are deemed to exceed the time, geographic or other scope limitations permitted by applicable Legal Requirement, then such provisions shall be reformed to the maximum time, geographic or scope limitations, as the case may be, permitted by applicable Legal Requirements.

(v) Company acknowledges that (1) the goodwill associated with the Business and customer relationships prior to the Closing are an integral component of the value of the Purchased Assets to Purchasers and are reflected in the payments to Company from Purchasers contemplated by this Agreement and (2) Company's agreements as set forth herein are necessary to preserve the value of the Purchased Assets and the Business, including its goodwill and customer relationships, for Purchasers following Closing. Company also acknowledges that the limitations of time, geographic scope and scope of activity agreed to in this Agreement are reasonable.

#### (b) <u>Third Closing</u>

(i) Effective as of the Third Closing, Holdco and each Seller agrees that, during the Noncompetition Period, Holdco and each Seller will not (and will cause all of their respective Affiliates not to), directly or indirectly, anywhere Company and the Company Subsidiaries are conducting their business as of the Closing Date: (i) engage in any Prohibited Business; or (ii) own or purchase any interest in, manage, operate, or finance any Person that engages in any Prohibited Business.

(ii) <u>Non-Solicitation of Employees</u>. During the Noncompetition Period, Holdco and each Seller shall not, directly or indirectly, whether through a third Person or Affiliate or otherwise, recruit, solicit, induce, invite or otherwise retain or encourage any Person who is employed by or has a contractual or consulting relationship with Purchasers or the Business to accept employment with or enter into a consulting or other business relationship with any Person other than Purchasers or their respective Affiliates, or terminate any such relationship with Purchasers or their respective affiliates.

(iii) <u>Non-Solicitation of Customers and Suppliers</u>. During the Noncompetition Period Holdco and each Seller shall not, for his own account or for the account of any other Person, call upon any Person that has purchased any Company Products or that has acted (or may act in the future) as a supplier or vendor of Company or Purchasers with respect to the Company Products for the purpose of soliciting or selling products or services in competition with the Business. During the Noncompetition Period, Holdco and each Seller shall not induce any current, former or potential customer, supplier or vendor of the Business, Purchasers or any of their respective Affiliates to cease doing business in whole or in part with, or otherwise interfere with the business of, the Business, Purchasers or any of their respective Affiliates.

(iv) The covenants contained in Subsections (i)-(iii) shall be construed as a series of separate covenants, one for each county, city, state, province and country throughout the world. Except for geographic coverage, each such separate covenant shall be deemed identical in terms to the covenant contained in Subsections (i)-(iii). If, in any Proceeding, a court refuses to enforce any of such separate covenants (or any part thereof), then such unenforceable covenant (or such part) shall be eliminated from this Agreement to the extent necessary to permit the remaining separate covenants (or portions thereof) to be enforced. In the event that the provisions of this <u>Section 6.13(b)</u> are deemed to exceed the time, geographic or other scope limitations permitted by applicable Legal Requirement, then such provisions shall be reformed to the maximum time, geographic or scope limitations, as the case may be, permitted by applicable Legal Requirements.

(v) Holdco and Sellers acknowledge that (1) the goodwill associated with the Business and customer relationships prior to Closing are an integral component of the value of the Purchased Assets to Purchasers and are reflected in the payments to Holdco and Sellers from Purchasers contemplated by this Agreement and (2) Holdco's and Sellers' agreements as set forth herein are necessary to preserve the value of the Purchased Assets and the Business, including its goodwill and customer relationships, for Purchasers following Closing. Holdco and Sellers also acknowledge that the limitations of time, geographic scope and scope of activity agreed to in this Agreement are reasonable.

51

#### 6.14 Certain Employee Matters.

(a) <u>Schedule 6.14</u> contains a list of key employees of Company to whom Purchasers may (in their sole discretion) make an offer of continued employment, including in some or all cases by offering to enter into an employment agreement with such employees, such offers to be contingent upon Closing ("*Key Employees*"). Sellers, Holdco and Company shall, prior to Closing, cooperate with Purchasers' reasonable requests for information and cooperation pertaining to employees, employee benefit matters and the employment by Company of the Key Employees and shall not attempt, or in any way discourage the Key Employees from accepting any offer of continued employment made to them by the Purchasers.

(b) Subject to the Closing and the terms of this Agreement, Purchasers agree to offer employment effective as of the Closing Date on terms substantially similar in the aggregate to those existing as of the Closing Date to all of the individuals who are employees of Company as at the Closing Date (the "*Designated Employees*").

(c) In this Agreement, "*Transferred Employees*" means those Designated Employees and Key Employees who have accepted the Purchasers' offer of employment made pursuant to this <u>Section 6.14</u> and "*Employee Start Date*" means the Closing Date or such later date on which a Transferred Employee commences active employment with Purchasers.

(d) Purchasers shall not assume any of the Company Benefit Plans or Liability for accrued benefits or any other Liability under or in respect of any of the Company Benefit Plans. The Transferred Employees shall, as of their applicable Employee Start Date, cease to accrue further benefits under the Company Benefit Plans.

(e) Without limiting Company's obligations in respect of Persons employed by Company prior to the Closing Date, Company shall be responsible for: (i) all Liabilities for salary, wages, bonuses, commissions, vacation pay and other compensation relating to employment of all Persons by the Company prior to the Closing Date and all Liabilities under or in respect of the Company Benefit Plans; (ii) all severance payments, damages for wrongful dismissal and all related costs in respect of the termination by Company of the employment of any Designated Employee who does not become a Transferred Employee based upon the existence of any written or verbal contract or arrangement with such Designated Employee (it being agreed that Purchasers will be responsible for all severance payments, damages for wrongful dismissal and all related costs in respect of the termination by Company of the employment of any Designated Employee who does not become a Transferred Employee where it is finally determined by a court of competent jurisdiction that the terms of offer made to such Designated Employee was not substantially similar in the aggregate to those existing as of the Closing Date for such Designated Employee); (iii) all Liabilities for claims for injury, disability, death or workers' compensation arising from or related to employment by Company prior to the Closing Date; and (iv) all employment-related claims, penalties and assessments in respect of employment by Company and arising out of matters which occurred prior to the Closing Date.

#### 6.15 Registration of CLNE Shares.

(a) Purchasers shall use their reasonable best efforts to cause the CLNE Shares to be registered under the Securities Act so as to permit the resale thereof from time to time in accordance with the Sellers' intended method of disposition (which may include, without

limitation, short sales or other hedging transactions) and in connection therewith shall use its reasonable best efforts to prepare and file a registration statement (the "*Registration Statement*") with the SEC with respect to the CLNE Shares prior to the Closing Date, and shall use their reasonable best efforts to cause the Registration Statement to become effective on or prior to the Closing Date; provided, however, that Company, Holdco and Sellers shall provide all such information and materials to Purchasers and take all such action as may be required in order to permit Purchasers to comply with all applicable

requirements of the SEC and to obtain any desired acceleration of the effective date of such Registration Statement. Such provision of information and materials is a condition precedent to the obligations of Purchasers pursuant to this <u>Section 6.15</u>.

(b) Purchasers shall: (i) prepare and file with the SEC the Registration Statement in accordance with <u>Section 6.15(a)</u> with respect to the CLNE Shares; (ii) prior to the Closing Date and, if requested by Sellers in connection with any subsequent resale of CLNE Shares, enter into an agreement (the "Underwriting Agreement") with an underwriter designated by the Sellers in connection with the resale of the CLNE Shares that is customary for underwritten offerings of equity securities in accordance with Sellers' intended method of distribution, in form and substance reasonably satisfactory to such underwriter, which Underwriting Agreement shall include, without limitation, customary provisions relating to the indemnification of, and contribution in connection with the liability of, the underwriter and its and its affiliates (including, without limitation, any such affiliate that is a counterparty to an equity derivative transaction with a Seller relating to the CLNE Shares), the provision of customary opinions, accountants' comfort letters and lawyers' negative assurance letters, and providing such underwriter a reasonable opportunity to conduct a due diligence investigation of the Purchasers; (iii) prepare and file with the SEC such amendments and supplements to the Registration Statement and the prospectus used in connection therewith, and take such other actions, as may be reasonably necessary to keep such Registration Statement continuously effective until the earlier of (A) the one-year anniversary of the Closing Date and (B) the date on which all of the CLNE Shares have been sold by the Sellers; and (iv) comply with the provisions of the Securities Act with respect to the offer and sale or other disposition of all securities proposed to be registered in the Registration Statement.

(c) Notwithstanding any other provision of this <u>Section 6.15</u>, Purchasers shall have the right at any time after the fifteenth (15<sup>th</sup>) NASDAQ Global Market trading day following the date on which the Registration Statement is declared effective to require that Sellers suspend open market offers and sales of CLNE Shares for the period of time specified in this <u>Section 6.15(c)</u> if, in the reasonable, good-faith judgment of Purchasers, there is in existence material undisclosed information or events with respect to Purchasers (the "*Suspension Right*"). In the event Purchasers exercise the Suspension Right, such suspension will continue for the shorter of (i) the period of time reasonably necessary for disclosure to occur at a time that is not materially detrimental to Purchasers or (ii) the period until such time as the information or event is no longer material, each as reasonably determined in good faith by Purchasers; provided that in no event may the aggregate number of days on which offers and sales have been suspended pursuant to this <u>Section 6.15(c)</u> exceed twenty (20) trading days. Purchasers will promptly give Company, Holdco and Sellers notice, in a writing signed by an executive officer of Clean Energy, of any such suspension (the "*Suspension Notice*"). Purchasers agree to notify Company, Holdco and Sellers promptly upon termination of the suspension (the "*Resumption Notice*").

53

(d) Purchasers shall pay all of the registration and filing fees and fees and disbursements of Purchasers' outside counsel and independent registered public accounting firm incurred in connection with any registration of CLNE Shares pursuant to this <u>Section 6.15</u>.

6.16 **Corporate Name**. Company will take all actions necessary to, effective as of the Closing Date, change its corporate name to one dissimilar to "I.M.W. Industries Ltd."

6.17 SEC Filings. Purchasers will use commercially reasonable efforts to file, as soon as reasonable practicable following Purchasers' receipt of the Audited Financial Statements, with the SEC the Audited Financial Statements, the pro forma financial information required by Form 8-K under the Exchange Act in connection with the Transaction (such pro forma financial information to be prepared by Purchasers after their receipt of the Audited Financial Statements) and a description of the business of Company and the Company Subsidiaries.

#### ARTICLE VII

#### CONDITIONS TO CLOSING

7.1 **Conditions to Purchasers' Obligation to Consummate First Closing**. The obligations of Purchasers to consummate the First Closing shall be subject to the satisfaction, on or prior to the First Closing, of each of the following conditions, any of which may be waived by Purchasers in writing:

(a) <u>Representations, Warranties and Covenants</u>. (i) The representations and warranties made by Company and Sellers in <u>Section 3.4</u>, <u>Section 3.17</u>, <u>Section 4.1</u> and <u>Section 4.8</u> in this Agreement shall be true and correct in all respects (disregarding the presence of any materiality or similar qualifiers contained therein) and (ii) all of the other representations and warranties made by Company, Holdco and Sellers in this Agreement and the Transaction Agreements and qualified as to materiality shall be true and correct, and those not so qualified shall be true and correct in all material respects, in each case, on the Closing Date with the same force and effect as if this Agreement had been executed on and as of the Closing Date. Company, Holdco and Sellers shall have duly performed all of the agreements and covenants and satisfied all of the conditions to be performed or complied with by them on or prior to the Closing Date (including agreements of Sellers to cause Company to take or refrain from taking certain actions).

(b) <u>Documents</u>. Company, Holdco and Sellers shall have delivered to Purchasers all of the documents and agreements set forth in <u>Section 2.1</u>.

(c) <u>No Proceedings</u>. Since the date of this Agreement, no Proceeding shall have been commenced or threatened against Purchasers, or against any Representative of Purchasers (a) involving any challenge to, or seeking Damages or other relief in connection with, the Transaction; or (b) that may have the effect of preventing, delaying, making illegal, imposing limitations or conditions on or otherwise interfering with the Transaction.

(d) <u>No Material Adverse Effect</u>. No event that has had, or could reasonably be expected to have, a Material Adverse Effect on Company or the Business shall have occurred since the date hereof.

54

(e) <u>Employment Agreements</u>. The Employment Agreements with Purchasers shall have been duly executed by no less than 80% of the Key Employees and all such Employment Agreements shall continue to be in full force and effect as of the Closing Date and no Employment Agreements shall have been amended.

(f) <u>Audited Financial Statements; SEC Filings</u>.

(i) Sellers and Company shall have delivered to Purchasers the Audited Financial Statements, which shall be in form and substance (1) acceptable to Purchasers and (2) necessary and appropriate (in Purchasers' sole judgment) for Purchasers to satisfy their Form 8-K filing requirements under the Exchange Act (including the requirement to file pro forma financial information in connection with the Transaction (such pro forma financial information to be prepared by Purchasers after their receipt of the Audited Financial Statements).

(ii) The Audited Financial Statements, the pro forma financial information required by Form 8-K under the Exchange Act in connection with the Transaction and a description of the business of Company and the Company Subsidiaries shall have been filed with the SEC.

(g) <u>National Security</u>. The Minister of Industry shall not have sent to Purchasers a notice under subsection 25.2(1) of the *Investment Canada Act* (Canada) and the Governor in Council shall not have made an order under subsection 25.3(1) of the *Investment Canada Act* (Canada) in relation to the Transaction or, if such a notice has been sent or such an order has been made, Purchasers shall have subsequently received (i) a notice under paragraph 25.2(4)(a) of the *Investment Canada Act* (Canada) indicating that no order for the review of the Transaction on grounds of national security will be made, (ii) a notice under paragraph 25.3(6)(b) of the *Investment Canada Act* (Canada) indicating that no further action will be taken in respect of the Transaction or (iii) a copy of an order under paragraph 25.4(1)(b) authorizing the Transaction, provided that such order is on terms and conditions satisfactory to Purchasers in their sole discretion.

(h) <u>Registration Statement</u>. The Registration Statement shall have been declared effective by the SEC and shall be effective on the Closing Date, and no stop order suspending effectiveness shall have been issued; no action, suit, proceeding or investigation by the SEC to suspend the effectiveness of the Registration Statement shall have been initiated and be continuing.

7.2 Conditions to Company's Obligation to Consummate First Closing. The obligation of Company to consummate the First Closing shall be subject to the satisfaction, on or prior to the First Closing, of each of the following conditions, any of which may be waived by Company in writing:

(a) <u>Deliveries</u>. Purchasers shall have delivered to Company, Sellers and Holdco all of the documents and agreements set forth in <u>Section 2.2</u>.

(b) <u>Registration Statement</u>. The Registration Statement shall have been declared effective by the SEC and shall be effective on the Closing Date, and no stop order suspending effectiveness shall have been issued; no action, suit, proceeding or investigation by the SEC to suspend the effectiveness of the Registration Statement shall have been initiated and be continuing.

55

7.3 Conditions to Purchasers' Obligation to Consummate Second Closing. The obligations of Purchasers to consummate the Second Closing shall be subject to the satisfaction, on or prior to the Second Closing, of each of the following conditions, any of which may be waived by Purchasers in writing:

(a) <u>First Closing Completed</u>. The parties shall have consummated the First Closing;

(b) <u>Asset Transfer</u>. Company shall have transferred to Holdco the First Closing Note, the First Closing Shares, the Earnout Shares and the Depreciable Assets Shares; and

Section 2.3.

(c)

Deliveries. Sellers, Company and Holdco shall have delivered to Purchaser all of the documents and agreements set forth in

7.4 **Conditions to Company's, Holdco's and Sellers' Obligations to Consummate Second Closing**. The obligation of Company, Holdco and Sellers to consummate the Second Closing shall be subject to the satisfaction, on or prior to the Second Closing, of each of the following conditions, any of which may be waived by Company, Holdco and Sellers in writing:

- (a) <u>First Closing Completed</u>. The parties shall have consummated the First Closing; and
- (b) <u>Deliveries</u>. Purchasers shall have delivered to Company, Sellers and Holdco all of the documents and agreements set forth in

Section 2.4.

**7.5 Conditions to Purchasers' Obligation to Consummate Third Closing**. The obligations of Purchasers to consummate the Third Closing shall be subject to the satisfaction, on or prior to the Third Closing, of each of the following conditions, any of which may be waived by Purchasers in writing:

- (a) <u>Closings Completed</u>. The parties shall have consummated the First Closing and the Second Closing;
- (b) <u>Asset Transfer</u>. Sellers shall have transferred to Holdco the Second Closing Shares; and
- (c) <u>Deliveries</u>. Sellers, Company and Holdco shall have delivered to Purchaser all of the documents and agreements set forth in Section 2.5.

**7.6 Conditions to Company's, Holdco's and Sellers' Obligations to Consummate Third Closing.** The obligation of Company, Holdco and Sellers to consummate the Third Closing shall be subject to the satisfaction, on or prior to the Third Closing, of each of the following conditions, any of which may be waived by Company, Holdco and Sellers in writing:

Section 2.6.

(b)

Deliveries. Purchasers shall have delivered to Company, Sellers and Holdco all of the documents and agreements set forth in

7.7 **Conditions to Obligations of Each Party to Close**. The respective obligations of each party to this Agreement to consummate the Transaction shall be subject to the satisfaction, on or prior to the Closing Date, of each of the following condition(s), any of which may be waived by Purchasers, Seller, Holdco or Company, as applicable, in writing:

(a) <u>No Legal Impediments to Closing</u>. There shall not be in effect any Order issued by any Governmental Authority preventing the consummation of the Transaction, seeking any Damages as a result of the Transaction, or otherwise affecting the right or ability of Purchasers to own, operate or control the Business or the Purchased Assets, nor shall any Proceeding be pending that seeks any of the foregoing. There shall not be any Legal Requirement prohibiting Sellers, Company or Holdco from selling or Purchasers from owning, operating or controlling the Business or the Purchased Assets or that makes this Agreement or the consummation of the Transaction illegal.

(b) <u>Consents of Governmental Authorities</u>. The waiting periods (and any extensions thereof) under any filings required under any applicable Antitrust Law shall have expired or been terminated and all filings required to be made prior to the Closing Date with, and all consents, approvals, permits and authorizations required to be obtained prior to the Closing Date from Governmental Authorities and third parties in connection with the execution and delivery of this Agreement and the consummation of the Transaction will have been made or obtained (as the case may be).

#### ARTICLE VIII

#### TERMINATION

8.1 **Circumstances for Termination**. At any time prior to the Closing, this Agreement may be terminated by written notice explaining the reason for such termination (without prejudice to other remedies which may be available to the parties under this Agreement, at law or in equity):

(a) by the mutual written consent of Purchasers, on the one hand, and Company, Holdco and Sellers, on the other hand;

(b) by either Purchasers, on the one hand, or Company, Holdco and Sellers, on the other hand, if (i) the non-terminating party is in material breach of any material provision of this Agreement and such breach shall not have been cured (to the extent curable) within ten (10) days of receipt by such party of written notice from the terminating party of such breach; and (ii) the terminating party is not, on the date of termination, in material breach of any material provision of this Agreement; and

(c) by either Purchasers, on the one hand, or Company, Holdco and Sellers, on the other hand, if (i) the Closing has not occurred on or prior to the Outside Closing Date for any reason; and (ii) the terminating party is not, on the date of termination, in material breach of any material provision of this Agreement; and

(d) by either Purchasers, on the one hand, or Company, Holdco and Sellers, on the other hand, if (i) satisfaction of a closing condition of the terminating party in <u>Article VII</u> is impossible; and (ii) the terminating party is not, on the date of termination, in material breach of any material provision of this Agreement.

For the avoidance of doubt, the failure of any of the representations and warranties made by Sellers, Company and Holdco in this Agreement to be true and correct in all material respects at any time on or prior to the Closing Date shall be deemed to be a material breach of a material provision of this Agreement.

**8.2** Effect of Termination. If this Agreement is terminated in accordance with <u>Section 8.1</u>, all obligations of the parties hereunder shall terminate, except for the obligations set forth in this <u>Article VIII</u> and <u>Article XI</u>; provided, however, that nothing herein shall relieve any party from liability for the breach of any of its representations, warranties, covenants or agreements set forth in this Agreement.

#### ARTICLE IX

#### INDEMNIFICATION

**9.1** Survival of Representations and Warranties. All representations and warranties of Company, Holdco and Sellers in this Agreement or any other Transaction Agreement shall survive the Closing and remain in full force and effect for a period ending on the date that is eighteen (18) months after the Closing Date (the "*Survival Date*"); provided, however, that (a) any claim for indemnification based upon a breach of any representation and warranty and asserted prior to the Survival Date by written notice in accordance with <u>Section 9.3</u> shall survive until final resolution of such claim; (b) all representations and warranties contained in <u>Section 3.1</u> (Organization, Good Standing, Qualification), <u>Section 3.4</u> (Authority, Binding Nature of Agreements), <u>Section 3.17</u> (Compliance with Laws), <u>Section 3.24</u> (Title; Sufficiency; Condition of Assets), <u>Section 4.8</u> (Finco Shares) and <u>Section 4.9</u> (Finco) shall survive the Closing and continue in full force and effect indefinitely; and (c) the representations and warranties set forth in <u>Section 3.16</u> (Company Benefit Plans), <u>Section 3.20</u> (Environmental Matters) and <u>Section 3.21</u> (Taxes) shall survive the Closing and remain in full force and effect until thirty (30) days after the expiration of the applicable statute of limitations (the representations and warranties set forth in the foregoing clauses (b) and (c) collectively, the "*Specified Reps*"). All of Purchasers' representations and warranties in this Agreement or any other Transaction Agreement shall terminate on the Closing Date, and shall thereafter be of no further force or effect.

(a) Subject to the limitations set forth in this <u>Article IX</u>, Sellers, Company and Holdco shall jointly and severally indemnify, defend and hold harmless Purchasers, their Affiliates and their Representatives (collectively, "*Purchaser Parties*") from any and all Liabilities, losses, damages, Taxes, debts, obligations, claims, costs or expenses, interest, awards, judgments, settlements, orders, fines and penalties (including reasonable attorneys' fees, costs and expenses), whether or not involving a third-party claim (collectively, "*Damages*"), arising out of, relating to or resulting from:

(i) any breach of a representation or warranty of Sellers, Holdco or Company contained in this Agreement or in any other Transaction Agreement;

Agreement;

(ii) any breach of any covenant of Sellers, Holdco or Company contained in this Agreement or in any other Transaction

(iii) any Excluded Asset or Excluded Liability (including Pre-Closing Taxes);

(iv) any untrue statement or alleged untrue statement of a material fact in any registration statement and any prospectus filed pursuant to <u>Section 6.15</u> or any post-effective amendment thereto, or any omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, which untrue statement or alleged untrue statement or omission or alleged omission was made in reliance upon and in conformity with information furnished in writing to Purchasers by Sellers, Holdco or Company for use in connection with such registration statement, prospectus or post-effective amendment; or

(v) any acts or omissions on or prior to the Closing Date of Company, any Company Subsidiary, Sellers or Holdco that relate in any respect to any provision of the U.S. Foreign Corrupt Practices Act of 1977 and the Legal Requirements associated therewith (collectively the "*FCPA*") (whether or not, at the time of any such acts or omissions, the FCPA applied to Sellers, Holdco, Company or the Company Subsidiaries).

(b) Materiality standards or qualifications in any representation, warranty or covenant shall only be taken into account in determining whether a breach of or default in connection with such representation, warranty or covenant (or failure of any representation or warranty to be true and correct) exists, and shall not be taken into account in determining the amount of any Damages with respect to such breach, default or failure to be true and correct.

(c) The representations, warranties, covenants and obligations of Sellers and Company, and the rights and remedies that may be exercised by the Purchaser Parties based on such representations, warranties, covenants and obligations, will not be limited or affected by any investigation conducted by Purchasers or any Purchaser Representative with respect to, or any knowledge acquired (or capable of being acquired) by Purchasers or any Purchasers or any representative with respect to, or any knowledge acquired (or capable of being acquired) by Purchasers or any Purchaser Representative at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to, the accuracy or inaccuracy of or compliance with any such representation, warranty, covenant or obligation. The waiver by Purchasers of any provision of this Agreement will not affect or limit the provisions of this Article IX.

59

#### 9.3 Procedures for Indemnification.

(a) Notice. Promptly after the appropriate management personnel of any Purchaser Party entitled to indemnification hereunder (an "Indemnified Party") (i) become aware of circumstances that have resulted in, or which are reasonably likely to result in, Damages for which the Indemnified Party intends to seek indemnification for or (ii) receive written notice of any third-party demand, claim or circumstances which, with the lapse of time, the giving of notice or both, are reasonably likely to give rise to a claim or the commencement (or threatened commencement) of any Proceeding that may result in Damages for which the Indemnified Party may be indemnified hereunder (an "Asserted Liability"), the Indemnified Party shall give notice thereof (a "Claims Notice") to any other party or parties obligated to provide indemnification hereunder (an "Indemnifying Party"). The Claims Notice shall describe the Damages or the Asserted Liability in reasonable detail to the extent known, and shall indicate the amount (estimated, if necessary and if reasonably capable of estimation) of the Damages that has been or that may be suffered by the Indemnified Party. The Claims Notice may be amended on one or more occasions with respect to the amount of the Asserted Liability or the Damages at any time prior to final resolution of the obligation to indemnify relating to the Asserted Liability or the Damages. If a Claims Notice is not provided promptly as required by this <u>Section 9.3</u>, the Indemnified Party nonetheless shall be entitled to indemnification or reimbursement by the Indemnifying Party except to the extent the Indemnifying Party has been materially prejudiced by such late receipt of a Claims Notice. Any Asserted Liability for indemnification for any matter not involving a third-party Proceeding shall be paid promptly after delivery to the Indemnifying Party of a Claims Notice relating to such Asserted Liability.

(b) Opportunity to Contest. The Indemnifying Party may, provided that it has acknowledged its responsibility to indemnify with respect to an Asserted Liability, elect to compromise or contest, at its own expense and with counsel of its choice reasonably acceptable to the Indemnified Party, such Asserted Liability; provided, that the Indemnifying Party shall not have the right to assume or continue the defense of any Asserted Liability if (i) in the reasonable opinion of the Indemnified Party, counsel for the Indemnifying Party could not adequately represent the interests of the Indemnified Party because its interests could be in conflict with those of the Indemnifying Party; (ii) such Proceeding is reasonably likely to have a material adverse effect on any other matter beyond the scope or limits of the indemnification obligation of the Indemnifying Party; or (iii) the Indemnifying Party shall not have assumed and diligently continued the defense of a Proceeding in a timely fashion and in a manner reasonably acceptable to Indemnified Party. In addition, if the Indemnifying Party has assumed the defense of a Proceeding, Indemnifying Party shall take all reasonably necessary steps to defend the relevant Asserted Liability to conclusion or settlement, keep the Indemnified Party informed of the progress of any such Asserted Liability, permit the Indemnified Party at its own expense to participate in such defense and provide the Indemnified Party with reasonable access to all reasonably relevant information and documentation relating to the Asserted Liability and the Indemnifying Party's defense thereof. If the Indemnified Party elects to compromise or contest such Asserted Liability, it shall, within 20 days (or sooner, if the nature of the Asserted Liability so requires), notify the Indemnified Party of its intent to do so by sending a notice to the Indemnified Party (a "**Contest Notice**"); and, in any such event, the Indemnified Party shall reasonably cooperate, at the expense of the Indemnifying Party, in the com

Asserted Liability, fails to notify, in a timely manner, the Indemnified Party of its election as herein provided or contests its obligation to indemnify under this Agreement with respect to such Asserted Liability, the Indemnified Party shall have the right to pay, compromise or contest such Asserted Liability on behalf

of and for the account and risk of the Indemnifying Party. Anything in this <u>Section 9.3(b)</u> to the contrary notwithstanding, (i) the Indemnified Party shall have the right, at its own cost and for its own account, to compromise or contest any Asserted Liability, (ii) the Indemnifying Party shall not, without the Indemnified Party's written consent, settle or compromise any Asserted Liability for Taxes, and (iii) the Indemnifying Party shall not, without the Indemnified Party's written consent, settle or compromise any Asserted Liability not for Taxes or consent to entry of any judgment with respect to such Asserted Liability which does not include an unconditional term releasing the Indemnified Party and its Affiliates from all Liability in respect of such Asserted Liability. In any event, the Indemnified Party shall reasonably cooperate with each other as to all Asserted Liabilities (at the expense of the Indemnifying Party), shall make available to each other, as reasonably requested, all information, records, and documents reasonably related to all Asserted Liabilities and shall preserve all such information, records, and documents until the termination of any Asserted Liability. The Indemnified Party also shall make available to each other, as reasonably requested (and at the reasonable expense of Indemnifying Party), their respective personnel, agents, and other representatives who are responsible for preparing or maintaining information, records, or other documents, or who may have particular knowledge with respect to any Asserted Liability.

# 9.4 Limitations on Indemnification.

(a) No amount shall be payable by Sellers, Company or Holdco to any Indemnified Party pursuant to <u>Section 9.2(a)(i)</u> (other than with respect to Damages related to breach of a Specified Rep) unless the aggregate amount of Damages indemnifiable by Sellers, Company or Holdco under <u>Section 9.2(a)(i)</u> (other than with respect to Damages related to breach of a Specified Rep) exceeds an amount (the "*Sellers' Basket*") equal to Seven Hundred Thousand Dollars (\$700,000), at which point Sellers, Company and Holdco shall become liable for all such Damages, including the Damages reflected in Sellers' Basket.

(b) Notwithstanding anything to the contrary contained in this Agreement, the maximum amount of aggregate indemnifiable Damages which may be recovered from Sellers, Company or Holdco under <u>Section 9.2(a)(i)</u> (other than with respect to Damages related to breach of a Specified Rep) shall be an amount equal to Twenty-One Million Dollars (\$21,000,000) ("*Sellers' Indemnification Cap*").

(c) The limitations set forth in subsections (a) and (b) of this <u>Section 9.4</u> shall not apply to any Seller, Holdco or Company indemnification obligation (x) arising out of, relating to or resulting from fraud or intentional misrepresentation by Holdco, Company or any Seller; (y) arising out of, relating to or resulting from a breach of any of the Specified Reps; or (z) arising out of, relating to or resulting under <u>Section 9.2(a)(ii)</u>, (iii), (iv) or (v).

61

(d) Notwithstanding anything to the contrary contained in this Agreement, the maximum amount of aggregate indemnifiable Damages which may be recovered from Sellers, Company or Holdco arising out of, relating to or resulting under <u>Section 9.2(a)(v)</u> shall not exceed an amount equal to the lesser of (i) fifty percent (50%) of the total Damages of the Purchaser Parties arising out of, relating to or resulting under <u>Section 9.2(a)(v)</u> or (ii) the Sellers' Indemnification Cap.

**9.5 Offset**. To the extent an Indemnified Party has provided a Claims Notice before a payment is due under Future Payment Note #3 and/or Section 1.9(c)(iv), and such Claims Notice is not resolved, Purchasers shall be permitted to withhold from such payment, and any subsequent payments, until such Claims Notice is resolved, a reasonable amount as a reserve to cover the potential indemnification obligations of Sellers, Company and Holdco with respect to such Claims Notice. Any amounts withheld by Purchasers pursuant to this <u>Section 9.5</u>, other than amounts otherwise owed by Sellers, Company and Holdco to satisfy Sellers', Company's and Holdco's indemnification obligations under this Agreement, shall be promptly released to Sellers, Company and Holdco following the resolution of any such Claims Notice. Notwithstanding anything to the contrary herein, neither the exercise nor non-exercise of Purchasers' rights to offset under this <u>Section 9.5</u> will constitute an election of remedies or limit Purchasers in any way in the enforcement of any other remedies that may be available to it under this Agreement.

9.6 **Remedies Cumulative**. The remedies provided in this Agreement shall be cumulative and shall not preclude any party from asserting any other right, or seeking any other remedies, against the other party.

**9.7 Nature of Indemnification Payments**. Any indemnification payment made by Sellers pursuant to this <u>Article IX</u> shall be deemed by the parties as an adjustment to the Purchase Price.

#### ARTICLE X

# TAX MATTERS

**10.1 Amended Returns; Tax Elections.** Except consistently with a resolved audit, Sellers and Company may not amend (or permit the amendment of) a Tax Return of any Company Subsidiary, or make, amend or revoke (or permit the making, amendment, or revocation of) any Tax election of any Company Subsidiary, in each case, with respect to a taxable period beginning before the Closing Date without the prior written consent of Purchasers.

**10.2 Straddle Period Allocations.** For purposes of this Agreement, Taxes of any of the Company Subsidiaries incurred with respect to a taxable period that includes but does not end on the Closing Date, shall be allocated to the portion of the taxable period ending on the Closing Date (i) except as provided in (ii) and (iii) below, to the extent feasible, on a specific identification basis, according to the date of the event or transaction giving rise to the Tax, and (ii) except as provided in (iii) below, with respect to periodically assessed ad valorem Taxes and Taxes not otherwise feasibly allocable to specific transactions or events, in proportion to the number of days in such taxable period occurring through the Closing Date compared to the total number of days in such taxable period, and (iii) in the case of any Tax based upon or related to income or receipts, in an amount equal to the Tax which would be payable if the relevant taxable period ended on the Closing Date.

**10.3 Tax Elections**. On or before the Closing Date, Sellers and Company shall provide Purchasers with an executed Section 85 Election, Section 22 Election, Section 20(24), Section 56.4(3) Election and Section 167 Election. Such elections will be held in escrow by counsel to Company until

completion of the procedures set forth in this Section 10.5.

**10.4 Termination of Tax Sharing Agreements.** All Tax sharing, allocation, indemnity or similar agreements with respect to or involving any of the Company Subsidiaries shall be terminated as of the Closing Date and, after the Closing Date, no Company Subsidiary shall be bound thereby or have any liability thereunder.

**10.5 Allocation of Purchase Price**. As soon as practicable after the Final Resolution Date, Purchasers shall provide to Seller, Company and Holdco for their review and approval (which approval shall not be unreasonably withheld) a proposed allocation of the Purchase Price, as adjusted pursuant to <u>Section 1.10</u>, among the various classes of Purchased Assets. None of the parties shall take a position on any Tax Return, before any Governmental Authority or in any judicial proceeding that is in any manner inconsistent with such allocation without the written consent of the other parties to this Agreement or unless specifically required pursuant to a determination by an applicable Governmental Authority. The parties shall promptly advise each other of the existence of any Tax audit, controversy or litigation related to any allocation hereunder. Any dispute arising under this <u>Section 10.5</u> shall be resolved in the same manner as disputes are resolved pursuant to <u>Section 1.10(c)</u> above.

**10.6 Cooperation, Access to Information, and Record Retention**. If any of Purchasers, Company, or Sellers receives any notice of a pending or threatened Tax assessment, adjustment, audit, litigation or other proceeding relating to Company or any of Company Subsidiaries, which may give rise to a liability of another party hereto (including but not limited to indemnification obligations hereunder), (i) the first party shall promptly notify such other party within ten (10) business days of receiving such notice, and (ii) both parties shall keep each other informed on a regular basis regarding the status thereof. Purchasers, Sellers, and Company shall cooperate, and cause their representatives and Affiliates to cooperate, to the extent reasonably requested by any other party hereto in connection with the preparation and filing of Tax Returns and any audit, litigation or other party's request) the provision of records and information which are reasonably relevant to any such Tax Return, audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Company, Company Subsidiaries, Purchasers and Sellers shall (i) retain all books and records with respect to Taxes and Tax Returns of Company and Company Subsidiaries for any taxable period beginning before the Closing Date until the expiration of the statute of limitations (and, to the extent notified, any extensions thereof) with respect thereto, (ii) abide by all record retention agreements entered into with any taxing authority, and (iii) give the other party hereto reasonable written notice prior to transferring, destroying or discarding any such books and records.

63

#### ARTICLE XI

#### MISCELLANEOUS PROVISIONS

**11.1 Expenses.** Whether or not the Transaction is consummated, each party shall pay its own fees, costs and expenses in connection with this Agreement and the Transaction (including the fees, costs and expenses of its advisers, accountants and legal counsel); provided, that Purchasers shall, following the Closing, reimburse Company and Sellers for (i) up to \$100,000 of reasonable fees and expenses of Company's and Sellers' legal counsel attributable to the Transaction structure, and (ii) the reasonable additional fees of Company's and Sellers' accountant directly attributable to work in connection with Purchasers' structure for the Transaction.

Interpretation. None of this Agreement or the Transaction Agreements or any uncertainty or ambiguity herein or therein shall be 11.2 construed or resolved against any party hereto, whether under any rule of construction or otherwise. No party to this Agreement shall be considered the draftsperson, and this Agreement and the Transaction Agreements have been reviewed, negotiated and accepted by all parties and their attorneys, and any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not be applied in the construction or interpretation of this Agreement or any of the Transaction Agreements. Unless the context shall otherwise require, words using the singular or plural number shall also include the plural or singular number, respectively. The words "include," "includes" and "including" herein or in any Transaction Agreement shall be deemed to be followed by the phrase "without limitation," and the word "or" shall include the meaning "either or both." Unless the context shall otherwise require, any reference herein or in any Transaction Agreement to any agreement or other instrument or statute or regulation is to such agreement, instrument, statute or regulation as amended and supplemented from time to time (and, in the case of a statute or regulation, to any successor provision). Unless the context otherwise provides, all pronouns used herein shall be deemed to refer to the masculine, feminine or neuter gender as the context requires. All references in this Agreement or any other Transaction Agreement to "Section," "Subsection" or "Article" (or similar references) shall be deemed to be references to a section, subsection or article of the agreement in which they occur unless the context otherwise requires. The table of contents and the captions and other headings contained in this Agreement or any Transaction Agreement as to the contents of particular articles, sections, paragraphs or other subdivisions contained herein or therein have been included for convenience of reference only and shall not, in any way, be construed as part of this Agreement or as limitations on the scope of the particular articles, sections, paragraphs or other subdivisions to which they refer and shall not affect the interpretation or meaning of this Agreement or any Transaction Agreement. Unless otherwise indicated, all references in this Agreement or any other Transaction Agreement to "Dollars" and/or "\$" shall be deemed to be references to U.S. Dollars.

**11.3** Further Assurances. Each party agrees (a) to furnish upon request to each other party such further information, (b) to execute and deliver to each other party such other documents, and (c) to do such other acts and things, all as another party may reasonably request for the purpose of carrying out the intent of this Agreement and the Transaction.

#### 11.4 Notices.

(a) All notices, requests, demands and other communications hereunder shall be either (i) delivered in person, (ii) sent by overnight courier service or other express commercial delivery service, or (iii) sent by facsimile with confirmation of receipt and, in each case, addressed as follows:

I.M.W. Industries Ltd. 43676 Progress Way Chilliwack, B.C. VR20C3

Facsimile: (604) 792-3806 Attention: Bradley N. Miller If to Sellers or Holdco: Bradley N. Miller 24641 87 Avenue Langley, B.C. V1M 2R3 Facsimile: (604) 888-5500 with copies to: Access Law Group #1700 — 1185 West Georgia St. Vancouver, B.C. V6E 4E6 Facsimile: (604) 484-4363 Attention: Peter Mogan If to Purchasers: Clean Energy 3020 Old Ranch Parkway, Suite 400 Seal Beach, CA 90740 Facsimile: (562) 493-4532 Attention: Harrison Clay, General Counsel with copies to: Morrison & Foerster LLP 12531 High Bluff Drive San Diego, CA 92130 Facsimile: (858) 720-5125 Attention: Nathan Jensen Morrison & Foerster LLP 425 Market Street San Francisco, CA 94105 Facsimile: (415) 268-7522 Attention: Andrew Thorpe

(b) All notices, requests, instructions or documents given to any party in accordance with this <u>Section 11.4</u> shall be deemed to have been given on the date of mailing or transmission, whether delivered by hand, by overnight courier service, or by facsimile, with confirmation of receipt on such date.

(c) Any party hereto may change its address specified for notices herein by designating a new address by notice given in accordance with this Section 11.4.

65

**11.5** Entire Agreement. This Agreement, the schedules, the exhibits and the Transaction Agreements constitute the entire agreement between the parties relating to the subject matter hereof and thereof and supersede all prior oral and written understandings, all contemporaneous oral negotiations and discussions, and all other writings and agreements relating to the subject matter of this Agreement.

**11.6 Modifications, Amendments and Waivers**. This Agreement and any Transaction Agreements cannot be amended or changed nor any performance, term, or condition waived in whole or in part, except by a writing signed by the party against whom enforcement of the amendment, change or waiver is sought. No delay or failure on the part of any party in exercising any rights hereunder, and no partial or single exercise thereof, will constitute a waiver of such rights or of any other rights hereunder.

**11.7 Successors and Assigns**. This Agreement cannot be assigned by any party without the prior written consent of the other parties hereto and any assignment in violation of this <u>Section 11.7</u> shall be null and void; provided that Purchasers may assign this Agreement and any Transaction Agreement (i) to Affiliates of Purchasers or (ii) following Closing, in connection with any merger, amalgamation, arrangement, sale of any Purchaser's assets or similar corporate transaction, including a sale or transfer of the Purchased Assets.

**11.8 Governing Law**. This Agreement is to be construed in accordance with and governed by the internal laws of the State of Delaware or any similar successor provision, without giving effect to any choice of law rule that would cause the application of the laws of any jurisdiction other than the internal laws of the State of Delaware to the rights and duties of the parties.

**11.9 Severability**. Should any one or more of the provisions of this Agreement be determined to be invalid, illegal or unenforceable, such invalid, illegal or unenforceable provisions shall be deemed severed herefrom, and the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby. The parties shall replace the invalid, illegal or unenforceable provisions with valid provisions, the economic effect of which comes as close as practicable to that of the invalid, illegal or unenforceable provisions.

**11.10 Specific Performance**. Each party agrees that irreparable harm, for which there will be no adequate remedy at law and for which the ascertainment of monetary damages would be difficult, would occur in the event any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Each party accordingly agrees that the other parties shall be entitled to specifically enforce this Agreement and to obtain an injunction or injunctions to prevent breaches of the provisions of this Agreement or any other Transaction Agreement and to enforce specifically the terms and provisions hereof or thereof, in each instance without being required to post bond or other security and in addition to, and without having to prove the adequacy of, other remedies at law.

**11.11 Counterparts**. This Agreement may be executed in any number of counterparts which may be delivered by facsimile, each of which shall be deemed to be an original, but all of which counterparts shall together constitute one and the same instrument.

# 11.12 Dispute Resolution.

(a) Except as set forth in <u>Section 1.9</u>, <u>Section 1.10</u> and <u>Section 10.5</u>, any controversy, claim or dispute of whatever nature arising between Purchasers, on the one hand, and Sellers, Holdco and Company, on the other hand (each, a "*Disputing Party*") under this Agreement or in connection with the Transaction, including those arising out of or relating to the breach, termination, enforceability, scope, validity, or making of this Agreement, whether such claim existed prior to or arises on or after the Closing Date (a "*Dispute*"), shall be resolved by good faith negotiations among the Disputing Parties, such negotiation not to exceed a period of thirty (30) consecutive days (the "*Negotiation Period*"). In the event a Dispute remains unresolved following the Negotiation Period, a Disputing Party may give written notice (a "*Dispute Notice*") to the other Disputing Party setting forth the nature of the Dispute. Following delivery of a Dispute Notice. The mediator shall have sixty (60) days from the time of his or her appointment to meet with the parties and help them resolve the dispute, unless the parties mutually consent to an extension of the deadline. The costs of the mediation, including fees and expenses, shall be borne equally by the parties, and the mediation shall take place in the locality of the party who did not give the Dispute Notice.

(b) If the Dispute is not resolved pursuant to <u>Section 11.12(a)</u>, any Dispute relating to this Agreement shall be brought solely in the federal courts located in Orange County, California and all obligations to personal jurisdiction and venue in any action, suit or proceeding so commenced are hereby expressly waived by all parties hereto; provided, however, that, a party may commence any action or proceeding in a court other than as set forth above solely for the purpose of enforcing an order or judgment issued by one of such courts. The parties waive personal service of any and all process on each of them and consent that all such service of process shall be made in the manner, to the party and at the address set forth in <u>Section 11.4</u> of this Agreement, and service so made shall be complete as stated in such section. To the extent not prohibited by applicable Legal Requirements or court rule, each party hereby waives and agrees not to assert, by way of motion, as a defense or otherwise in any such proceeding, any claim (i) that it is not subject to the jurisdiction of the above-named courts, (ii) that the proceeding is brought in an inconvenient forum, (iii) that it is immune from any legal process with respect to itself or its property, (iv) that the venue of the proceeding is improper or (v) that this Agreement or the subject matter hereof or thereof may not be enforced in or by such courts.

**11.13 Attorneys' Fees, Costs and Expenses.** In any Proceeding to enforce the terms of this Agreement or the other documents executed in connection herewith, the prevailing party shall be entitled to recover reasonable attorneys' fees, costs and expenses incurred in connection with such enforcement Proceeding.

**11.14 No Third-Party Beneficiaries**. The terms and provisions of this Agreement are intended solely for the benefit of each party hereto and their respective successors and permitted assigns, and the parties do not intend to confer third-party beneficiary rights upon any other Person.

67

**11.15 Privacy**. Prior to the First Closing, none of the parties shall use the Disclosed Personal Information for any purposes other than those related to the performance of this Agreement and the completion of the Transaction. Each of the parties acknowledges and confirms that the disclosure of Personal Information is necessary for the purposes of determining if the parties shall proceed with the Transaction, and that the disclosure of Personal Information relates solely to the carrying on of the business, or the completion of the Transaction. Purchasers shall at all times keep strictly confidential all Disclosed Personal Information provided to them. If the First Closing does not occur, Purchasers shall cease all use of the Disclosed Personal Information acquired by Purchasers in connection with this Agreement and, at Sellers' request, will return to Sellers or destroy in a secure manner, the Disclosed Personal Information.

#### [Signatures Follow On a Separate Page]

68

IN WITNESS WHEREOF, each of the parties has caused this Agreement to be executed on its behalf by their respective officers thereunto duly authorized all as of the date first written above.

"Purchasers"

Clean Energy By: /s/ Andrew J. Littlefair Name: Andrew J. Littlefair Title: President and Chief Executive Officer

0884808 B.C. Ltd. By: /s/ Barclay F. Corbus Name: Barclay F. Corbus Title: President

0884810 B.C. Ltd. By: /s/ Barclay F. Corbus Name: Barclay F. Corbus Title: President

"Company" I.M.W. Industries Ltd. By: /s/ Bradley N. Miller Name: Bradley N. Miller Title: President

"Holdco" 652322 B.C. Ltd. By: /s/ Bradley N. Miller Name: Bradley N. Miller Title: President

"*Sellers*" Bradley N. Miller By: /s/ Bradley N. Miller

Miller Family Trust By: /s/ Bradley N. Miller Name: Bradley N. Miller Title: Sole Trustee

SIGNATURE PAGE

# EXHIBIT A

#### **CERTAIN DEFINITIONS**

"Accounts Payable" shall mean and include all accounts payable and other amounts owed to trade creditors of Company and any Company Subsidiary.

"Accounts Receivable" shall mean and include all accounts receivable, notes receivable and other monies due to Company for sales and deliveries of goods, performance of services and other transactions (whether or not on the books of Company).

"Additional Units" means, with respect to each Earn-Out Period, any CNG Units sold in the U.S. in excess of the Base Units for such Earn-Out Period.

"Adjusted Gross Profit" shall mean the Gross Profit of the Business for each Earn-Out Period, <u>minus</u> (i) the Gross Profit of the Business relating to sales of CNG Units in the U.S. during such Earn-Out Period, <u>plus</u> (ii) an amount equal to 38% of the sales price for the Base Units sold in the U.S. during such Earn-Out Period, <u>plus</u> (iii) an amount equal to 19% of the sales price for the Additional Units sold in the U.S. during such Earn-Out Period. In addition, Gross Profit generated during any Earn-Out Period from any Post-Closing Acquisition will be included in the calculation of Adjusted Gross Profit for such Earn-Out Period, provided that a reasonable amount will be deducted from such Adjusted Gross Profit for costs and expenses relating to any such Post-Closing Acquisition.

"Adjusted Gross Profit Target" shall mean each of the Year 1 Earn-Out Target, the Year 2 Earn-Out Target, the Year 3 Earn-Out Target and the Year 4 Earn-Out Target.

"Affiliate" shall mean, with respect to any Person, any member of the immediate family of such Person (including spouse, brother, sister, descendant, ancestor or in-law) or any officer, director or holder of 5% or more of the outstanding equity interests of such Person or any corporation, partnership, trust or other entity in which such Person or any such family member has a five percent (5%) or greater interest or is a director, officer, partner or trustee. The term Affiliate shall also include, with respect to any Person, any other Person directly or indirectly controlling, controlled by, or under common control with such Person.

"AGP Resolution Date" shall have the meaning specified in Section 1.9(d).

"AGP Statement" shall have the meaning specified in Section 1.9(d).

"AGP Statement Dispute Notice" shall have the meaning specified in Section 1.9(e).

"*Agreement*" shall mean the Asset Purchase Agreement to which this <u>Exhibit A</u> is attached (including the Company Disclosure Schedule and all other schedules and exhibits attached hereto), as it may be amended from time to time.

"Antitrust Law" shall have the meaning specified in Section 6.7(b).

"Asserted Liability" shall have the meaning specified in Section 9.3(a).

1

"Assumed Liabilities" shall have the meaning specified in Section 1.5.

"Audited Financial Statements" means the audited consolidated balance sheet, and the related statements of operations, changes in shareholders' equity, and cash flows, of Company and the Company Subsidiaries as of and for the fiscal year ended April 30, 2010, together with a reconciliation to GAAP.

"Bangladesh Closing Net Working Capital" shall mean the actual net working capital of IMW Bangladesh as of the Closing Date, calculated in accordance with applicable accounting principles generally accepted in Bangladesh ("Bangladesh GAAP").

"Bangladesh Closing Net Working Capital Adjusted Amount" shall mean the difference, positive or negative, between the Bangladesh Estimated Net Working Capital and the Bangladesh Closing Net Working Capital, converted to U.S. Dollars using the applicable exchange rate in effect on the Closing Date.

"Bangladesh Estimated Net Working Capital" shall mean the estimated net working capital of IMW Bangladesh as of the Closing Date, calculated in accordance with Bangladesh GAAP.

"*Bangladesh Estimated Net Working Capital Adjusted Amount*" shall mean the difference, positive or negative, between the Bangladesh Estimated Net Working Capital and the Bangladesh Historical Net Working Capital, converted to U.S. Dollars using the applicable exchange rate in effect on the date the Preliminary Closing Statement is delivered to Purchasers.

"Bangladesh Historical Net Working Capital" shall mean the net working capital of IMW Bangladesh as of the fiscal year ended June 30, 2009, calculated in accordance with Bangladesh GAAP.

"*Base Units*" shall mean (i) with respect to the Year 1 Earn-Out Period, forty (40) CNG Units, (ii) with respect to the Year 2 Earn-Out Period, forty-eight (48) CNG Units, (iii) with respect to the Year 3 Earn-Out Period, fifty-eight (58) CNG Units, and (iv) with respect to the Year 4 Earn-Out Period, sixty-nine (69) CNG Units.

"Books and Records" shall have the meaning set forth in Section 1.1(m).

"Business" shall have the meaning set forth in the first Recital.

"Business Day" means any day other than (i) a Saturday or a Sunday or (ii) a day on which banking and savings and loan institutions are authorized or required by law to be closed in Orange County, California.

"*Canadian AcqCo*" shall have the meaning set forth in the preamble and shall also mean any successor to Canadian AcqCo as a result of an amalgamation after the Closing.

"*Canadian GAAP*" means accounting principles generally accepted as set-out in the Handbook of the Canadian Institute of Chartered Accountants at the relevant time applied on a consistent basis.

2

"*Canadian OpCo*" shall have the meaning set forth in the preamble.

"*Change of Control*" means, with respect to CLNE, the acquisition, by a Person who is not an Affiliate of CLNE at the time of such acquisition, of CLNE Stock representing more than fifty-one percent (51%) of the voting power of CLNE; provided that any acquisition in connection with estate planning, tax planning, philanthropic endeavors or other related transactions shall not be deemed a Change of Control under this Agreement.

"China Closing Net Working Capital" shall mean the actual net working capital of IMW China as of the Closing Date, calculated in accordance with accounting principles generally accepted in China ("Chinese GAAP").

"China Closing Net Working Capital Adjusted Amount" shall mean the difference, positive or negative, between the China Estimated Net Working Capital and the China Closing Net Working Capital, converted to U.S. Dollars using the applicable exchange rate in effect on the Closing Date.

"China Estimated Net Working Capital" shall mean the estimated net working capital of IMW China as of the Closing Date, calculated in accordance with Chinese GAAP.

"*China Historical Net Working Capital*" shall mean the net working capital of IMW China as of the fiscal year ended December 31, 2009, calculated in accordance with Chinese GAAP.

"China Estimated Net Working Capital Adjusted Amount" shall mean the difference, positive or negative, between the China Estimated Net Working Capital and the China Historical Net Working Capital, converted to U.S. Dollars using the applicable exchange rate in effect on the date the Preliminary Closing Statement is delivered to Purchasers.

"Claims Notice" shall have the meaning set forth in Section 9.3(a).

- "*Clean Energy*" shall have the meaning set forth in the preamble.
- "CLNE" means Clean Energy Fuels Corp., a Delaware corporation.
- "CLNE Shares" shall have the meaning specified in Section 1.9(c)(i).
- "CLNE Stock" means the common stock of CLNE.
- "Closing" shall have the meaning specified in Section 1.8(c).
- "Closing Date" shall have the meaning specified in Section 1.8(a).
- "Closing Net Working Capital" shall have the meaning specified in Section 1.10(b).
- "CNG" shall mean compressed natural gas.

"CNG Units" shall mean CNG compressors or related systems.

"Colombia Closing Net Working Capital" shall mean the actual net working capital of IMW Colombia as of the Closing Date, calculated in accordance with accounting principles generally accepted in Colombia ("Colombia GAAP").

"*Colombia Closing Net Working Capital Adjusted Amount*" shall mean the difference, positive or negative, between the Colombia Estimated Net Working Capital and the Colombia Closing Net Working Capital, converted to U.S. Dollars using the applicable exchange rate in effect on the Closing Date.

"Colombia Estimated Net Working Capital" shall mean the estimated net working capital of IMW Colombia as of the Closing Date, calculated in accordance with Colombia GAAP.

"Colombia Estimated Net Working Capital Adjusted Amount" shall mean the difference, positive or negative, between the Colombia Estimated Net Working Capital and the Colombia Historical Net Working Capital, converted to U.S. Dollars using the applicable exchange rate in effect on the date the Preliminary Closing Statement is delivered to Purchasers.

"Colombia Historical Net Working Capital" shall mean the net working capital of IMW Colombia as of the fiscal year ended December 31, 2009, calculated in accordance with Colombia GAAP.

"Company" shall mean I.M.W. Industries Ltd., a British Columbia corporation.

"Company Benefit Plans" shall have the meaning specified in Section 3.16(a).

"Company Claims" shall have the meaning specified in Section 1.1(l).

"Company Closing Net Working Capital" shall mean the actual net working capital of Company as of the Closing Date, calculated in the manner in which the Company Historical Net Working Capital was calculated.

"*Company Closing Net Working Capital Adjusted Amount*" shall mean the difference, positive or negative, between the Company Estimated Net Working Capital and the Company Closing Net Working Capital, converted to U.S. Dollars using the applicable exchange rate in effect on the Closing Date.

"Company Contracts" shall have the meaning specified in Section 3.13(a).

"Company Disclosure Schedule" shall have the meaning specified in Article III.

"*Company Estimated Net Working Capital*" shall mean the estimated net working capital of Company as of the Closing Date, calculated in the manner in which the Company Historical Net Working Capital was calculated.

"Company Estimated Net Working Capital Adjusted Amount" shall mean the difference, positive or negative, between the Company Estimated Net Working Capital and the Company Historical Net Working Capital, converted to U.S. Dollars using the applicable exchange rate in effect on the date the Preliminary Closing Statement is delivered to Purchasers.

4

"*Company Historical Net Working Capital*" shall mean the Three Million Five Hundred Seventeen Thousand Eight Hundred Canadian Dollar (CDN\$3,517,800) net working capital of Company calculated as set forth on <u>Exhibit C</u> attached to the Agreement.

"Company Financial Statements" shall have the meaning specified in Section 3.7(a).

"*Company IP Assets*" shall mean all Intellectual Property Rights and Technology that Company or any Company Subsidiary owns or has a license to, excluding licenses from customers that only allow Company or any Company Subsidiary to manufacture and supply to such customers.

"Company Products" shall mean all products and services manufactured, made, designed, maintained, supported, developed, sold, licensed, marketed, or otherwise distributed or provided (or planned or envisioned to be manufactured, made, designed, maintained, supported, developed, sold, licensed, marketed, or otherwise distributed or provided) by or for Company or any Company Subsidiary (including all versions, models and releases thereof, whether already distributed or provided, under development, planned or conceived, or otherwise), together with any related materials, information or data, including, without limitation, the names, numbers (e.g., part numbers) and packaging associated with such products and services.

"Company Subsidiaries" shall have the meaning specified in <u>Section 3.6</u>. For purposes of this Agreement, Finco shall be a Company Subsidiary.

"Competing Party" shall have the meaning specified in Section 6.3.

"Competing Transaction" shall have the meaning specified in Section 6.3.

"Confidential Information" shall mean all Trade Secrets and other confidential and/or proprietary information of a Person, including information derived from reports, investigations, research, work in progress, codes, marketing and sales programs, manufacturing processes, financial projections, cost summaries, pricing formula, contract analyses, financial information, projections, confidential filings with any state or federal agency, and all other confidential concepts, methods of doing business, ideas, materials or information prepared or performed for, by or on behalf of such Person by its employees, officers, directors, agents, Representatives, or consultants.

"Consent" shall mean any approval, consent, ratification, permission, waiver or authorization (including any Governmental Approval).

"Contest Notice" shall have the meaning specified in Section 9.3(b).

"*Contract*" shall mean any agreement, contract, consensual obligation, promise, understanding, arrangement, commitment or undertaking of any nature (whether written or oral and whether express or implied), whether or not legally binding.

"Copyright Assignment" has the meaning specified in Section 2.1(b)(iii).

"*Copyrights*" shall mean all copyrights, including in and to works of authorship and all other rights corresponding thereto throughout the world, whether published or unpublished, including rights to prepare, reproduce, perform, display and distribute copyrighted works and copies, compilations and derivative works thereof.

"Damages" shall have the meaning specified in Section 9.2(a).

"Deposits and Advances" shall have the meaning specified in Section 1.1(i).

"*Depreciable Assets Shares*" shall mean a number of Canadian OpCo preferred shares having a value equal to the value of the depreciable assets included in the Purchased Assets.

"Designated Employees" shall have the meaning specified in <u>Section 6.14(b)</u>.

"Disclosed Personal Information" means any Personal Information disclosed to Purchasers pursuant to this Agreement or the Transaction by or on behalf of Sellers.

"Disclosing Party" shall have the meaning specified in <u>Section 6.8(b)</u>.

"Dispute" shall have the meaning specified in Section 11.12(a).

"Dispute Notice" shall have the meaning specified in Section 11.12(a).

"Disputing Party" shall have the meaning specified in Section 11.12(a).

"Domain Name Assignment" shall have the meaning specified in Section 2.1(b)(iv).

"Domain Names" shall mean: imwind.com; imwindustries.com; imw.bc.ca; imw.ca.

"Earn-Out Payment(s)" shall have the meaning specified in Section 1.9(c)(iv).

"*Earn-Out Period*" shall mean each of (i) the period beginning on April 1, 2010 and ending on March 31, 2011 (the "*Year 1 Earn-Out Period*"), (ii) the period beginning on April 1, 2011 and ending on March 31, 2012 (the "*Year 2 Earn-Out Period*"), (iii) the period beginning on April 1, 2012 and ending on March 31, 2013 (the "*Year 3 Earn-Out Period*") and (iv) the period beginning on April 1, 2013 and ending on March 31, 2014 (the "*Year 4 Earn-Out Period*"). *(Earn-Out Period*"), (iii) the period beginning on April 1, 2014 (the "*Year 4 Earn-Out Period*").

"Earnout Shares" shall mean a number of Canadian OpCo preferred shares having a value equal to the value of the Earn-Out Payments.

"Employee Start Date" shall have the meaning specified in Section 6.14(c).

"Employment Agreements" shall have the meaning specified in Section 2.1(f).

"*Encumbrance*" shall mean any lien, pledge, hypothecation, charge, mortgage, security interest, encumbrance, equity, trust, equitable interest, claim, preference, right of possession, lease, tenancy, license, encroachment, covenant, infringement, interference, Order, proxy, option, right of first refusal, preemptive right, community property interest, legend, defect, impediment, exception, reservation, limitation, impairment, imperfection of title, condition or restriction of any nature (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, any restriction on the use of any asset and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

6

"*Entity*" shall mean any corporation (including any non-profit corporation), partnership, general partnership, limited partnership, limited liability partnership, joint venture, unincorporated association, estate, trust or company (including any limited liability company, unlimited liability company or joint stock company).

"Environmental and Safety Laws" shall have the meaning specified in Section 3.20(a).

"Estimated Net Working Capital" shall have the meaning specified in Section 1.10(a).

"Estimated Net Working Capital Adjustment Amount" shall have the meaning specified in Section 1.10(a).

"Estoppel Letters" shall have the meaning specified in Section 2.1(d).

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

"Excluded Assets" shall have the meaning specified in Section 1.4.

"Excluded Liabilities" shall have the meaning specified in Section 1.6.

"*Facilities*" shall have the meaning specified in <u>Section 3.20(a)</u>.

"FCPA" shall have the meaning specified in Section 9.2(a)(v).

"Final Closing Statement" shall have the meaning specified in <u>Section 1.10(b)</u>.

"Final Closing Statement Dispute Notice" shall have the meaning specified in Section 1.10(c).

"Final Net Working Capital" shall have the meaning specified in Section 1.10(c).

"Final Resolution Date" shall have the meaning specified in Section 1.10(b).

"Financial Statements" shall have the meaning specified in Section 3.7(d).

"Finco" shall mean a British Columbia corporation formed following the date of this Agreement.

*"Finco Note"* shall mean a promissory note, issued by Holdco to Finco, in an aggregate principal amount not to exceed an amount equal to Fifteen Million Dollars (\$15,000,000).

"Finco Shares" shall mean the outstanding capital shares and any other outstanding equity or ownership interests of Finco.

"FIRPTA Certificate" shall have the meaning specified in <u>Section 2.1(r)</u>.

"First Closing" shall have the meaning specified in <u>Section 1.8(a)</u>.

7

"*First Closing Note*" shall mean a promissory note in the aggregate principal amount of Fifty Million Dollars (\$50,000,000) issued by Canadian OpCo to Company at the First Closing.

"First Closing Cash Payment" shall have the meaning specified in <u>Section 1.9(a)(i)</u>.

"First Closing Purchased Assets" shall have the meaning specified in Section 1.1.

"First Closing Shares" shall mean preferred shares of Canadian OpCo having a fair market value of Ten Million Dollars (\$10,000,000).

"Future Payment Notes" shall mean a series of three promissory notes (referred to individually as "Future Payment Note #1", "Future Payment Note #2" and "Future Payment Note #3") issued by Canadian OpCo to Company at the First Closing. The Future Payment Notes will have an aggregate principal amount equal to Fifty Million Dollars (\$50,000,000) minus the Year 1 Note Principal Amount (the "Aggregate Future Payment Notes Principal Amount"). Each individual Future Payment Note will have a principal amount equal to one-third (1/3) of the Aggregate Future Payment Notes Principal Amount (such principal amount of each Future Payment Note the "Future Note Principal Amount"), will not bear interest and will be secured by a subordinated security interest in the First Closing Purchased Assets. Further, the Note Payment Date for each Future Payment Note will be accelerated (a) in the event of a Change of Control of CLNE or (b) if the Business is sold by Purchasers or CLNE. Each Future Payment Note will provide that, on the applicable Note Payment Date, Purchasers will (i) pay to the holder of such note (by wire transfer of immediately available funds to a bank account designated no less than five (5) Business Days prior to such date) the Note Cash Payment for such Note Payment Date; and (ii) if necessary, deliver to the holder of such note a number of CLNE Shares calculated pursuant to the following formula:

<u>applicable Future Note Principal Amount minus the Note Cash Payment</u> the lower of (x) the Note Payment Price and (y) the Note Payment Midpoint Price

"GAAP" means U.S. generally accepted accounting principles in effect on the date on which they are to be applied pursuant to this Agreement, applied consistently throughout the relevant periods.

"General Assignment and Bill of Sale" shall have the meaning specified in Section 2.1(a).

"*Governmental Approval*" shall mean any: (a) permit, license, certificate, concession, approval, consent, ratification, permission, clearance, confirmation, exemption, waiver, franchise, certification, designation, rating, registration, variance, qualification, accreditation or authorization issued, granted, given or otherwise made available by or under the authority of any Governmental Authority or pursuant to any Legal Requirement; or (b) right under any Contract with any Governmental Authority.

"*Governmental Authority*" shall mean any: (a) nation, principality, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, provincial, local, municipal, foreign or other government; (c) governmental or quasi governmental authority of any nature (including any governmental division, subdivision, department, agency, minister, governor-in-council, cabinet, bureau, branch, office, commission,

council, board, instrumentality, officer, official, representative, organization, unit, body or Entity and any court or other tribunal); (d) multinational organization or body; or (e) individual, Entity or body exercising, or entitled to exercise, any executive, legislative, judicial, administrative, regulatory, police,

military or taxing authority or power of any nature.

"Government Antitrust Entity" shall have the meaning specified in Section 6.7(b)(i).

"Gross Profit" shall mean gross profit, as determined in accordance with GAAP.

"Hazardous Materials" shall have the meaning specified in Section 3.20(a).

"Holdco" shall have the meaning specified in the preamble, and shall also include any successor to Holdco as a result of an amalgamation prior to Closing.

"Holdco Preferred Shares" shall have the meaning specified in Section 3.21(h).

"HSR Act" shall mean the Hart Scott Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations thereunder.

"HSR Threshold" shall have the meaning specified in <u>Section 1.9(g)</u>.

"*Improvements*" shall have the meaning specified in <u>Section 3.23(c)</u>.

"*IMW Bangladesh*" shall have the meaning specified in <u>Section 3.7(d)</u>.

"IMW Bangladesh Financial Statements" shall have the meaning specified in Section 3.7(d).

"IMW China" shall have the meaning specified in Section 3.7(c).

"IMW China Financial Statements" shall have the meaning specified in Section 3.7(c).

"IMW Colombia" shall have the meaning specified in Section 3.7(b).

"IMW Colombia Financial Statements" shall have the meaning specified in Section 3.7(b).

"Indebtedness" means, with respect to any Person at any date, without duplication: (i) all obligations of such Person for borrowed money or in respect of loans or advances and other third-party financing parties; (ii) all obligations of such Person evidenced by bonds, debentures, notes or other similar instruments; (iii) all obligations of such Person that are not characterized as short term Liabilities under GAAP; (iv) all obligations in respect of letters of credit, whether or not drawn, and bankers' acceptances or similar credit transactions, whether or not then due, issued for the account of such Person; (v) all capitalized lease obligations of such Person; (vi) all interest rate protection agreements of such Person (valued on a market quotation basis); (vii) all obligation or financing-type arrangement in respect of the deferred purchase price of property or property received as of the Closing with respect to which such Person is liable, contingently or otherwise, as obligor or otherwise; (x) all earn-out obligations of such Person; (xi) any accrued interest, prepayment premiums or penalties or other costs or expenses related to any of the foregoing; and (xii) all obligations of any other Person of the type referred to in clauses (i) through (xi) which are secured by any Encumbrance on any property or asses of such Person.

9

"Indemnified Party" shall have the meaning specified in <u>Section 9.3(a)</u>.

"Indemnifying Party" shall have the meaning specified in Section 9.3(a).

"Independent Accounting Firm" shall have the meaning specified in Section 1.9(e).

"Insurance Policies" shall have the meaning specified in Section 3.22.

"Intellectual Property Licenses" shall mean (i) any grant to a Person of any right or authorization to use or practice or any covenant not to sue under or release from any claim of damages for infringement, unauthorized use or disclosure, or misappropriation of any Company IP Assets, and (ii) any grant to Company of any right or authorization to use or practice or any covenant not to sue under or release from any claim of damages for infringement, unauthorized use or disclosure, or misappropriation of any claim of damages for infringement, unauthorized use or disclosure, or misappropriation of any Person's Intellectual Property Rights or Technology which is necessary, useful, or otherwise related to the Business.

"Intellectual Property Rights" shall mean all right, title, and interest arising from or in respect of any of the following, whether protected, created or arising under the laws of Canada, the United States or any other jurisdiction: (i) all Patents, (ii) all Trademarks, (iii) all Copyrights, (iv) any Trade Secrets; and all rights, benefits, privileges, causes of action, and remedies relating to any of the foregoing, whether before or hereafter accrued, including the exclusive rights to apply for and maintain all registrations, renewals, and extensions, to sue for all past, present, and future infringements, unauthorized uses or disclosures, or misappropriations of any rights relating thereto, and to settle and retain proceeds from any such actions, and all contractual and other entitlements to royalties and other payments for the use or practice thereof.

"*Inventory*" means all inventory of Company Products and all raw materials, work in process and finished goods used, held for use or intended to be used in the Business, wherever located and whether held by Company or third parties.

"IRS Notice" shall have the meaning specified in Section 2.1(r).

*"ITU Application*" shall mean any intent-to-use trademark application with respect to any Purchased IP Asset that is pending at any time between the date of the Agreement and the Closing.

"Key Employees" shall have the meaning specified in Section 6.14(a).

"*Knowledge*" or "*knowledge*." Company, Holdco and Sellers shall be deemed to have "Knowledge" or "knowledge" of a particular fact or other matter if any of Company's directors or officers has actual knowledge of such fact or other matter or could be expected to discover or otherwise become aware of such fact or other matter in the course of conducting a reasonably comprehensive investigation concerning the truth or existence of such fact or other matter.

10

"Leased Real Property" shall have the meaning specified in Section 3.23(b).

"*Legal Requirement*" shall mean any federal, state, provincial, local, municipal, foreign or other law, statute, legislation, constitution, principle of common law, resolution, ordinance, code, Order, edict, decree, proclamation, treaty, convention, rule, regulation, permit, ruling, directive, pronouncement, requirement (licensing or otherwise), specification, determination, decision, opinion or interpretation that is, has been or may in the future be issued, enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Governmental Authority.

"*Liability*" or "*Liabilities*" shall mean any debt, obligation, duty, liability or Tax of any nature (including any unknown, undisclosed, unmatured, unaccrued, unasserted, contingent, indirect, conditional, implied, vicarious, derivative, joint, several or secondary liability), regardless of whether such debt, obligation, duty, liability or Tax would be required to be disclosed on a balance sheet prepared in accordance with generally accepted accounting principles and regardless of whether such debt, obligation, duty or liability is immediately due and payable.

"Machinery and Equipment" shall have the meaning specified in Section 1.1(c).

"Management Services Agreement" shall mean the Restated Management Services Agreement between 416854 B.C. Ltd., a corporation incorporated under the laws of the Province of British Columbia, and Company, dated July 1, 2010.

"*Material Adverse Effect*" means with respect to Sellers, Holdco, Company or any Company Subsidiary, any event, change or effect that, when taken individually or together with all other adverse events, changes and effects, is or is reasonably likely (a) to be materially adverse to the condition (financial or otherwise), properties, assets, Liabilities, business, operations, results of operations or prospects of Company, any Company Subsidiary or the Business or (b) to prevent or materially delay consummation of the Transaction or otherwise to prevent Sellers or Company from performing their obligations under this Agreement or any of the Transaction Agreements.

"Material Contracts" shall have the meaning specified in Section 3.13(a).

"Midpoint Price" means the sum of the Pre-Announcement Price plus the Pre-Closing Price, divided by two.

"*Moral Rights*" shall mean all moral, artist's, personal author's and paternity rights, including any right to claim authorship of a work or to attribution, right of integrity, respect for one's work, or to object to any distortion or other modification of a work, right with respect to publication, withdrawal, retraction, or correction, or right to receive remuneration.

"Negotiation Period" shall have the meaning specified in Section 11.12(a).

"Net Working Capital Adjustment" shall have the meaning specified in Section 1.10(d).

"Noncompetition Period" shall have the meaning specified in Section 6.13(a).

"Nondisclosing Party" shall have the meaning specified in Section 6.8(b).

11

"*Note Cash Payment*" means, for each Note Payment Date, the amount of cash paid by Purchasers to the holder of the applicable note, which shall be a minimum amount of Five Million Dollars (\$5,000,000) and may, at the sole option of Purchasers, be up to a maximum amount equal to the Future Note Principal Amount of the applicable Future Payment Note.

"Note Payment Date" means (i) with respect to Future Payment Note #1, January 31, 2012, (ii) with respect to Future Payment Note #2, January 31, 2013, and (iii) with respect to Future Payment Note #3, January 31, 2014.

"*Note Payment Midpoint Price*" means, for each Note Payment Date, the sum of the Pre-Closing Price and the Note Payment Price for such Note Payment Date, divided by two.

"*Note Payment Price*" means, for each Note Payment Date, the average closing price of CLNE Stock, as reported on the NASDAQ Stock Market, for the three trading days prior to such Note Payment Date.

"Order" shall mean any: (a) temporary, preliminary or permanent order, judgment, injunction, edict, decree, ruling, pronouncement, determination, decision, opinion, verdict, sentence, stipulation, subpoena, writ or award that is or has been issued, made, entered, rendered or otherwise put into effect by or under the authority of any court, administrative agency or other Governmental Authority or any arbitrator or arbitration panel; or (b) Contract with any Governmental Authority that is or has been entered into in connection with any Proceeding.

"Original Filing Date" shall have the meaning specified in <u>Section 6.7(b)</u>.

"Outside Closing Date" shall mean the date that is thirty (30) days following the date on which the last of the following is filed with the SEC: (i) the Audited Financial Statements, (ii) the pro forma financial information required by Form 8-K under the Exchange Act with respect to the Transaction and (iii) a description of the business of Company and the Company Subsidiaries, provided that if the Registration Statement is subject to review by the SEC, the Outside Closing Date shall be extended for a number of days equal to the number of days in the period beginning on the date the Registration Statement is filed with the SEC and ending on the date the Registration Statement is declared effective by the SEC.

#### "Owned and Leased Vehicles" shall have the meaning specified in Section 1.1(d).

"*Patents*" shall mean all letters patent and rights accorded under patent law systems, utility models, or equivalent forms of protection for intellectual property, and applications therefor, including continuations, divisionals, continuations-in-part, reissues, reexaminations, substitutions, renewals, and extensions thereof, and patents issuing thereon.

"Patent Assignment" shall have the meaning specified in Section 2.1(b)(ii).

"Permitted Encumbrances" shall mean the Encumbrances set forth on Schedule 6.9.

"Person" shall mean any individual, Entity or Governmental Authority.

12

"*Personal Information*" means information about an identifiable individual, but does not include an individual's name, position name or title, business telephone number, business address, business email or business fax number.

"Personal Property" shall have the meaning specified in Section 1.1(e).

"Personal Property Leases" shall have the meaning specified in Section 1.1(g).

"*Personally Identifiable Information*" shall mean any information that can be used to identify a specific individual as defined by applicable Legal Requirement in the relevant jurisdictions, such as the individual's name, address, telephone number, fax number, email address, credit card or financial account number, medical information, or health insurance information.

"*Post-Closing Acquisition*" shall mean any assets or businesses (a) acquired following the Closing with the approval of the Board of Directors of CLNE, (b) related to the Business and (c) operated in conjunction with the Business.

"*Pre-Announcement Price*" shall mean the average closing price of the CLNE Stock, as reported on the NASDAQ Stock Market, for the three trading days before the first public announcement of the Transaction.

"*Pre-Closing Period*" shall mean any taxable period ending on or before the close of business on the Closing Date or, in the case of any taxable period which includes, but does not end on, the Closing Date, the portion of such period up to and including the Closing Date.

"*Pre-Closing Price*" shall mean the average closing price of the CLNE Stock, as reported on the NASDAQ Stock Market, for the three trading days before the Closing Date.

"*Pre-Closing Taxes*" shall mean (i) any Taxes imposed on or payable by Company or any of the Company Subsidiaries for taxable periods (and portions thereof) beginning on or before the Closing Date, and (ii) any Taxes imposed on or payable by Company, Sellers or Holdco resulting from the transactions contemplated by this Agreement.

"Preliminary Closing Statement" shall have the meaning specified in Section 1.10(a).

"*Proceeding*" shall mean any action, suit, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), prosecution, contest, hearing, inquiry, inquest, audit, examination or investigation that is, has been or may in the future be commenced, brought, conducted or heard at law or in equity or before any Governmental Authority or any arbitrator or arbitration panel.

"Prohibited Business" shall have the meaning specified in Section 6.13(a).

"Property" shall have the meaning specified in Section 3.20(a).

"*Publicity Rights*" shall mean any rights with respect to publicity, privacy, name, nickname, sobriquet, signature, likeness, photograph, voice, sound effect, identity, personality, biographical information or materials, endorsement, quotation, attribution, or the like.

13

"Purchase Price" shall have the meaning specified in Section 1.9.

"**Purchased Assets**" shall mean, collectively, the First Closing Purchased Assets, the Purchased Shares, the Earnout Shares, the Depreciable Assets Shares, the First Closing Promissory Note, the First Closing Shares and the non-competition and non-solicitation agreements set forth in <u>Section 6.13</u> of the Agreement.

"*Purchased Shares*" shall mean all outstanding capital shares and any other outstanding equity or ownership interests of the Company Subsidiaries, together with any rights and options to acquire additional capital shares or other equity or ownership interests of the Company Subsidiaries.

"Purchasers" shall have the meaning specified in the preamble.

"Purchaser Parties" shall have the meaning specified in Section 9.2(a).

"*Real Property Leases*" means all leases, subleases, licenses, concessions and other agreements or Contracts (written or oral), including all amendments, extensions, renewals, guaranties, and other agreements with respect thereto, pursuant to which Company or any Company Subsidiary holds any Leased Real

Property, including the right to all security deposits, prepaid rent or prepaid expenses and other amounts and instruments deposited by or on behalf of Company or any Company Subsidiary.

"Registered IP Assets" shall have the meaning specified in Section 3.14(a).

"Registration Statement" shall have the meaning specified in Section 6.15(a).

"Release" shall have the meaning specified in <u>Section 3.20(a)</u>.

"*Representatives*" shall mean officers, directors, employees, attorneys, accountants, advisors, agents, distributors, licensees, shareholders, subsidiaries and lenders of a party. In addition, all Affiliates of Sellers and Company shall be deemed to be "Representatives" of Sellers.

"Resolutions" shall have the meaning specified in Section 3.2(a).

"Restricted Asset" shall have the meaning specified in Section 1.7(a).

"Resumption Notice" shall have the meaning specified in Section 6.15(c).

"Retained Purchased Shares" shall mean one (1) Purchased Share of IMW Colombia and one (1) Purchased Share of IMW Bangladesh.

"Return Policies" shall have the meaning specified in Section 3.28.

"SEC" shall mean the United States Securities and Exchange Commission.

"Second Closing" shall have the meaning specified in <u>Section 1.8(b)</u>.

14

"Second Closing Shares" shall mean a number of Canadian OpCo preferred shares having a value equal to the Finco Shares.

"Section 20(24) Election" shall mean a joint election to be executed by Company and Canadian OpCo under Section 20(24) of the *Income Tax Act* (Canada) and the corresponding sections of any other applicable provincial statute and any regulations under such statutes with respect to the sale, assignment, transfer and conveyance of any deferred revenue forming part of the Assumed Liabilities.

"Section 22 Election" shall mean a joint election to be executed by Company and Canadian OpCo under Section 22 of the *Income Tax Act* (Canada) and the corresponding sections of any other applicable provincial statute and any regulations under such statutes with respect to the sale, assignment, transfer and conveyance of the Accounts Receivable.

"Section 56.4(3) Election" shall mean joint election(s) to be executed, if appropriate in the circumstances, by the applicable Sellers and the applicable Purchasers under previously proposed Subsection 56.4(3) of the *Income Tax Act* (Canada), or any provision that may be enacted in replacement thereof and the corresponding provisions of any other applicable provincial statute in respect of restrictive covenants, with the elected amount based on the amount determined in accordance with <u>Section 10.5</u>.

"Section 85 Election" shall mean a joint election to be executed by Company and Canadian OpCo under Section 85 of the *Income Tax Act* (Canada) and the corresponding provisions of any other applicable provincial statute in respect of the sale, assignment, conveyance and transfer of the First Closing Purchased Assets.

"Section 167 Election" shall mean a joint election to be executed by Company and Canadian OpCo under Section 167 of the *Excise Tax Act* (Canada) and the corresponding provisions of any other applicable provincial statute and any regulations under such statutes on the forms prescribed for such purposes along with any documentation necessary or desirable in order to effect the transfer of the First Closing Purchased Assets by Company without payment of any HST. Canadian OpCo shall file the election forms referred to above, along with any documentation necessary or desirable to give effect to such, with Canada Revenue Agency together with Canadian OpCo's HST return for the reporting period in which the transactions contemplated herein are consummated.

"Securities Act" shall mean the Securities Act of 1933, as amended.

"Seller (and referred to together as "Sellers)" shall have the meaning specified in the preamble.

"Sellers' Basket" shall have the meaning specified in Section 9.4(a).

"Sellers' Indemnification Cap" shall have the meaning specified in Section 9.4(b).

"Specified Reps" shall have the meaning specified in Section 9.1.

"Subsidiary" means, with respect to any Person, any corporation, limited liability company, partnership, association, or other business entity of which (i) if a corporation, a majority of the total voting power of shares entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers, or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof or (ii) if a limited liability company, partnership, association, or other business entity, a majority of the partnership or other similar ownership interests thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more Subsidiaries of that Person or a combination thereof and for this purpose, a Person or Persons own a majority ownership interest in such a business entity (other than a corporation) if such

Person or Persons shall be allocated a majority of such business entity's gains or losses or shall be or control any managing director or general partner of such business entity (other than a corporation).

"Subsidiary Accounts Receivable" shall mean and include all accounts receivable, notes receivable and other monies due to any Company Subsidiary for sales and deliveries of goods, performance of services and other transactions (whether or not on the books of any Company Subsidiary).

"Suspension Notice" shall have the meaning specified in Section 6.15(c).

"Suspension Right" shall have the meaning specified in <u>Section 6.15(c)</u>.

"Survival Date" shall have the meaning specified in <u>Section 9.1</u>.

"*Tax*" (and, with correlative meaning, "*Taxes*") shall mean (a) any federal, provincial, state, local or foreign income, alternative or add-on minimum income, ad valorem, business license, capital, custom, disability, documentary, employment, environmental, excise, franchise, gains, gross income, gross receipts, import, license, occupation, payroll, personal property, premium, profits, property transfer, real property, recording, registration, sales, services, severance, social security, stamp, transfer, unemployment, unemployment insurance, Canada Pension Plan contributions, use, value added, wage, windfall profit or withholding tax, custom, duty, levy or other governmental assessment, charge or fee in the nature of a tax (whether payable directly or by withholding); (b) any liability for the payment of any amounts of the type described in (a) above as a result of any express or implied obligation to indemnify any other Person or as a result of any obligations under any agreements or arrangements with any other Person with respect to such amounts and including any liability for Taxes of a predecessor entity; and (c) any estimated Tax, interest, fines, penalties or additions to Tax with respect to amounts referred to in clauses (a) or (b) hereof.

"*Tax Return*" shall mean any return, report, estimate, declaration of estimated tax, claim for refund, information statement or other document relating to, or required to be filed in connection with, any Taxes, including information returns or reports with respect to backup withholding and other payments to third parties.

"*Technology*" shall mean, collectively, all technology, formulae, recipes, specifications (including information regarding materials, ingredients, tools, apparatus, sources, vendors), procedures, processes, methods, techniques, ideas, creations, inventions and discoveries (whether patentable or unpatentable and whether or not reduced to practice), improvements, know-how, research and development, technical data, designs, models, algorithms, subroutines, software,

16

works of authorship, copyrightable subject matter, and other similar materials, recordings, graphs, drawings, reports, analyses, and other writings, confidential or proprietary information, and general intangibles of like nature, and all tangible embodiments of the foregoing, in any form whether or not specifically listed herein.

"*Third Closing*" shall have the meaning specified in <u>Section 1.8(c)</u>.

"Trade Secrets" shall mean any trade secrets or similar forms of protection for confidential or proprietary information.

"Trademark Assignment" shall have the meaning specified in Section 2.1(b)(i).

"*Trademarks*" shall mean all trademarks, service marks, trade names, service names, brand names, trade dress, logos, designs, artwork or variants thereof, promotional materials, Internet domain names, IP addresses, email addresses, fictitious and other business names, personal names, identities, privacy rights, and general intangibles of like nature, together with the goodwill associated with any of the foregoing, and all applications, ITU Applications, registrations and renewals thereof.

"Transaction" shall mean, collectively, the transactions contemplated by this Agreement.

"*Transaction Agreements*" shall mean this Agreement and all other agreements, certificates, instruments, documents and writings delivered by Purchasers, Company, Holdco and/or Sellers in connection with the Transaction.

"Transferred Employees" shall have the meaning specified in Section 6.14(c).

"*Warranty*" shall mean any obligation to service, repair, replace, credit, refund or other obligations based upon or arising out of express and implied warranties made or deemed made in connection with the license or sale of goods or the performance of services related to the Business (collectively "*Warranties*").

"Workers" shall have the meaning specified in <u>Section 3.15(a)</u>.

"Year 1 Earn-Out Target" shall mean Fifteen Million One Hundred Thousand Dollars (\$15,100,000).

"Year 1 Midpoint Price" means the sum of the Pre-Closing Price and the Year 1 Price, divided by two.

"*Year 1 Payment*" means the amount of cash paid by Purchasers to the holder of the Year 1 Note, which shall be a minimum amount of Five Million Dollars (\$5,000,000) and may, at the sole option of Purchasers, be up to a maximum amount equal to the Year 1 Principal Amount.

"*Year 1 Price*" means the average closing price of CLNE Stock, as reported on the NASDAQ Stock Market, for the three trading days prior to the Year 1 Payment Date.

"Year 1 Note" shall mean a promissory note issued by Canadian AcqCo to Holdco at the Third Closing. The Year 1 Note will have a principal amount of no more than Twelve Million Five Hundred Thousand Dollars (\$12,500,000) (such principal amount, the "Year 1 Note Principal Amount"), will not bear interest, will be secured by a subordinated security interest in the First Closing Purchased Assets and will be due and payable on January 31, 2011 (the "Year 1 Payment Date"). The Year 1 Note will provide that, on the Year 1 Payment Date, Purchasers will (i) pay to the holder of such note (by wire transfer of immediately available funds to a bank account designated no less than five (5) Business Days prior to such date) the Year 1 Payment; and (ii) if necessary, deliver to the holder of such note a number of CLNE Shares calculated pursuant to the following formula:

Year 1 Principal Amount minus the Year 1 Payment

the lower of (x) the Year 1 Price and (y) the Year 1 Midpoint Price

"Year 2 Earn-Out Target" shall mean Eighteen Million Five Hundred Seventy-Three Thousand Dollars (\$18,573,000).

"Year 3 Earn-Out Target" shall mean Twenty-Three Million Six Hundred Four Thousand Dollars (\$23,604,000).

"Year 4 Earn-Out Target" shall mean Twenty-Seven Million Two Hundred One Thousand Dollars (\$27,201,000).

18

# EXHIBIT B

Example #1 of the Calculation of the Earn-Out Payment:

If, for the Year 1 Earn-Out Period, the Business achieves Adjusted Gross Profit of \$15,100,000 (100% of the Year 1 Earn-Out Target), then Canadian AcqCo would pay to Holdco \$2,718,000 (18% of the Adjusted Gross Profit of the Business for the Year 1 Earn-Out Period).

Example #2 of the Calculation of the Earn-Out Payment:

If, for the Year 2 Earn-Out Period, the Business achieves Adjusted Gross Profit of \$22,288,000 (120% of the Year 2 Earn-Out Target), then Canadian AcqCo would pay to Holdco \$5,126,000 (23% of the Adjusted Gross Profit of the Business for the Year 2 Earn-Out Period).

Example #3 of the Calculation of the Earn-Out Payment:

If, for the Year 3 Earn-Out Period, the Business achieves Adjusted Gross Profit of \$21,243,000 (90% of the Year 3 Earn-Out Target), then Canadian AcqCo would pay to Holdco \$2,018,000 (9.5% of the Adjusted Gross Profit of the Business for the Year 3 Earn-Out Period).

Example #4 of the Calculation of the Earn-Out Payment:

If, for the Year 4 Earn-Out Period, the Business achieves Adjusted Gross Profit of \$21,761,000 (80% of the Year 4 Earn-Out Target), then Canadian AcqCo would pay to Holdco \$979,000 (4.5% of the Adjusted Gross Profit of the Business for the Year 4 Earn-Out Period).

1

#### EXHIBIT C

#### **Company Historical Net Working Capital Calculation**

Current Assets:	
Accounts Receivable	\$ 9,595,635
Accrued A/R	3,242,226
Misc. Receivables	466,518
WIP	920,871
Inventory	7,066,580
Prepaid Expenses	226,594
Unrealized Foreign Exchange Gain	390,585
Income Taxes Refundable	222,042
	\$ 22,131,051
Current Liabilities:	
Unrealized Foreign Exchange Loss	\$ 
Bank Indebtedness	6,658,725
Accounts Payable & PO Accruals	6,857,711
Accrued Liabilities	1,865,488
Unearned Revenue	2,802,827
Accrued Warranty Provision	428,500
Current Portion of Long Term Debt	
Income Tax Payable	
	\$ 18,613,250
Company Historical Net Working Capital	\$ 3,517,800

# AMENDMENT TO ASSET PURCHASE AGREEMENT

**This Amendment to Asset Purchase Agreement** (this "*Amendment*") dated as of September 7, 2010, by and among Clean Energy, a California corporation ("*Clean Energy*"), 0884808 B.C. Ltd., a British Columbia corporation and a wholly-owned subsidiary of Clean Energy ("*Canadian AcqCo*") and Clean Energy Compression Corp, a British Columbia corporation formerly known as 0884810 B.C. Ltd and a wholly-owned subsidiary of Canadian AcqCo ("*Canadian OpCo*" and collectively with Clean Energy and Canadian AcqCo, "*Purchasers*"), on the one hand, and I.M.W. Industries Ltd., a British Columbia Corporation ("*Company*"), B&M Miller Equity Holdings Inc., a successor by amalgamation to 652322 B.C. Ltd., a British Columbia corporation ("*B&M*"), Bradley N. Miller, Marion G. Miller and Miller Family Trust (each of Miller Family Trust, Bradley N. Miller and Marion G. Miller, a "*Seller*" and together, "*Sellers*"), on the other hand, is entered into with reference to the following facts:

**WHEREAS**, Purchasers, Company, Holdco, Bradley N. Miller and Miller Family Trust are parties to that certain Asset Purchase Agreement dated as of July 1, 2010 (the "*Agreement*");

**WHEREAS**, pursuant to <u>Section 11.6</u> of the Agreement the parties desire to enter into this Amendment in order to amend the Agreement as set forth herein.

**NOW, THEREFORE**, the parties agree as follows.

1. **Amendment to Parties to the Agreement**. The Agreement is hereby amended to add "Marion G. Miller as a "**Seller**" and one of the "**Sellers**" for any and all purposes under the Agreement. In addition, all references to 652322 B.C. Ltd. in the Agreement are hereby amended, from and after August 23, 2010, to be references to B&M (and B&M shall be deemed to be "**Holdco**" for all purposes under the Agreement).

2. <u>Amendment of Section 1.3(a) of the Agreement</u>. <u>Section 1.3(a)</u> of the Agreement is hereby amended by adding "(other than the IMW China Shares and the IMW Bangladesh Shares)" immediately after the phrase "Purchased Shares" in such section.

3. <u>Amendment of Section 1.7 of the Agreement</u>. <u>Section 1.7</u> of the Agreement is hereby amended by adding the following new Section 1.7(d):

"(d) Without limiting the foregoing provisions of this <u>Section 1.7</u>, Purchasers, Company, Holdco and Sellers agree that in the event that, following the Closing, the assignment by Company to Purchasers of the Contract listed on <u>Schedule 1.7(d)</u> and/or the Patent listed on <u>Schedule 1.7(d)</u> is challenged by the other party to such Contract in a manner that affects adversely the rights of Purchasers in such Patent, then Company will at its expense (and Holdco and Sellers will cause Company to) use commercially reasonable efforts to effect an alternate arrangement, in the form of a license or other arrangement, in any case satisfactory to Purchasers, which results in Purchasers receiving all the benefits and bearing all the ordinary course costs, liabilities and other obligations with respect to such Patent."

4. <u>Amendment of Section 1.9(a)(i) of the Agreement</u>. Section 1.9(a)(i) of the Agreement is hereby amended by replacing "Fifteen Million Dollars (\$15,000,000)" with "Fourteen Million Seven Hundred Twelve Thousand Dollars (\$14,712,000)".

5. <u>Amendment of Section 1.9(c)(ii) of the Agreement</u>. <u>Section 1.9(c)(ii)</u> of the Agreement is hereby amended by adding "(other than the IMW China Shares and the IMW Bangladesh Shares)" immediately after the phrase "Purchased Shares" in the first line of such section.

6. <u>Amendment of Section 1.9 of the Agreement</u>. <u>Section 1.9</u> of the Agreement is hereby amended by adding the following new Section 1.9(i):

"(i) Purchasers, Company, Holdco and Sellers agree that (i) the transfer of the IMW China Shares to Purchasers will be governed by the terms of the Equity Transfer Agreement dated July 23, 2010 between Canadian OpCo and Company, and (ii) the transfer of the IMW Bangladesh Shares to Purchasers will occur as soon as possible under applicable Law following the Closing."

7. **<u>Amendment of Section 2.1(f) of the Agreement</u>**. <u>Section 2.1(f) of the Agreement is hereby amended and restated in its entirety to read as</u> follows:

"(f) <u>Employment Agreements</u>. An Employment Agreement duly executed by Bradley N. Miller (the "*Employment Agreements*").

8. <u>Amendment of Section 2.5(b) of the Agreement</u>. <u>Section 2.5(b)</u> of the Agreement is hereby amended by adding "(other than the IMW China Shares and the IMW Bangladesh Shares)" immediately after the phrase "Purchased Shares" in such section.

9. <u>Amendment of Section 7.1(e) of the Agreement</u>. <u>Section 7.1(e)</u> of the Agreement is hereby amended and restated in its entirety to read as follows:

"(e) <u>Employment</u>. No less than 80% of the Key Employees shall have accepted employment with Purchasers."

10. **Amendment of Section 10.3 of the Agreement**. Section 10.3 of the Agreement is hereby amended and restated in its entirety to read as follows:

"10.3 **Tax Elections**. On or before the completion of the procedures set forth in <u>Section 10.5</u>, Sellers and Company shall provide Purchasers with an executed Section 85 Election, Section 20(24) Election, Section 56.4(3) Election and Section 167 Election."

11. **Amendment of Exhibit A of the Agreement**. Exhibit A of the Agreement is hereby amended to include the following additional defined

terms:

*"IMW Bangladesh Shares"* shall mean all outstanding capital shares and any other outstanding equity or ownership interests of IMW Bangladesh, together with any rights and options to acquire additional capital shares or other equity or ownership interests of IMW Bangladesh."

"*IMW China Shares*" shall mean all outstanding capital shares and any other outstanding equity or ownership interests of IMW China, together with any rights and options to acquire additional capital shares or other equity or ownership interests of IMW China."

12. **Counterparts**. This Amendment may be executed in counterparts with the same force and effect as if each of the signatories had executed the same instrument.

- 13. **Definitions**. Capitalized terms not otherwise defined herein shall have the respective meanings set forth in the Agreement.
- 14. **Other**. Except as provided herein, the Agreement shall continue in full force and effect in accordance with its terms.

# [Remainder of page intentionally left blank]

IN WITNESS WHEREOF, each of the parties has caused this Agreement to be executed on its behalf by their respective officers thereunto duly authorized all as of the date first written above.

"Purchasers"		
Clean Energy		
By:	/s/ Andrew J. Littlefair	
Name:	Andrew J. Littlefair	
Title:	President and CEO	
0884808	B.C. Ltd.	
By:	/s/ Andrew J. Littlefair	
Name:	Andrew J. Littlefair	
Title:	President and CEO	
Clean En	ergy Compression Corp.	
By:	/s/ Andrew J. Littlefair	
Name:	Andrew J. Littlefair	
Title:	President and CEO	
"		
"Compar	-	
	ndustries Ltd. /s/ Bradley N. Miller	
By: Name:	Bradley N. Miller	
Title:	President	
me.	President	
" <b>B&amp;</b> M"		
B&M Mi	iller Equity Holdings Inc.	
By:	/s/ Bradley N. Miller	
Name:	Bradley N. Miller	
Title:	President	
"Sellers"		
Bradley I	N. Miller	
By:	/s/ Bradley N. Miller	
Marion G	G. Miller	
By:	/s/ Marion G. Miller	
Miller Fa	mily Trust	
By:	/s/ Bradley N. Miller	
Name:	Bradley N. Miller	
Title:	Trustee	
muc.	1105000	

MORRISON | FOERSTER

425 MARKET STREET SAN FRANCISCO CALIFORNIA 94105-2482

TELEPHONE: 415.268.7000 FACSIMILE: 415.268.7522

WWW.MOFO.COM

MORRISON & FOERSTER LLP

NEW YORK, SAN FRANCISCO, LOS ANGELES, PALO ALTO, SAN DIEGO, WASHINGTON, D.C.

DENVER, NORTHERN VIRGINIA, SACRAMENTO, WALNUT CREEK

TOKYO, LONDON, BEIJING, SHANGHAI, HONG KONG, BRUSSELS

September 7, 2010

Clean Energy Fuels Corp. 3020 Old Ranch Parkway, Suite 400 Seal Beach, CA 90740

Re: 4,017,408 Shares of Common Stock of Clean Energy Fuels Corp.

Ladies and Gentlemen:

This opinion is furnished to Clean Energy Fuels Corp., a Delaware Corporation (the "<u>Company</u>"), in connection with the resale by B&M Miller Equity Holdings Inc. (the "Selling Stockholder") of up to 4,017,408 shares of the Company's Common Stock, par value \$0.0001 per share (the "<u>Shares</u>"), pursuant to a Registration Statement on Form S-3 (File No. 333-168433) originally filed with the Securities and Exchange Commission (the "<u>Commission</u>") under the Securities Act of 1933, as amended (the "<u>Act</u>") on July 30, 2010 (the "<u>Registration Statement</u>") and the related prospectus included therein (the "<u>Prospectus</u>") and the prospectus supplement filed with the Commission pursuant to Rule 424(b)(3) promulgated under the Act on September 7, 2010 (the "<u>Prospectus</u> <u>Supplement</u>"). All of the Shares are to be sold by the Selling Stockholder as described in the Registration Statement and the related Prospectus and Prospectus Supplement.

In connection with this opinion, we have examined the Company's Restated Certificate of Incorporation, and the Company's By-laws, both as currently in effect, such other records of the corporate proceedings of the Company and certificates of the Company's officers as we have deemed relevant, the Registration Statement and the exhibits thereto, and the related Prospectus and Prospectus Supplement. In addition, we have examined such records, documents, certificates of public officials and the Company, made such inquiries of officials of the Company and considered such questions of law as we have deemed necessary for the purpose of rendering the opinion set forth herein. In such examination, we have assumed the genuineness of all signatures and the authenticity of all items submitted to us as originals and the conformity with originals of all items submitted to us as copies.

Based upon, subject to and limited by the foregoing, we are of the opinion that the Shares have been duly authorized, validly issued, and are fully paid and nonassessable.

We express no opinion as to matters governed by any laws other than the Delaware General Corporation Law in effect on the date hereof.

We hereby consent to the filing of this opinion letter as Exhibit 5.1 to the Registration Statement and to reference to us under the caption "Legal Matters" in the Prospectus Supplement. In giving such consent, we do not hereby admit that we are acting within the category of persons whose consent is required under Section 7 of the Act or the rules or regulations of the Commission thereunder.

Very truly yours,

/s/ Morrison & Foerster LLP

# **Consent of Independent Registered Public Accounting Firm**

We consent to the incorporation by reference in the registration statements (Nos. 333-152306, 333-145434, 333-150331, 333-156776, 333-159799, 333-164301 and 333-168433) on Form S-3 and Form S-8 of Clean Energy Fuels Corp. of our report dated July 15, 2010, with respect to the combined balance sheets of IMW Industries Group as of April 30, 2010 and the related combined statements of loss and comprehensive loss, retained earnings (deficit) and cash flows for the year then ended, which report appears in the Form 8-K of Clean Energy Fuels Corp. dated September 3, 2010.

/s/ KPMG LLP

Chilliwack, Canada September 7, 2010

## IMW INDUSTRIES GROUP

Year ended April 30, 2010

# **KPMG** Enterprise<sup>™</sup>

200 - 9123 Mary Street

Chilliwack BC V2P 4H7

Telephone (604) 793-4700

Fax (604) 793-4747

# **AUDITORS' REPORT**

To the Shareholder of IMW Industries Group

We have audited the accompanying combined balance sheet of IMW Industries Group as at April 30, 2010 and the related combined statements of loss and comprehensive loss, retained earnings (deficit) and cash flows for the year then ended. These financial statements are the responsibility of the Group's management. Our responsibility is to express an opinion on these financial statements based on our audit.

We conducted our audit in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provides a reasonable basis for our opinion.

In our opinion, the combined financial statements referred to above present fairly, in all material respects, the financial position of the IMW Industries Group as at April 30, 2010 and the results of its operations and its cash flows for the year ended April 30, 2010 in conformity with Canadian generally accepted accounting principles.

Canadian generally accepted accounting principals vary in certain significant respects from US generally accepted accounting principles. Information related to the nature and effect of such differences is presented in Note 21 to the combined financial statements.

/s/ KPMG LLP

**Chartered Accountants** 

Chilliwack, Canada

July 15, 2010

#### **IMW INDUSTRIES GROUP**

**Combined Balance Sheet** 

April 30, 2010

	 2010
Assets	
Current assets:	
Cash	\$ 621,868
Accounts receivable (Note 3)	8,025,899
Accrued revenue	3,008,820
Inventories (Note 4)	9,398,685
Future income taxes (Note 15)	252,879
Prepaid expenses and deposits	915,571
	22,223,722
Due from related parties (Note 8)	154,939
Property and equipment (Note 5)	2,612,190
Deferred development costs (Note 6)	 186,403

	\$ 25,177,254
Liabilities and Shareholder's Equity	
Liabilities and Shareholder's Equity	
Current liabilities:	
Bank indebtedness (Note 7)	\$ 6,693,305
Accounts payable and accrued liabilities	11,351,804
Warranty provision	590,329
Income taxes payable	30,176
Unearned revenue (Note 9)	1,706,851
Current portion of long-term debt (Note 10)	322,247
Current portion of obligations under capital lease (Note 11)	94,061
	 20,788,773
Long-term debt (Note 10)	133,985
Obligations under capital lease (Note 11)	98,957
Due to related parties (Note 12)	1,278,507
Preferred shares (Note 14)	450,000
Future income taxes (Note 15)	196,841
Shareholder's Equity:	
Share capital (Note 14)	16,966
Contributed surplus	2,561,726
Cumulative translation account	2,763
Deficit	(351,264)
	 2,230,191
Commitments (Note 17)	
Communents (Note 17)	
Contingent liability (Notes 7 and 13)	
Subsequent event (Note 20)	 
	\$ 25,177,254
See accompanying notes to combined financial statements.	

See accompanying notes to combined financial statements.

# IMW INDUSTRIES GROUP

Combined Statement of Loss and Comprehensive Loss

Year ended April 30, 2010

	2010
Revenue	\$ 47,639,404
Cost of goods sold	41,292,784
	6,346,620
Expenses:	
General and administration	5,522,595
Selling expenses	2,605,305
Amortization of development costs	141,702
Amortization	463,711
	8,733,313
Loss from operations	(2,386,693)
Other income (expense):	
Foreign exchange gain	1,625,011
Interest on operating line	(262,070)
Interest on long-term debt	(39,618)
	1,323,323
Loss before income taxes	(1,063,370)
Income taxes recovery (Note 15):	

Current Future	 (145,282) (143,759) (289,041)
Net loss and comprehensive loss	\$ (774,329)

See accompanying notes to non-consolidated financial statements.

# IMW INDUSTRIES GROUP

Combined Statement of Retained Earnings (Deficit)

Year ended April 30, 2010

		2010
Retained earnings, beginning of year	\$	423,065
Net loss for the year		(774,329)
Deficit, end of year	<u>\$</u>	(351,264)

See accompanying notes to combined financial statements.

# IMW INDUSTRIES GROUP

Combined Statement of Cash Flows

Year ended April 30, 2010

2010

# Cash provided by (used in):

Operations:

Net loss	\$ (774,329)
Items not involving cash:	
Amortization of property and equipment	463,711
Amortization of deferred development costs	141,702
Future income taxes	(143,759)
Foreign exchange gain	(23,991)
Change in non-cash operating working capital (Note 16)	(406,423)
	 (743,089)

Financing:

Increase in bank indebtedness	1,667,680
Proceeds of long-term debt	356,997
Repayment of long-term debt	(234,104)
Payment of obligations under capital lease	(22,425)
	1,768,148

# Investments:

Advances from related parties	270,388
Property and equipment additions	(945,748)
Investment in projects - deferred development costs	(248,697)
Advances to related party	(153,708)
	(1,077,765)
Change in cash position	(52,706)
Cash position, beginning of year	674,574
Cash position, end of year	\$ 621,868
Cubit position, chi of year	,

Supplemental cash flow information (Note 16)

See accompanying notes to combined financial statements.

# General:

The primary business activity of the IMW Industries Group (the "Group") is the design, manufacture and sale of natural gas compressors in Chilliwack, BC, and throughout the world.

#### 1. Significant accounting policies:

(a) Basis of combination:

The combined financial statements include the accounts of IMW Industries Ltd. ("IMW Canada"), IMW CNG Bangladesh Ltd. ("IMW Bangladesh"), IMW Colombia Ltda. ("IMW Colombia"), IMW Compressors (Shanghai) Co. Ltd. ("IMW Shanghai") and IMW Industries Inc. ("IMW US"). All of the entities in the Group are 100% controlled directly or indirectly by 652322 B.C. Ltd. All significant inter-company transactions and balances have been eliminated.

(b) Basis of presentation:

These combined financial statements are presented in Canadian dollars and have been prepared in accordance with Canadian generally accepted accounting principles. A reconciliation to amounts presented in accordance with United States generally accepted accounting principles ("US GAAP") is detailed in Note 21. In prior years, no combined financial statements were completed.

(c) Inventories:

Raw material and components are stated at the lower of cost, which is direct material and freight, and net realizable value, on a first-in, first-out basis. Work in progress and finished goods inventory is recorded at the lower of cost, which includes the direct cost of materials plus freight, labour and overhead, and net realizable value.

# 1. Significant accounting policies (continued):

(d) Property and equipment:

Property and equipment are initially recorded at cost.

Amortization is provided using the following methods and annual rates:

Asset	Basis	Rate
Production equipment	Declining balance	9-25%
Computer equipment	Declining balance	18-30%
Computer software	Declining balance	100%
Office equipment	Declining balance	10-20%
Automotive equipment	Declining balance	20-50%

The Group assesses the carrying amount of long-lived assets, including those under capital lease, initially based on the net recoverable amounts determined on an undiscounted cash flow basis. If the carrying amount of an asset exceeds its net recoverable amount, an impairment loss is recognized to the extent that fair value is below the asset's carrying amount. Fair value is determined based on quoted market prices when available, otherwise on discounted cash flows over the life of the asset.

(e) Deferred development costs:

The Group expenses development expenditures as incurred unless certain criteria to capitalize an asset have been met. Capitalized expenditures are amortized over the estimated period of production. When it is determined that capitalized costs will not likely be recovered by the future revenue, such capitalized costs are expensed.

(f) Government assistance:

#### Repayable government contributions

Claims for assistance under various repayable government contribution agreements are recorded as a liability when conditions indicate that the future repayment is likely. When conditions indicate that repayment is not likely, the grant is applied to reduce development costs. When an agreement expires the balance remaining is recorded as a reduction in the cost of the related asset or expense.

#### Government grants and assistance

Government grants and assistance are recorded as a reduction of the cost of related assets or expenditures in the period in which eligible expenditures are incurred and the government grant is earned. Grants in excess of costs incurred are included as income.

# 1. Significant accounting policies (continued):

(g) Revenue recognition:

Equipment contract revenue is accounted for under the percentage of completion method prorated based on costs incurred for each contract compared to the estimated total costs. When a loss on a contract can be reasonably estimated, the total estimated amount of the loss is charged to income for the year. Percentage of completion revenues in excess of amounts billed are recorded as accrued revenues. Progress billings in excess of costs incurred and deposits on contracts are recorded as unearned revenue.

Revenues from contracts with multiple-element arrangements, such as those including installation, maintenance and training fees, are recognized as each element is earned based on the relative fair value of each element and only when there are no undelivered elements that are essential to the functionality of the delivered elements.

Revenue from all other products and services is recognized when the product is completed and shipped and the customer takes ownership and assumes risk of loss, collection of the relevant receivable is probable, persuasive evidence of an arrangement exists and the sales price is fixed or determinable.

(h) Foreign currency translation:

Transaction amounts denominated in foreign currencies are translated into their Canadian dollar equivalents at exchange rates prevailing at the transaction dates. Carrying values of monetary assets and liabilities reflect the exchange rates at the balance sheet date. Translation gains and losses are included in current earnings.

The Group's subsidiaries are accounted for as self-sustaining foreign subsidiary operations. Subsidiary transaction amounts denominated in foreign currencies are translated into their Canadian dollar equivalents at exchange rates prevailing at the transaction dates. Carrying values of monetary assets and liabilities reflect the exchange rates at the balance sheet date. Translation gains and losses are included in current earnings. Translation gains and losses related to the net investment in foreign subsidiaries are included in the Cumulative Translation Account.

# 1. Significant accounting policies (continued):

(i) Future income taxes:

The Group follows the asset and liability method of accounting for future income taxes. Under this method, future income tax assets and liabilities are based on temporary differences between the carrying amount of balance sheet items and their corresponding tax bases. In addition, the future benefits of income tax assets, including unused tax losses, are recognized, subject to a valuation allowance, to the extent that it is more likely than not that such future benefits will ultimately be realized. Future income tax assets and liabilities are measured using substantially enacted tax rates and laws expected to apply when the tax liabilities or assets are to be either settled or realized.

(j) Financial instruments:

The Group accounts for its financial assets and liabilities in accordance with Canadian generally accepted accounting principles.

The financial instruments are classified into one of five categories: held-for-trading, held-to maturity, loans and receivables, available-for-sale financial assets or other financial liabilities. All financial instruments, including derivatives, are measured in the balance sheet at market value except for loans and receivables, held to maturity investments and other financial liabilities which are measured at amortized cost. Subsequent measurement and changes in market value will depend on their initial classification, as follows: held-for-trading financial assets are measured at market value and changes in market value are recognized in net earnings; available-for-sale financial instruments are measured at market value with changes in market value recorded in comprehensive income.

The Group has designated the following:

- (i) Cash is held-for-trading, being measured at fair value.
- (ii) Accounts receivable and due from related parties are classified as loans and receivables, which are measured at amortized cost.
- (iii) Bank indebtedness, accounts payable and accrued liabilities, long-term debt, capital leases, preferred shares, and due to related parties are classified as other financial liabilities, which are measured at amortized cost.

# 1. Significant accounting policies (continued):

(k) Derivative financial instruments:

Derivative financial instruments are financial contracts whose value is derived from interest rates, foreign exchange rates, or other financial indices.

In the ordinary course of business, the Group enters into various derivative contracts, including foreign exchange contracts. Derivative financial instruments may be undertaken to manage risk from the Group's exposures to fluctuation in exchange rates related to the U.S. dollar or other currencies and are accounted from on an accrual basis to the extent they meet hedging criteria. Derivative instruments may be undertaken for trading purposes and are accounted for on a mark-to-market basis with the resulting net gain or loss recognized as other income in the Statement of Operations in the current period. The Group does not apply hedge accounting to any of its derivative contracts.

(l) Use of estimates:

The preparation of financial statements in conformity with Canadian generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. Areas of significant estimate include accounts receivable as to their collectability, provisions for slow moving and obsolete inventory, warranties on products sold, useful lives of capital assets, and the repayability of government contributions.

These estimates are reviewed periodically and, as adjustments become necessary they are reported in earnings in the period in which they become known.

#### 2. Change in accounting policy:

Effective May 1, 2009, the Group adopted the following accounting standards:

- (a) CICA Section 3031, *Inventories*, which provides more extensive guidance on measurement and expands disclosure requirements to increase transparency. Reversal of previous write-downs to net realizable value when there is a subsequent increase in the value of inventories is now required. Adoption of this standard did not result in any effect on the Group's financial statements.
- (b) CICA Section 1535, *Capital Disclosures*, which requires the Group to disclose the nature of any externally imposed capital requirements, whether the Group has complied with those requirements during the period and, when the Group has not complied with externally imposed capital requirements, what the consequences are of such non-compliance.

#### 2. Change in accounting policy (continued):

This standard has not resulted in any additional disclosure in these financial statements as the Group is not subject to any externally imposed capital restrictions.

- (c) CICA Section 1400, General Standards of Financial Statement Presentation. This Section changes the guidance related to management's responsibility to assess the ability of the Group to continue as a going concern. Management is required to make an assessment of the Group's ability to continue as a going concern and should take into account all available information about the future, which is at least, but not limited to, 12 months from the balance sheet date. Disclosure is required of material uncertainties related to events that may cast significant doubt upon the Group's ability to continue as a going concern. Adoption of this standard did not result in any effect on the Group's financial statements.
- (d) CICA Section 3064, *Goodwill and Intangible Assets*. This new Section provides guidance on the recognition, measurement, presentation and disclosure of goodwill and intangible assets. Adoption of this standard did not have any effect on the Group's financial statements.
- (e) Amendments to CICA Section 3862, *Financial Instruments-Disclosures* which enhanced disclosure requirements for financial instruments using a fair value hierarchy. Adoption of this standard did not have any effect on the Group's financial statements.

#### 3. Accounts receivable:

 2010
\$ 7,219,216
372,000
434,683
\$ 8,025,899
\$ 

#### 4. Inventories:

	2010	
Raw materials and components	\$ 8,121,8	
Work in progress	783,9	
Finished goods	492,8	36
	\$ 9,398,6	85

Included in cost of goods sold is \$27,771,211 in inventory costs.

#### 5. Property and equipment:

	 Cost	 ccumulated nortization	 2010 Net book value
Leasehold improvements	\$ 279,827	\$ 72,254	\$ 207,573

Production equipment	3,104,39	7 1,531,323	1,573,074
Computer equipment	735,62	6 483,925	251,701
Computer software	848,81	4 577,422	271,392
•			
Office equipment	284,62	6 110,073	174,553
Automotive equipment	349,35	9 215,462	133,897
	\$ 5,602,64	9 \$ 2,990,459	\$ 2,612,190
Deferred development costs:			
r			
			 2010
Costs, beginning of year			\$ 79,408
Costs incurred in the year			248,697
Amortization			(141,702)
			\$ 186,403

# 7. Bank indebtedness:

6.

The Group has an operating line of credit to a maximum of \$7,750,000 bearing interest at prime plus 1.25%. In addition, the Group has a bank guarantee line with a limit of \$3,000,000, a forward exchange contract line with a limit of \$13,750,000, and a MasterCard limit with a maximum of \$150,000. These lines of credit are secured by a general security agreement providing a first priority security interest in all present and after acquired personal property of the Group, including specific charges on all serial numbered goods, inventory and other assets, assignment of risk insurance, unlimited corporate guarantees from the shareholder and unlimited personal guarantees. The various lines of credit are due on demand and are subject to financial covenants.

The Group has an operating line of credit for 4,000,000 RMB (\$593,000) bearing interest at 6 month Peoples Bank of China rate plus 2.5% and has a bank guarantee line of 1,000,000 RMB (\$148,000). These lines are secured by the same security as above.

The Group has a 16,750,000 BDT (\$239,000) operating line of credit bearing interest at 14%.

The Group has a 320,000,000 COP (\$166,000) operating line of credit bearing interest at Colombia benchmark rate plus 7 to 9%.

The bank guarantee lines allow the Group to provide guarantees and/or standby letters of credit to overseas suppliers or bid/performance deposits on contracts. These amounts are contingencies guaranteed by the Group's banks.

The forward exchange contract line allows the Group to enter into foreign exchange forward contracts up to the notional limit of \$13,750,000. No forward exchange contracts were outstanding at April 30, 2010.

Amounts owing under the MasterCard line are included in accounts payable.

The following tables summarize the various operating lines of credit, the guarantee lines of credit and the amounts outstanding on the lines (all in Canadian dollar equivalent amounts) at April 30, 2010:

	 Approved Amount		Amount outstanding
Operating lines of credit:			
Canada	\$ 7,750,000	\$	5,901,733
China	593,000		578,961
Bangladesh	239,000		147,864
Colombia	166,000		64,747
	\$ 8,748,000	\$	6,693,305

#### 7. Bank indebtedness (continued):

Borrowings under operating lines of credit are limited based on amounts of approved receivables and inventories.

	 Approved amount	 Guarantees outstanding
Guarantee lines (contingent liability):		
Canada	\$ 3,000,000	\$ 1,859,265
China	148,000	

\$	3.148.000	\$	1 859 265
Ψ	5,140,000	Ψ	1,000,200

The Group was in violation of certain restrictive covenants related to the Canadian lines during the year but was within the covenants at April 30, 2010. The bank has indicated that they are aware of the default and have agreed to forbear the breach.

# 8. Due from related parties:

		2010
Manly Marine Closures Ltd.	\$	150,000
416854 B.C. Ltd		3,857
818509 B.C. Ltd		1,082
	<u>\$</u>	154,939

Manly Marine Closures Ltd., 416854 B.C. Ltd. and 818509 B.C. Ltd. and the Group are controlled by the same shareholders.

The advances are without interest or fixed terms of repayment.

# 9. Unearned revenue:

Unearned revenue consists of the following:

\$ 1,268,872
 437,979
\$ 1,706,851
\$

# 10. Long-term debt:

	 2010
Epicor Software Corp, note payable in monthly installments of \$15,458 including interest at 5% due March 5, 2011; unsecured	\$ 165,865
GE, capital equipment loan in monthly installments of \$4,223 including interest at 6.32% due January 21, 2013; secured by specific equipment	123,350
Banco De Bogota loan payable in Colombian Pesos in monthly installments of 5,685,276 COP (\$2,883) plus interest at 12.6%; due November 25, 2012	89,373
GE, capital equipment loan in monthly installments of \$4,223 including interest at 6.32% due January 21, 2013; secured by specific equipment	40,145
Vancity Capital loan payable in monthly installments of \$4,167 plus interest at prime plus 11.75%; secured by a pari passu interest in all present and acquired intellectual property, corporate and personal guarantees by various related parties, assignment of business insurance proceeds and an assignment of life insurance proceeds; due May 15, 2010	37,499
	456,232
Current portion of long-term debt	 322,247
	\$ 133,985

Long term debt is subject to certain covenants. The Vancity Capital loan due May 15, 2010 has continued on a month to month basis with no change in terms subsequent to the due date.

Principal repayments on long-term debt in each of the next three years are estimated as follows:

2011 \$ 2012 2013	3	322,247 101,826
\$		32,159
*	4	456,232

# 11. Obligations under capital lease:

The Company has financed certain equipment by entering into capital lease arrangements. Capital lease repayments are due as follows:

	 2010
2011	\$ 105,220
2012	102,194
2013	11,206
	 218,620
Less amount representing interest, at rates between 6.7% and 14%	(25,602)
Present value of net minimum lease payments	193,018
Current portion of obligations under capital lease	94,061
	\$ 98,957

2010

Interest expense of \$10,498 relating to capital lease obligations is included in interest on long-term debt.

The net book value of assets under capital lease is \$194,321.

#### **12. Due to related party:**

	 2010
652322 B.C. Ltd.	\$ 1,252,507
Advanced Bending Technologies Inc.	26,000
	\$ 1,278,507

The advances from 652322 B.C. Ltd. and Advanced Bending Technologies Inc. are non-interest bearing and without fixed terms of repayment. The repayment of these advances have been postponed under the Group's banking agreements.

652322 B.C. Ltd. is the parent company of the Group.

Advanced Bending Technologies Inc. and the Group are controlled by the same shareholders.

# **13. Contingent liability:**

The Group has received, net of a holdback of \$372,000 outstanding, \$1,860,000 from Natural Resources Canada ("NRC") towards the development of a high pressure compressed hydrogen fuelling system (the "Contribution Agreement"). Under the Contribution Agreement, the grant is repayable over a ten year period commencing March 31, 2010 at a rate of 2% of revenue related to the project. The development project is now complete and the final report has been sent to NRC. The grant has been netted against specific project costs incurred. Future hydrogen compressor revenues, if any, will result in the 2% payment to NRC in the year earned. As of April 30, 2010, no revenue has been earned related to this project.

The repayment provisions remain in effect for ten years or until Natural Resource Canada has received an amount equal to the total contributions made.

#### 14. Share capital:

IMW US and IMW Shanghai are wholly-owned subsidiaries of IMW Canada; accordingly, all shares of these companies are eliminated upon combination. IMW Bangladesh is owned 98% by IMW Canada and 2% by 652322 B.C. Ltd. IMW Colombia is 50% owned by each of IMW Canada and 652322 B.C. Ltd.

The authorized and issued share capital is as follows:

2	2010
Authorized:	
IMW Canada	
An unlimited number of Class A voting common shares without par value	
An unlimited number of Class B voting common shares without par value	
An unlimited number of Class C non-voting common shares without par value	
An unlimited number of Class D non-voting common shares without par value	
An unlimited number of Class E redeemable, retractable non-voting preferred shares without par value, redeemable at \$7,885 each.	
An unlimited number of Class F, retractable non-voting preferred shares with a par value of \$0.01 each	
An unlimited number of Class G, retractable non-voting preferred shares without par value, redeemable at \$1,000 each.	
An unlimited number of Class H, retractable non-voting preferred shares with a par value of \$0.01 each, redeemable at \$100 each.	
IMW Bangladesh	
10,000 Common shares	
IMW Colombia	
1,000 Common shares	

Issued:	
100 Class A common shares - IMW Canada (100%)	\$ 100
3 Common shares - IMW Bangladesh (2%)	3
500 Common shares - IMW Colombia (50%)	16,863
	\$ 16,966

In addition, the Group has issued 450 Class G non-voting preferred shares with a total redemption value of \$450,000. As these shares are redeemable, on demand, by the holder, they are recorded as a liability of the Group.

# 15. Income taxes:

The tax effect of temporary differences that give rise to significant portions of the future tax assets and future tax liabilities at April 30, 2010 are presented below:

	 April 30, 2010
Future tax assets:	
Tax losses carried forward – Shanghai	\$ 17,049
Financing costs - tax deductions	1,668
Warranty provision	164,111
Unrealized inter-company profit	87,100
Deferred development costs	 45,926
	 315,854
Less valuation allowance - Shanghai tax losses	(17,049)
	 298,805
Future tax liabilities:	
Capital assets - net book values in excess of undepreciated capital cost	242,767
	 242,767
Net future tax asset	\$ 56,038
The net future tax asset is comprised of the following:	
Current future tax asset	\$ 252,879
Long-term future tax liability	(196,841)
	\$ 56,038

Income tax expense differs from the amount that would be computed by applying the Canadian and Federal and Provincial statutory income tax rates of 29.5% to income before taxes. The reasons for the differences are as follows:

# 15. Income taxes (continued):

		2010
Loss before income taxes	<u>\$</u>	(1,063,370)
Computed tax recovery	\$	(313,694)
Add (deduct):		
Impact of higher income tax rates in foreign jurisdictions		26,031
Permanent differences		16,212
Impact of losses in foreign jurisdiction		20,812
Other		(38,402)
	\$	(289,041)

# 16. Supplementary statement of cash flow information:

	 2010
Changes in non-cash operating working capital:	
Accounts receivable	\$ (2,659,961)
Accrued revenue	702,135
Inventories	894,762
Prepaid expenses and deposits	(554,086)
Foreign currency contracts	33,675

	1 000 055
Accounts payable and accrued liabilities	1,320,857
Warranty provision	109,031
Unearned revenue	(159,997)
Income taxes	(92,839)
	\$ (406,423)
Supplemental cash flow information:	
Interest paid	\$ 300,567
Income taxes paid	\$ 90,371
Non-cash transaction:	
Capital assets acquired financed by capital leases	\$ 29,498

# 17. Commitments:

The Group has entered into various operating lease agreements with estimated minimum annual payments as follows:

2011	\$ 972,100
2012	772,900
2013	718,800
2014	721,100
2015	750,000
Thereafter	2,062,500
	\$ 5,997,400

#### 18. Related party transactions:

Included in expenses for the year ended April 30, 2010 are management fees totaling \$360,000 paid to 416854 B.C. Ltd., a company affiliated with a shareholder of 652322 B.C. Ltd.

This transaction was conducted in the normal course of operations and is measured at the exchange amount, which is the amount of consideration established and agreed to by the related parties.

#### **19. Financial instruments:**

# Credit concentration

As at April 30, 2010, one customer accounted for 27% of accounts receivable and 9% of revenue. Another customer accounts for 14% of revenues but had no receivable balance at April 30, 2010. Approximately 82% of the Group's accounts receivable are insured by Export Development Canada or are supported by letters of credit from the customer's bank. The Group believes that there is no unusual exposure associated with the collection of these receivables. The Group performs regular credit assessments of its customers and provides allowances for potentially uncollectable accounts receivable.

#### Interest rate risk

Interest rate risk is the risk that the value of a financial instrument might be adversely affected by a change in the interest rates. In seeking to minimize the risks from interest rate fluctuations, the Group manages exposure through its normal operating and financing activities. The Group is exposed to interest rate risk primarily through its long-term borrowing totaling \$456,232 and through its bank indebtedness totaling \$5,270,946, both of which are subject to changes in the prime rate.

# **19.** Financial instruments (continued):

A 1% change in prime rate would change interest expense by approximately \$52,700 per year.

# Foreign exchange risk

Foreign exchange risk is the risk that the value a financial instrument might be adversely affected by due to a change in exchange rates. The Group purchases most of its materials in US dollars and sells its product in various currencies, including US Dollars, Colombian Pesos, Bangladesh Taka and the Renmimbi Yuan. To manage the foreign exchange risk, the Group enters into various derivative foreign exchange contracts.

At April 30, 2010, there were no derivative foreign exchange contracts outstanding.

# Fair values

For the Group's financial instruments that include accounts receivable, bank indebtedness, accounts payable and accrued liabilities and current portions of long term debt and obligations under capital lease, fair values approximate carrying values due to the immediate or short-term maturity of these financial instruments.

For amounts due to or from related parties and preferred shares, it is not practical to estimate fair values as repayment dates, interest rates or other factors are not known.

#### 20. Subsequent event:

On July 1, 2010, the shareholders of the combined companies agreed to sell the majority of the business assets and operations of the IMW Industries Group.

The closing date for the sale has yet to be determined.

# 21. Reconciliation to US Generally Accepted Accounting Principles ("US GAAP"):

The Group prepares its combined financial statements in accordance with Canadian GAAP, which, as applied in these combined financial statements, conform in all material respects to the measurement principles in US GAAP, except as summarized below:

# Reconciliation

The effect of material measurement differences between Canadian GAAP and US GAAP on the combined balance sheet and combined statements of loss and comprehensive loss, retained earnings (deficit) and cash flows is summarized as follow:

# 21. Reconciliation to US Generally Accepted Accounting Principles ("US GAAP") (continued):

		 2010
(i)	Assets	
	Total assets, under Canadian GAAP	\$ 25,177,254
	Deferred development costs expensed (a)	 (186,403)
	Total assets, under US GAAP	\$ 24,990,851
(ii)	Liabilities	
	Total liabilities, under Canadian GAAP	\$ 22,947,063
	Preferred shares reclassified (b)	 (450,000)
	Total liabilities, under US GAAP	\$ 22,497,063
(iii)	Temporary equity	
	Total temporary equity, under Canadian GAAP	\$ —
	Preferred shares reclassified (b)	 450,000
	Total temporary equity, under US GAAP	\$ 450,000
(iv)	Retained earnings (deficit)	
	Opening retained earnings, under Canadian GAAP	\$ 423,065
	Deferred development costs expensed, under US GAAP - prior year (a)	(79,408)
	Net loss for the year, under US GAAP (Note 23(v))	 (881,324)
	Closing deficit, under US GAAP	\$ (537,667)
(v)	Net loss and comprehensive loss for the year	
	Net loss and comprehensive loss for the year, under Canadian GAAP	\$ (774,329)
	Current year deferred development costs expensed, under US GAAP (a)	(248,697)
	Amortization of deferred development costs reversed (a)	 141,702
	Net loss and comprehensive loss for the year, under US GAAP	\$ (881,324)

# 21. Reconciliation to US Generally Accepted Accounting Principles ("US GAAP") (continued):

			2010
(vi)	Cash flows		
	Cash used in operating activities, under Canadian GAAP	\$	(743,089)
	Adjustment for deferred development costs (a)		(248,697)
	Cash used in operating activities, under US GAAP	\$	(991,786)
	Cash used in financing activities, under Canadian and US GAAP	\$	1,768,148
	Cash used in investing activities, under Canadian GAAP	\$	(1,077,765)
	Adjustment for deferred development costs (a)	<u> </u>	248,697
	Cash used in investing activities, under US GAAP	\$	(829,068)

(a) Deferred development costs:

US GAAP requires development costs to be charged to expense when incurred. Under Canadian GAAP, development costs are deferred when certain criteria are met, and are amortized over the estimated period of production. The deferred costs would be written off if it is determined that costs will not likely be recovered by the future revenue.

Accordingly, the Group has expensed all development costs as incurred and reversed previously recognized amortization expense for US GAAP purposes for the year presented. For Canadian GAAP, cash flows relating to deferred development costs are reported as investing activities. For US GAAP, these costs would be characterized as operating activities.

# (b) Preferred shares:

The Group has issued preferred shares redeemable at the option of the holder (Note 14). US GAAP requires these preferred shares to be presented as temporary equity outside of equity and liabilities. Any changes in the carrying value of the preferred shares under US GAAP are recorded in equity. Under Canadian GAAP, the preferred shares are classified as a financial liability.

# 21. Reconciliation to US Generally Accepted Accounting Principles ("US GAAP") (continued):

(c) Income taxes:

US GAAP requires deferred tax assets and liabilities to be measured using tax rates that are enacted at the balance sheet date. Under Canadian GAAP, future tax assets and liabilities are measured using tax rates that are substantively enacted at the balance sheet date. There were no significant differences between enacted and substantively enacted tax rates for the year presented.

In June 2006, FASB issued guidance on recognition and measurement of uncertainties in income taxes under ASC 740 (Financial Interpretation No. 48, Accounting for Uncertainty in Income Taxes - An Interpretation of SFAS Statement No. 109). This guidance is effective for the Group's fiscal year ended April 30, 2008. The adoption of this guidance did not have a significant effect on the Group's results of operations or financial position.

(d) Recent adopted US accounting pronouncements:

# The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles

In July 2009, US GAAP switched to a new codification scheme aimed to simplify US GAAP accounting research. The statement is effective for financial statements issued for interim and annual periods ending after September 15, 2009 and is adopted by the Group in these combined financial statements for the year ended April 30, 2010. The codification did not affect the Group's combined financial statements as the codification did not change US GAAP.

#### Subsequent Events

In May 2009, the FASB issued guidance on subsequent events included in ASC Subtopic 855-10 (Statement of Financial Accounting Standards No. 165, Subsequent Events) ("SFAS 165"). The guidance is effective for financial statements ending after June 15, 2009. SFAS 165 establishes general standards of accounting for and disclosure of subsequent events that occur after the balance sheet date. The adoption of this standard did not impact the Group's combined financial statements for the year ended April 30, 2010.

# 21. Reconciliation to US Generally Accepted Accounting Principles ("US GAAP") (continued):

In December 2007, the FASB issued a revised standard on accounting for business combinations in ASC Topic 805 (SFAS-141R). The guidance is effective for periods beginning on or after December 15, 2008 and requires fair value measurement for all business acquisitions, including preacquisition contingencies. The guidance also expands the existing definition of a business and removes certain acquisition related costs from the purchase price consideration. The provisions are effective for the fiscal year beginning January 1, 2009. The adoption of this standard did not impact the Group's combined financial statements.

# Non-Controlling Interests

In December 2007, the FASB issued ASC Topic 805 ("SFAS-160") "Non-controlling Interests in Consolidated Financial Statements", which specifies that non-controlling interests are to be treated as a separate component of equity, not as a liability or other item outside of equity. Because non-controlling interests are an element of equity, increases and decreases in the parent's ownership interest that leave control intact are accounted for as capital transactions. The statement is effective for business combinations entered into on or after December 15, 2008, and is to be applied prospectively to all non-controlling interests, including any that arose before the effective date. The adoption of this standard did not impact the Group's combined financial statements for the year ended April 30, 2010.

#### Multiple-Deliverable Revenue Arrangements

In October 2009, FASB provided amendments to the criteria for separating consideration in multiple-deliverable arrangements, established a selling price hierarchy for determining the selling price of a deliverable, and eliminated the residual method of allocation of consideration by requiring that arrangement consideration be allocated at the inception of the arrangement to all deliverables using the relative selling price method.

FASB also requires expanded disclosures related to multiple-deliverable revenue arrangements, including information about the significant judgments made and changes to those judgments, as well as how the application of the relative selling-price method affects the timing and amount of revenue recognition. These amendments will be effective prospectively for revenue arrangements entered into or materially modified in fiscal years beginning on or after June 15, 2010. The Group is currently assessing the impact of these amendments on its combined financial statements.

#### **Unaudited Pro Forma Condensed Combined Financial Information**

On July 6, 2010, Clean Energy Fuels Corp, a Delaware corporation (the "Registrant" or the "Company"), announced that it had entered into an agreement to purchase the advanced natural gas fueling compressor and related equipment manufacturing and servicing business of I.M.W. Industries Ltd., a British Columbia corporation ("IMW"), pursuant to the terms of an asset purchase agreement (the "Purchase Agreement"). The acquisition contemplated by the Purchase Agreement closed on September 7, 2010.

The following unaudited pro forma condensed combined financial statements combine the historical consolidated balance sheets and statements of operations of the Registrant and IMW, giving effect to the acquisition of IMW using the purchase method of accounting.

The pro forma information presented, including allocation of the purchase price, is based on preliminary estimates of the fair values of assets acquired and liabilities assumed. The Company's judgments used to determine the estimated fair value assigned to each class of assets acquired and liabilities assumed, as well as asset lives, can materially impact the Company's results from operations. The finalization of the Company's purchase accounting assessment will result in changes in the valuation of assets and liabilities acquired which could be material. The Company will finalize the purchase price allocation as soon as practicable within the measurement period in accordance with applicable US GAAP accounting rules, but in no event later than one year following the closing of the acquisition. The purchase price allocation for the Company is preliminary in all respects. Certain historical balance sheet and income statement amounts of IMW have been reclassified to conform to the financial statement presentation of the Registrant.

The unaudited pro forma condensed combined balance sheet information as of June 30, 2010 gives effect to the acquisition of IMW as if it had occurred on June 30, 2010. The unaudited pro forma condensed combined statements of operations for the six months ended June 30, 2010 and the year ended December 31, 2009 give effect to the acquisition as if the acquisition had occurred at the beginning of the respective periods.

The unaudited pro forma condensed combined financial information, including the notes thereto, does not give effect to any cost savings or other synergies that could result from the acquisition. The unaudited pro forma condensed combined financial statements are presented for illustrative purposes only and are not necessarily indicative of the consolidated financial position or consolidated results of operations of the Registrant that would have been reported had the acquisition occurred on the dates indicated, nor do they represent a forecast of the consolidated financial position of the Registrant at any future date or the consolidated results of operations of the Registrant at any future date or the consolidated results of operations of the Registrant for any future period.

The unaudited pro forma condensed combined financial statements, including the notes thereto, should be read in conjunction with the historical consolidated financial statements, including the notes thereto, and other information of the Registrant included in its Annual Report on Form 10-K for the year ended December 31, 2009 and Quarterly Report on Form 10-Q for the quarterly period ended June 30, 2010, respectively, and the financial statements of IMW included in Exhibits 99.10f this Form 8-K.

# Clean Energy Fuels Corp. and Subsidiaries Unaudited Pro Forma Condensed Combined Balance Sheet June 30, 2010

(In thousands)

		Historical Pro Form			Forma	na		
	Clean Energy		IMW	А	diustments		C	ombined
Assets	 Life B		111111		ajuounento			<u>omonica</u>
Current assets:								
Cash and cash equivalents	\$ 55,701	\$	610	\$	(14,738) (i(	(a))	\$	41,573
Restricted cash	2,500							2,500
Accounts receivable, net of allowance for doubtful accounts	19,067		13,458		(946) (v	vi)		31,579
Other receivables	9,208		3,207		(356) (v	vi)		12,059
Inventory, net	7,880		8,817		150 (v	vi)		16,847
Prepaid expenses and other current assets	8,712		1,505					10,217
Total current assets	 103,068		27,597		(15,890)			114,775
Land, property and equipment, net	179,596		2,460		(456) (v	vi)		181,600
Deferred development costs			129		(129) (ii	i)		
Capital lease receivables	1,175		—					1,175
Notes receivable and other long-term assets	11,287		145					11,432
Investments in other entities	11,075		—					11,075
Goodwill	21,572		—		44,773 (ii	ii)		66,345
Intangible assets, net of accumulated amortization	32,905		—		81,400 (iv	v)		114,305
Total assets	\$ 360,678	\$	30,331	\$	109,698		\$	500,707
Liabilities and Stockholders' Equity				-				
Current liabilities:								
Current portion of long-term debt and capital lease obligations	\$ 2,420	\$	7,813	\$	12,125 (i(	(b))	\$	22,358
Accounts payable	12,998		10,178		(946) (v	vi)		22,230
Accrued liabilities	16,625		5,310		2,228 (i(	(c))		24,178
					288 (i(	(a))		
					(273) (v	vi)		
Deferred revenue	2,468		2,230		—			4,698
Total current liabilities	 34,511		25,531		13,422			73,464
Long-term debt and capital lease obligations, less current portion	9,525		219		31,375 (i(	(b))		41,119
Preferred shares			423		(423) (v	7)		_
Other long-term liabilities	38,068		2,136		(1,915) (v	vi)		45,361
					7,072 (i(	(c))		
Total liabilities		_						

	82,104	28,309	49,531	159,944
Total equity	278,574	2,022	(2,022) (v)	340,763
			62,189 (i(a))	
Total liabilities and equity	\$ 360,678	\$ 30,331	\$ 109,698	\$ 500,707

See accompanying notes to unaudited pro forma condensed combined financial statements.

# Clean Energy Fuels Corp. Notes to the Unaudited Pro Forma Condensed Combined Balance Sheets (In thousands, except share information)

# Note 1 — Basis of Presentation

The unaudited pro forma condensed combined financial statements included herein have been prepared by the Company, without audit, pursuant to the rules and regulation of the Securities and Exchange Commission. The Company believes that the disclosures are adequate to make the information not misleading. The Company remeasured IMW's balance sheet at June 30, 2010, utilizing the closing spot US dollar to Canadian dollar exchange rate on that day.

# Note 2 — Pro Forma Adjustments

#### (i) Purchase Consideration

The estimated total purchase consideration of the acquisition is based on the estimated fair values of the following purchase considerations:

Upfront consideration (a)	\$ 76,927
Anniversary payments (b)	43,500
Contingent consideration (c)	9,300
Acquisition holdback payment (d)	288
Total purchase consideration	\$ 130,015

(a) payment at closing of \$15.6 million in cash, inclusive of \$0.2 million of non-recurring acquisition related costs and \$0.6 million related to a working capital adjustment (which will be allocated to the appropriate working capital items), and the issuance of 4.0 million shares of the Registrant's common stock totaling \$62.2 million in common stock based on its closing stock price as of September 3, 2010; (b) the estimated fair value of the anniversary payments of \$43.5 million, consisting of payments on each of January 31, 2011, 2012, 2013, 2014 of \$12.5 million (each payment consisting of \$5.0 million in cash and \$7.5 million in cash and/or shares of common stock, with the exact number of cash and/or common stock to be determined at the Company's option; (c) the estimated fair value of an earn-out arrangement based on IMW achieving certain minimum amounts of adjusted gross profit (capped at \$40 million); (d) amount represents a holdback payment that will be released to the Sellers once the Chinese regulatory authorities approve the transfer of IMW Compressors (Shanghai) Co., Ltd. to the Company.

#### (ii) Deferred Development Costs

U.S. generally accepted accounting principles requires development costs to be charged to expense as incurred. Under Canadian generally accepted accounting principles, development costs are deferred when certain criteria are met, and are amortized over the estimated period of production. This amount represents the adjustment required to conform to US generally accepted accounting principles.

# (iii) Goodwill

The estimated total purchase price of the acquisition is based on the price of \$130.0 million for the net assets of IMW. The total transaction value was allocated to IMW's assets and liabilities on a fair value basis and resulted in goodwill of approximately \$44.8 million.

#### (iv) Identifiable Intangible Assets

Amount represents the pro forma adjustments to record the estimated fair values of IMW's identifiable intangible assets relating to technology, trade names, certain customer relationships, or other intangible assets acquired as a result of the acquisition based on preliminary estimates. The fair values and asset lives of these assets are subject to adjustments upon completion of the Company's valuations.

(v) Total Stockholders' Equity

Amount represents the pro forma adjustments to eliminate the historical stockholders' equity of IMW.

(vi) Assets & Liabilities Not Purchased and Elimination of Intercompany Transactions

Amounts represent adjustments to assets & liabilities not purchased by the Company and elimination of intercompany sales transactions.

# Unaudited Pro Forma Condensed Statement of Operations

For the Six Months Ended June 30, 2010 (In thousands, except share and per share amounts)

		Historical			Pro Forma				
	Clean Energy			IMW		Adjustments		Combined	
Revenue:		Energy			<u> </u>			Somonieu	
Product revenues	\$	73,706	\$	20,877	\$	(529) (v)	\$	94,054	
Service revenues		9,317		5,157		—		14,474	
Total revenues		83,023		26,034		(529)		108,528	
Operating expenses:									
Cost of sales:									
Product cost of sales		54,188		19,005		(275) (v)		72,918	
Service cost of sales		3,986		3,779				7,765	
Selling, general and administrative		28,527		4,119		(504) (ii)		32,142	
Depreciation and amortization		10,061		347		2,763 (i)		13,171	
Derivative loss on Series I warrant valuation		1,989						1,989	
Total operating expenses		98,751		27,250		1,984		127,985	
Operating loss		(15,728)		(1,216)		(2,513)		(19,457)	
Interest income (expense), net		87		(147)		(78) (iii)		(138)	
Other income, net		5		325		_		330	
Income from equity method investments		105		_				105	
Loss before income taxes		(15,531)		(1,038)		(2,591)		(19,160)	
Income tax benefit (expense)		1,127		119		882 (iv)		2,128	
Net loss		(14,404)		(919)		(1,709)		(17,032)	
Income attributable to noncontrolling interest		(67)						(67)	
Net loss	\$	(14,471)	\$	(919)	\$	(1,709)	\$	(17,099)	
Loss per share:									
Basic	\$	(0.24)				(vi)	\$	(0.27)	
Diluted	\$	(0.24)				(vi)	\$	(0.27)	
Weighted average common shares outstanding:									
Basic		60,494,148				(vi)		64,511,557	
Diluted		60,494,148				(vi)		64,511,557	

See accompanying notes to unaudited pro forma condensed combined financial statements.

# **Clean Energy Fuels Corp. and Subsidiaries** Unaudited Pro Forma Condensed Combined Statement of Operations For the Twelve Months Ended December 31, 2009 (In thousands, except share and per share amounts)

	Historical			Pro Forma				
		Clean Energy		IMW		Adjustments		Combined
Revenue:						<u> </u>		
Product revenues	\$	116,635	\$	28,145	\$	(868) (v)	\$	143,912
Service revenues		14,868		8,177		—		23,045
Total revenues		131,503		36,322		(868)		166,957
Operating expenses:								
Cost of sales:								
Product cost of sales		76,766		25,774		(605) (v)		101,935
Service cost of sales		6,155		5,410		_		11,565
Selling, general and administrative		47,509		6,589		(152) (ii)		53,946
Depreciation and amortization		16,992		559		5,617 (i)		23,168
Derivative loss on Series I warrant valuation		17,367		—		_		17,367
Total operating expenses		164,789		38,332		4,860		207,981
Operating loss		(33,286)		(2,010)		(5,728)		(41,024)
Interest expense, net		(32)		(293)		(156) (iii)		(481)
Other income (expense), net		(310)		1,794		—		1,484
Income from equity method investments		244		—		—		244
Loss before income taxes		(33,384)		(509)		(5,884)		(39,777)
Income tax benefit (expense)		(304)		75		1,734 (iv)		1,505
Net loss		(33,688)		(434)		(4,150)		(38,272)
Loss attributable to noncontrolling interest		439		—		—		439
Net loss	\$	(33,249)	\$	(434)	\$	(4,150)	\$	(37,833)
Loss per share:								
Basic	\$	(0.60)				(vi)	\$	(0.64)

Diluted	<u>\$ (0.60</u> )	(vi)	<u>\$ (0.64)</u>
Weighted average common shares outstanding:			
Basic	55,021,961	(vi)	59,039,370
Diluted	55,021,961	(vi)	59,039,370

See accompanying notes to unaudited pro forma condensed combined financial statements.

# Clean Energy Fuels Corp. Notes to the Unaudited Pro Forma Condensed Combined Statements of Operations (In thousands)

# Note 1 — Basis of Presentation

The unaudited pro forma condensed combined financial statements included herein have been prepared by the Registrant, without audit, pursuant to the rules and regulation of the Securities and Exchange Commission. The Registrant believes that the disclosures are adequate to make the information not misleading. The Registrant remeasured IMW's statements of operations for the six months ended June 30, 2010 and the year ended December 31, 2009 at the average exchange rates over the respective periods.

# Note 2 — Pro Forma Adjustments

# (i) Amortization and Development Costs

To reflect the pro forma effect on amortization expense of IMW's identifiable intangible assets to their estimated fair values at the date of the acquisition. The amount of this adjustment may change as the values and asset lives of the underlying asset valuations are finalized.

U.S. generally accepted accounting principles requires development costs to be charged to expense as incurred. Under Canadian generally accepted accounting principles, development costs are deferred when certain criteria are met, and are amortized over the estimated period of production. This amount represents the adjustment required to conform to US generally accepted accounting principles for the respective periods presented.

# (ii) Selling General and Administrative

Amounts adjust for related party management fees, non recurring acquisition related costs and deferred development costs for the six month period ended June 30, 2010.

Amounts adjust for related party management fees and deferred development costs for the twelve month period ended December 31, 2009.

# (iii) Interest Income (Expense), Net

Adjustment to record reduction in estimated interest income earned for the six months ended June 30, 2010 and the year ended December 31, 2009 on cash and cash equivalents as a result of paying the upfront cash consideration at closing of the acquisition.

# (iv) Income Tax Provision

To record the income tax benefit of the net pro forma adjustments at IMW's Canadian Federal and Provincial statutory rates of 29.5%.

Tax expense related to the Company's net pro forma adjustments for US tax purposes was not recorded as the Company is in a taxable loss position for US tax purposes and the Company has provided a full valuation allowance against its net US deferred tax assets.

# (v) Intercompany Transactions

Amounts represent intercompany transactions eliminated for the respective periods presented.

#### (vi) Earnings per Share

The following table summarizes the computation of the unaudited pro forma combined weighted-average basic shares outstanding and weighted-average dilutive shares outstanding. These two amounts are used in the calculation of loss per share.

	Six Months Ended June 30, 2010
Historical Registrant weighted-average basic and dilutive shares outstanding	60,494
Shares issued at closing of acquisition	4,017
Weighted-average basic and dilutive shares outstanding	64,511
	Year Ended December 31, 2009
Historical Registrant weighted-average basic and dilutive shares outstanding	55,022
Shares issued at closing of acquisition	4.017
Shares issued at closing of acquisition	4,017

The following table recalculates basic and diluted loss per share.

Numerator	
Net loss	\$ (17,099)
Denominator	
Weighted-average basic and dilutive shares outstanding	64,511
Loss per share:	
Basic	\$ (0.27)
Diluted	\$ (0.27)
	ear Ended nber 31, 2009
Numerator	
Net loss	\$ (37,833)
Denominator	
Weighted-average basic and dilutive shares outstanding	59,039
Loss per share:	
Basic	\$ (0.64)
Diluted	\$ (0.64)



3020 Old Ranch Parkway, Suite 400 Seal Beach, California 90740 USA 562.493.2804 fax: 562.546.0097 www.cleanenergyfuels.com

**News Release** 

# **Clean Energy Completes Acquisition of IMW**

# – Positions Clean Energy as a World Leader in Natural Gas Fueling and Compression Technology for the Growing US and International NGV Markets –

# Conference Call To Be Held Today, September 7 at 11 am PDT/2 pm EDT

Seal Beach, Calif. (September 7, 2010) – Clean Energy Fuels Corp. (Nasdaq: CLNE) has completed the acquisition of substantially all of the assets of IMW Industries, Ltd. (IMW). IMW manufactures and services advanced natural gas fueling compressors and related equipment that is sold to station operators and commercial fleets and has provided more than 1,000 installations in over 20 countries. IMW is headquartered near Vancouver, British Columbia and has a second manufacturing facility near Shanghai, China.

The purchase price includes \$15.6 million in cash and 4,017,408 shares of Clean Energy stock that will be issued to IMW's shareholder. An additional \$288,000 will be paid by Clean Energy once the Chinese regulatory authorities approve the transfer of IMW Compressors (Shanghai) Co. Ltd. to the Company, which is anticipated within the next forty-five days. The shares issued at closing are registered and available for immediate resale by the IMW shareholder. Clean Energy has also agreed to pay IMW's shareholder four additional annual payments of \$12.5 million each (consisting of \$5 million in cash plus an additional \$7.5 million in cash or Clean Energy common stock to be determined at Clean Energy's option) on January 31, 2011 through 2014. IMW's shareholder may also receive an earn-out payment based on certain percentages of future gross profits earned by IMW over the next four years, assuming the gross profit exceeds certain minimum thresholds, up to a maximum of \$40 million in the aggregate.

"This acquisition positions Clean Energy strategically in the forefront of the rapidly growing natural gas vehicle industry worldwide," said Andrew J. Littlefair, Clean Energy President and CEO. "It also strengthens our service package domestically as an integrated provider of advanced compression technology, station-building and fueling."

# **Conference Call Information**

Investors interested in participating in the live call to be held today, Tuesday, Sept. 7 at 11 am Pacific time, can dial 877.407.4018 from the U.S. International callers can dial 201.689.8471. A telephone replay will be available approximately two hours after the call concludes and will be available through Thursday, October 7, 2010 by dialing 877.660.6853 from the U.S., or 201.612.7415 from international locations, and entering account number 3055 and conference ID number 356627. There also will be a simultaneous, live webcast available on the Investor Relations section of the Company's web site at www.cleanenergyfuels.com. The webcast will be archived and available for replay for 30 days.

**Clean Energy** (Nasdaq: CLNE) is the leading provider of natural gas (CNG and LNG) for transportation in North America. It has a broad customer base in the refuse, transit, ports, shuttle, taxi, trucking, airport and municipal fleet markets, fueling more than 18,300 vehicles at 200 strategic locations across the United States and Canada. Clean Energy owns and operates two LNG production plants, one in Willis, Texas and one in Boron, Calif., with combined capacity of 260,000 LNG gallons per day and designed to expand to 340,000 LNG gallons per day as demand increases. It owns and operates a landfill gas facility in Dallas, Texas, that produces renewable methane gas, or biomethane, for delivery in the nation's gas pipeline network. Its subsidiary, IMW Industries, is a leading supplier of compressed natural gas (CNG) equipment for vehicle fueling and industrial applications with more than 1,000

1

installations in over 20 countries. Clean Energy also owns BAF Technologies, Inc. of Dallas, a leading provider of natural gas vehicle systems and conversions for taxis, limousines, vans, pick-up trucks and shuttle buses. Please visit www.cleanenergyfuels.com

**Forward-Looking Statements** – This news release contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934 that involve risks, uncertainties and assumptions, including statements about future payments that may be made by Clean Energy to the IMW shareholder, the timing for the transfer of IMW Compressors (Shanghai) Co. Ltd. and future sales opportunities for IMW's products and services. Actual results and the timing of events could differ materially from those anticipated in these forward-looking statements due to lack of demand for IMW's products and services, unanticipated regulatory delays, or deterioration of the U.S. or global economy and changes in governmental regulations in other countries that promote natural gas vehicle programs. The forward-looking statements made herein speak only as of the date of this press release and the company undertakes no obligation to publicly update such forward-looking statements to reflect subsequent events or circumstances.

#### Contacts

**News Media** Bruce Russell, 310/559-4955 x101 brussell@cleanenergyfuels.com

**Investors** Ina McGuinness, 805-427-1372